IDENTIFICATION AND EVALUATION OF BIOFLOCCULANT FROM *Bacillus salmalaya* 139SI FOR ITS APPLICATION IN WASTEWATER TREATMENT

ZAYED M. M. ABU TAWILA

FACULTY OF SCIENCE UNIVERSITY OF MALAYA KUALA LUMPUR

2019

IDENTIFICATION AND EVALUATION OF BIOFLOCCULANT FROM *Bacillus salmalaya* 139SI FOR ITS APPLICATION IN WASTEWATER TREATMENT

ZAYED M. M. ABU TAWILA

THESIS SUBMITTED IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

INSTITUTE OF BIOLOGICAL SCIENCES FACULTY OF SCIENCE UNIVERSITY OF MALAYA KUALA LUMPUR

2019

UNIVERSITY OF MALAYA ORIGINAL LITERARY WORK DECLARATION

Name of Candidate: ZAYED M. M. ABU TAWILA

Matric No: SHC150018

Name of Degree: **DOCTOR OF PHILOSOPHY**

Title of Thesis:IDENTIFICATIONANDEVALUATIONOFBIOFLOCCULANT FROM Bacillus salmalaya139SI FOR ITS APPLICATIONIN WASTEWATER TREATMENT

Field of Study: **BIOHEALTH SCIENCE**

I do solemnly and sincerely declare that:

- (1) I am the sole author/writer of this Work;
- (2) This Work is original;
- (3) Any use of any work in which copyright exists was done by way of fair dealing and for permitted purposes and any excerpt or extract from, or reference to or reproduction of any copyright work has been disclosed expressly and sufficiently and the title of the Work and its authorship have been acknowledged in this Work;
- (4) I do not have any actual knowledge nor do I ought reasonably to know that the making of this work constitutes an infringement of any copyright work;
- (5) I hereby assign all and every rights in the copyright to this Work to the University of Malaya ("UM"), who henceforth shall be owner of the copyright in this Work and that any reproduction or use in any form or by any means whatsoever is prohibited without the written consent of UM having been first had and obtained;
- (6) I am fully aware that if in the course of making this Work I have infringed any copyright whether intentionally or otherwise, I may be subject to legal action or any other action as may be determined by UM.

Candidate's Signature

Date:

Subscribed and solemnly declared before,

Witness's Signature

Date:

Name:

Designation:

IDENTIFICATION AND EVALUATION OF BIOFLOCCULANT FROM Bacillus salmalaya 139SI FOR ITS APPLICATION IN WASTEWATER TREATMENT

ABSTRACT

Bioflocculants are flocculating compounds produced by microorganisms during their growth and has recently received extensive consideration from researchers due to their biodegradable, non-toxicity and lack of secondary pollution from degradation intermediates characteristics. The production, optimization, and Characterisation of bioflocculant QZ-7 produced by a novel Bacillus salmalaya strain 139SI which was isolated from a private farm soil in Selangor, Malaysia, were determined. Meanwhile, the optimal culture condition for bioflocculant production was achieved after cultivation at 35.5 °C for 72 h at pH 7, with an inoculum size of 5% (v/v) and sucrose, glucose as carbon source and yeast extract, urea as nitrogen sources. A bioflocculant yield of 2.72 g was recovered from 1 L of broth culture, with maximum flocculating activity that was found to be 92.6%. Chemical analysis revealed that the pure bioflocculant QZ-7 consisted of 79.08% carbohydrates and 15.68% proteins. Infrared spectrometry analysis showed the presence of carboxyl (COO-), hydroxyl (-OH), and amino (-NH3) groups, which are typically from polysaccharides and proteins. The NMR spectroscopy analysis confirmed the result of FTIR, through the presence of functional groups of the QZ-7. Scanning electron microscopy (SEM) analysis showed that QZ-7 exhibited a clear crystalline brickshaped structure. The average molecular weight of the bioflocculant QZ-7 was calculated to be 5.13×10^{5} Da. LC-MS analysis confirmed that QZ-7 was a glycoprotein compound detected at 741m/z-745m/z. Moreover, the presence of glucose at 182.96 m/z, rhamnose at 354.3m/z, and glucuronic acid at 212.8 m/z. SEM- EDX analysis indicated the existence of C, O, N, P and S in this macromolecule as 55.74%, 42.74%, 0.54%, 0.93%

and 0.06%, respectively. Thermogravimetric analysis (TGA) of the bioflocculant QZ-7 contained thermos-stable and thermo-labile molecules. Bioflocculant QZ-7 exhibited wide pH stability that ranged from 4 to 7, with a flocculation activity of more than 70%. In addition, QZ-7 was thermally stable and retained more than 80% of its flocculating efficiency after being heated at 60 °C for 30 min. The highest bioflocculating activity of 93.6% was obtained for Ca⁺² at 2 mg/L of QZ-7 concentration at pH 7. The treatment of river water by purified bioflocculant QZ-7 showed high performance in the removal of turbidity, total suspended solids and COD. After treating the wastewater, the bioflocculant QZ-7 showed significant flocculating performance with a COD removal efficiency of 93%, whereas a BOD removal efficiency of 92.4% was observed in the B. salmalaya strain 139SI. In addition, results for the removal of heavy metals from industrial wastewater revealed that the bioflocculant QZ-7 was capable of removing the heavy metals. For example, the maximum adsorption of As (89.8 %), and Zn⁺² (77.4 %), and Cu⁺² (58.4%). Moreover, the bioflocculant QZ-7 had significant removal efficiency of different pharmaceutical compounds, such as Simvastatin (92.45%), Salbutamol (88.69%), Acetaminophen (69%), and Caffeine (66.52%). Furthermore, B. salmalaya 139SI strain and pure bioflocculant QZ-7 could synthesise AgNPs. Also, an antibacterial activity of the AgNPs was detected against test bacterial strains, such as Escherichia coli ATCC35401, Salmonella enteritidis ATCCBAA-711, Staphylococcus aureus ATCC2592 and *Pseudomonas aeruginosa*, as application of AgNPs.

Keywords: *Bacillus salmalaya* 139SI, Optimization, Production, Bioflocculant, Characterisation, Application.

PENGECAMAN DAN PENILAIAN BIOFLUKULAN DARI Bacillus salmalaya

139SI DAN APLIKASINYA DALAM RAWATAN AIR KUMBAHAN

ABSTRAK

Gumpalan bioflok adalah bahan yang dihasilkan oleh mikroorganisma semasa berlakunya proses tumbesaran dan sejak akhir ini telah mendapat perhatian khusus daripada para penyelidik berdasarkan sifat boleh urai, tidak toksik, dan tidak menyebabkan pencemaran kedua berlaku disebabkan proses penguraian. Penghasilan, pengoptimum, dan ciri gumpalan QZ-7 ditentukan oleh bakteria novel Bacillus salmalaya strain 139SI diambil dari tanah ladang milik persendirian di Selangor, Malaysia. Keadaan kultur optimum penghasilan gumpalan dicapai pada pH 7 \pm 0.2, dengan saiz inokulum adalah 5% (v/v) dan sukrosa, glukosa sebagai sumber karbon dan ekstrak yis, urea sebagai sumber nitrogen selepas pengkulturan pada suhu 35.5 °C selama 72 jam. Hasil gumpalan adalah 2.72 g telah diperoleh daripada 1 L kultur broth, dengan kadar maksimum aktiviti gumpalan iaitu 92.6%. Analisa kimia mendedahkan bahawa gumpalan asli QZ-7 mengandungi 79.08% karbohidrat dan 15.68% protein. Analisa meggunakan spektrometri inframerah menunjukkan kehadiran kumpulan karboksi (COO-), hidroksi (-OH), dan amino (-NH3); yang mana tergolong dalam kumpulan polisakarida dan protein. Analisa NMR menggunakan hasil dari FTIR melalui kehadiran kelpelbagaian kumpulan QZ-7. Analisa menggunakan mikroskop electron (SEM) menunjukkan bahawa QZ-7 mempamerkan struktur berbentuk bata jernih. The average molecular weight of the bioflocculant QZ-7 was calculated to be 5.13×10⁵Da. LC-MS analisa membuktikan biofluculant QZ-7 adalah kampaun glycoprotein dikesan pada 741m/z -745 m/z. manakala kehadiran glucose olikasan pada 203.1m/z, rhamnose 354.3m/z. olan glucuronic acid 189.1 m/z. Analisa menggunakan SEM-EDX pula menunjukkan kewujudan C, O, N, P dan S dalam makromolekul ini masing-masing adalah sebanyak 55.74%, 42.74%, 0.54%, 0.93% dan 0.06%. Analisa termogravimetrik (TGA) ke atas gumpalan QZ-7 menunjukkan QZ-7 adalah molekul yang stabil haba dan ubahsuai haba. Gumpalan QZ-7 mempunyai kadar kestabilan pH adalah daripada skala 4 hingga 7, dengan 85% aktiviti gumpalan pada pH 7. Tambahan lagi, QZ-7 mempunyai haba yang stabil dan dapat dikekalkan melebihi 80% kecekapan gumpalan selepas dipanaskan pada suhu 80 °C selama 30 minit. Aktiviti gumpalan paling tinggi (93.6%) dapat dicapai oleh Ca⁺² pada pH 7.0. Rawatan air sungai yang dilakukan oleh gumpalan QZ-7 menunjukkan kecekapan yang tinggi dalam mengurangkan kekeruhan, pepejal yang tergantung dan COD. Selepas merawat air sisa, gumpalan QZ-7 menunjukkan signifikasi antara kecekapan gumpalan dengan pengurangan COD sebanyak 93% dengan pengurangan BOD dalam bakteria B. salmalaya strain 139SI pula adalah sebanyak 92.4%. Nilai-nilai tersebut menunjukkan bahawa kecekapan aplikasi gumpalan QZ-7 dalam rawatan air sisa. Selain itu, keputusan mendedahkan bahawa gumpalan QZ-7 mempunyai kemampuan untuk menghapuskan logam berat. Di samping itu, hasil penyingkiran logam berat dari air sisa industri menunjukkan bahawa bioflocculant QZ-7 mampu mengeluarkan logam berat. Sebagai contoh, penjerapan maksimum As (89.8%), dan Zn⁺² (77.4%), dan Cu⁺² (58.4%). Selain itu, gumpalan juga menunjukkan kecekapan dalam menghapuskan partikel farmaseutikel seperti Simvastatin 92.45%, Salbutamol 88.69%, Acetaminophen 69%, Caffeine 66.52%. Tambahan lagi, gumpalan QZ-7 boleh mengsintesis AgNPs dan disalut dengan biopolymer yang akan menghasilkan daya tujahan antara partikel untuk menjauhkan diri antara satu sama lain dan dapat mengelakkan daripada perlekatan. Juga, didapati bahawa aplikasi sintesis AgNPs oleh gumpalan mampu melawan strain bakteria seperti Escherichia coli ATCC35401, Salmonella enteritidis ATCCBAA-711, Staphylococcus aureus ATCC2592 and Pseudomonas aeruginosa.

Kata Kunci: Bacillus salmalaya 139SI, Pengoptimuman, Pengeluaran, Bioflocculant, Pencirian, Permohonan.

ACKNOWLEDGEMENTS

In the name of Allah, the most Gracious and the most Merciful Praise to Allah S.W.T, for the gift of life, strength, health, patience and providing me this opportunity and granting me the capability to proceed successfully. This research has been extremely fulfilling, not because of how much I learned, but especially because it may eventually to the protection of the environment and contribution in the development of better strategies for approaching environmental contamination.

First and foremost, I would like to express my gratitude to Prof. Dr. Salmah Ismail for her supervision, guidance, stimulating suggestions and encouragement and support in various ways. Her truly scientist insight has made her as a constant oasis of ideas and passion in science, which exceptionally inspire and enrich my growth as a student, a researcher and a scientist want to be. I am indebted to her more than she knows.

Secondly, I would like to thank everyone in general who had directly or indirectly helped me throughout the study especially my friends and colleagues from the Institute of Biological science, University of Malaya, and Al-zhar University-Gaza, Palestine.

This thesis is dedicated to my family (late parents, and late brother sobhy) whose dreams for me have resulted in this achievement. My deepest appreciation and love go to my wife Annel and sons, daughters (Fatima, Eman, Mohammed, Mamoon, Hala, Qussay and Haya) for their patient, emotional support, and unwavering faith given to me in the journey towards completing this doctoral degree.

Finally, this work is also dedicated to my brothers and sisters for their endless encouragement and supports in different ways.

TABLE OF CONTENTS

ABS	STRAC	Γ	. iii
ABS	STRAK.		v
ACI	KNOWI	LEDGEMENTS	vii
TAF	BLE OF	CONTENTS	. viii
LIS	F OF F	IGURES	. xvi
LIS	ГOFT	ABLES	, xix
LIS	Г OF SY	YMBOLS AND ABBREVIATIONS	xxii
LIS	Г OF A	PPENDICES	xxiv
CHA	APTER	1: INTRODUCTION	1
1.1	Genera	al Overview	1
1.2	Resear	ch Problems	7
1.3	Hypot	hesis	9
1.4	Scope	of Study	9
1.5	Object	ives of Study	10
1.6	Signifi	icance of Study	10
1.7	Thesis	Structure	11
CHA	APTER	2: LITERATURE REVIEW	12
2.1	Introdu	uction	12
2.2	Floccu	llation Process in Water Treatment	14
2.3	Classif	fication of Flocculants	15
	2.3.1	Organic Flocculants	16
	2.3.2	Inorganics Flocculants	. 17
	2.3.3	Naturally Occurring Flocculants	19

		2.3.3.1 Chitosan	20
		2.3.3.2 Sodium Alginate	21
		2.3.3.3 Tannin	21
		2.3.3.4 Cellulose	22
		2.3.3.5 Bioflocculant	23
2.4	Compo	sition Analysis and Flocculating Potential of some Bioflocculants	25
2.5	The Fu	nctional Groups in Bioflocculant Molecule	28
2.6	Biofloc	culation Mechanisms	29
	2.6.1	Charge Neutralization	29
	2.6.2	Polymer Bridging	30
2.7	Factors	Affecting Bioflocculants Production	44
	2.7.1	Effect of Carbon and Nitrogen Sources	32
	2.7.2	Effect of Temperature	41
	2.7.3	Effect of Inoculum Size	41
	2.6.4	Effect of Initial pH	42
	2.7.5	Effect of Agitation Speed on Bioflocculant Production	43
	2.7.6	Effect of Metals on the Bioflocculant Production	44
2.8	Factors	Affecting Flocculation Efficiency of Bioflocculants	44
	2.8.1	Effect of pH on Flocculating Efficiency	44
	2.8.2	Effect of Bioflocculant Concentration on Flocculation Efficiency	46
	2.8.3	Effect of Temperature on Flocculating Efficiency	47
	2.8.4	Effect of Metal Ions on Flocculating Efficiency	48
2.9	Cost-E	ffective Substrates for Bioflocculant Production	50
	2.9.1	Molasses as Substrate	51
	2.9.2	Palm Oil Wastewater	52
2.10	Applica	ation of Bioflocculant	53
	2.10.1	Treatment of Wastewater	54

		2.10.1.1 Removal of COD and Suspended Solids from Wastewater	54				
		2.10.1.2 Drug Removal	55				
		2.10.1.3 Heavy Metal Removal	58				
2.11	Synthe	esis of Nanoparticles	62				
2.12	Conclu	usion and Future Prospects	63				
CHA	APTER	3: MATERIALS AND METHODS	65				
3.1	Experi	mental Design and Sampling Sites	65				
3.2	Chemi	cals and Reagents	67				
3.3	Cultiva	ation and Isolation of the Bacteria	68				
3.4	Screen	ing for Bioflocculant-Producing Bacteria	68				
3.5	Detern	nination of Flocculating Activity	69				
3.6	Optim	Optimization of Cultural Conditions of B. salmalaya 139SI for					
	Bioflo	Bioflocculant Production					
	3.6.1	Effect of Temperature, Static Conditions and Shaking Speed on Bioflocculant Production	70				
	3.6.2	Effect of pH	70				
	3.6.3	Effect of Inoculum Size	71				
	3.6.4	Effect of Cultivation Time on Bioflocculant Production	71				
	3.6.5	Effect of Carbon and Nitrogen Sources on Bioflocculant Production	71				
	3.6.6	Effect of Various Salts on Bioflocculant Production	72				
	3.6.7	Time Course Assay for Bioflocculant Production by <i>B. Salmalaya</i> 139SI	72				
	3.6.8	Extraction and Purification of Bioflocculant	73				
3.7	Analys	sis and Characterisation for Purified Bioflocculant QZ-7	73				
	3.7.1	Phenol-Sulfuric Acid Method for Total Carbohydrates	73				
	3.7.2	Bradford Method for Total Protein	74				
	3.7.3	Carbazole Assay for Uronic Acid	74				

	3.7.4	Fourier-Transform Infrared Spectroscopy (FTIR)	. 75
	3.7.5	Nuclear Magnetic Resonance (NMR)	. 75
	3.7.6	Determination Molecular Weight of Purified Bioflocculant QZ-7	. 76
	3.7.7	Liquid Chromatography Mass Spectrometry (LC/MS)	. 76
	3.7.8	Scanning Electron Microscopy (SEM)	. 76
	3.7.9	Energy Dispersion X-ray (EDX) Spectroscopy	. 77
	3.7.10	Thermogravimetric Analysis (TGA)	. 77
3.8	Factors Biofloc	Affecting the Flocculation Performance of Purified culant QZ-7	. 77
	3.8.1	Effect of Bioflocculant QZ-7 Concentration on the Flocculation Efficiency	. 77
	3.8.2	Effect of pH on the Flocculating Efficiency of Purified Bioflocculant QZ-7	. 78
	3.8.3	Effect of Temperature on the Flocculation Efficiency of Bioflocculant QZ-7	. 78
	3.8.4	Effect of Cations on the Flocculating Efficiency of Bioflocculant QZ-7	. 79
3.9	Applica	ations of Bioflocculant QZ-7	. 79
	3.9.1	Surface Water Treatment	. 80
	3.9.2	Removal of COD from Wastewater with Bioflocculant QZ-7	. 80
	3.9.3	Removal of BOD from Wastewater with <i>B. salmalaya</i> strain 139SI	. 81
	3.9.4	Heavy Metal Adsorption from Aqueous Solution	. 81
	3.9.5	Removal of Heavy Metals from Wastewater using Bioflocculant QZ-7	. 82
	3.9.6	Removal of Pharmaceuticals from Wastewater using Bioflocculant QZ-7	. 83
		3.9.6.1 Preparation of Standard Solutions	. 83
		3.9.6.2 Sample Collection and Preparation	. 83
		3.9.6.3 Solid-Phase Extraction (SPE)	. 83
		3.9.6.4 UPLC-ESI-MS/MS Analysis	. 84

		3.9.6.5 Mass Spectrometry	85
3.10) Statist	ical Analysis	85
3.1	1 Synthe	esis of Silver Nanoparticles (AgNPs) by using Bioflocculant	86
	3.11.1	In Nutrient Broth Medium	86
	3.11.2	Using Cell-Free Supernatant	86
	3.11.3	Using Purified Bioflocculant QZ-7	87
3.12	2 Charao	cterisation of Silver Nanoparticles (AgNPs)	87
		3.12.1.1 UV-vis Spectrophotometry	87
		3.12.1.2 Fourier-Transform Infrared Spectroscopy (FT-IR)	87
		3.12.1.3 X-ray Diffractometry (XRD)	87
		3.12.1.4 <i>Field Emission</i> Scanning Electron Microscopy (FE-SEM) and EDAX <i>Analysis</i>	88
3.1	3 Antiba	acterial Activity	88
СН	APTER	4: RESULTS	90
4.1	Introd	uction	90
4.2	Select	ion of Bacterial Strain for Bioflocculant Production	90
	4.2.1	Morphological Identification	90
	4.2.2	Microscopic Examination	91
4.3	Screen	ing Strain for Bioflocculant Production	91
4.4	Optim	ization of Bioflocculant Production by <i>B. salmalaya</i> 139SI	92
	4.4.1	Effect of pH on Bioflocculant Production	92
	4.4.2	Effect of Inoculum Size on Bioflocculant Production	92
	4.4.3	Effect of Temperature on Bioflocculant Production	93
	4.4.4	Effect of Shaking Speed on Bioflocculant Production	94
	4.4.5	Effect of Static Condition on Bioflocculant Production	95
	4.4.6	Effect of Carbon Source on Bioflocculant Production	96
	4.4.7	Effect of Nitrogen Sources on Bioflocculant Production	

	4.4.8	Effect of Different Inorganic Salts						
4.5	Time Course of Bioflocculant Production by <i>B. salmalaya</i> 139SI							
4.6	Charac	haracterisations of Bioflocculant QZ-7 100						
	4.6.1	Qualitative Chemical Analysis of the Bioflocculant QZ-7 100						
	4.6.2	Quantitative Chemical Analysis of the Bioflocculant QZ-7 101						
	4.6.3	Fourier-Transform Infrared Spectroscopy (FTIR)102						
	4.6.4	Molecular Weight Analysis						
	4.6.5	H-NMR Spectrum						
	4.6.6	Liquid Chromatography Mass Spectrometry (LC-MS) 105						
	4.6.7	Scanning Electron Microscopy (SEM) Imaging 106						
	4.6.8	Energy Dispersive X-ray (SEM- EDX) 107						
	4.6.9	Thermogravimetric (TGA) Determination of the Bioflocculant QZ-7						
4.7	Factors	s Affecting of Bioflocculant QZ-7 Performance 108						
	4.7.1	Effect of pH on the Flocculation Efficiency of QZ-7 108						
	4.7.2	Effect of Temperature on the Flocculation Efficiency of QZ-7 109						
	4.7.3	Effect of Bioflocculant QZ-7 Concentration on the Flocculation Efficiency						
	4.7.4	Effect of Cations on the Flocculation Efficiency of QZ-7 111						
	4.7.5	Effect of Salinity on Flocculation Efficiency of QZ-7 112						
4.8	Applic	ation of Bioflocculant QZ-7113						
	4.8.1	Surface Water Treatment by Bioflocculant QZ-7114						
		4.8.1.1 Removal of Turbidity						
		4.8.1.2 Total Suspended Solid Treatment						
		4.8.1.3 Chemical Oxygen Demand (COD) Removal 117						
	4.8.2	Removal of Heavy Metals using Bioflocculant118						
		4.8.2.1 Effect of pH on the Heavy Metal Adsorption						
		4.8.2.2 Effect of the Bioflocculant QZ-7 Concentration on HeavyMetals Adsorption						

		4.8.2.3	Effect of Heavy Metals Concentration on the
			Adsorption using of Bioflocculant QZ-7123
	4.8.3	Treatme	nt of Real Wastewater 126
		4.8.3.1	Removal of Heavy Metals from Industrial Wastewater by Bioflocculant QZ-7
		4.8.3.2	Removal of BOD and COD from Industrial Wastewater 127
		4.8.3.3	Removal of Pharmaceutical Compounds using Bioflocculant QZ-7
4.9	Synthes	sis of Nan	oparticle
	4.9.1	Nutrient	Broth Medium
	4.9.2	Using Co	ell- Free Supernatant 129
	4.9.3	Synthesi	s of AgNPs using Purified Bioflocculant QZ-7130
4.10	Charac	terisation	of AgNPs130
	4.10.1	Characte	erisation of Ag NPs by UV-Vis Spectroscopy 130
	4.10.2	Characte	erisation of AgNPs by FTIR131
	4.10.3	FE-SEM	and EDAX Analysis
	4.10.4	Characte	erisation of AgNPs by XRD Analysis133
4.11	Antibad	cterial Act	tivity of AgNPs Synthesized by Bioflocculant QZ-7134
СНА	PTER :	5: DISCU	JSSION
5.1	Selectio	on of Bact	terial Strain for Bioflocculant Production135
5.2	Optimi	zation of]	Bioflocculant Production by <i>B. salmalaya</i> 139SI135
5.3	Time C	ourse for	Bioflocculant Production by <i>B. salmalaya</i> Strain 139SI 141
5.4	Charact	teristics o	f Bioflocculant QZ-7143
5.5	Factors	Affecting	g of Bioflocculant QZ-7 Performance 148
5.6	Applica	ation of B	ioflocculant QZ-7151
	5.6.1	Surface	Water Treatment by Bioflocculant QZ-7152

	5.6.2	Removal of Heavy Metals Aqueous Solution using Bioflocculant QZ-7	154
	5.6.3	Removal of Heavy Metal from Industrial Wastewater	150
	5.6.4	Removal of BOD and COD from Industrial Wastewater	15′
5.7	Bioflo	cculant QZ-7 Mediated Synthesis of AgNPs, Characterisation, and	
	Antiba	cterial Activity	158
	5.7.1	Synthesis of AgNPs	159
	5.7.2	Characterisation of AgNPs	160
	5.7.3	Antibacterial Activity of AgNPs Synthesised by Purified Bioflocculant QZ-7	163
СН А 6.1	APTER Summ	6: CONCLUSION AND RECOMMENDATIONS	 16 4 164
6.2	Recom	mendation for Future Research	166
R	REFERE	ENCES	167
R L	REFERF AST OF	ENCES	167 195

LIST OF FIGURES

Figure 2.1	:	Classes of Flocculants	16
Figure 2.2	:	Charge Neutralisation Flocculation Mechanism (Reproduced with Permission from the Library, Bhavnagr University, India)	30
Figure 2.3	:	Bioflocculant Bridging Flocculation Mechanism (Reproduced with Permission from the Library, Bhavnagr University, India)	31
Figure 2.4	:	Representation for Source and the Fate of Pharmaceutical in the Environment (Reproduced with Permission from Elsevire)	56
Figure 3.1	:	Overall Experimental Design	66
Figure 4.1	:	 a. Front View of Pure Colony of <i>B. salmalaya</i> 139SI Strain; b. Back View of β-haemolytic <i>B. salmalaya</i> 139SI Activity 	90
Figure 4.2	:	Gram Stain of B. salmalaya 139SI	91
Figure 4.3	:	Effect of pH on Bioflocculant Production	92
Figure 4.4	:	Inoculum Size Effect on Bioflocculant Production	93
Figure 4.5	:	Effect of Cultivation Temperature on Bioflocculant Production	94
Figure 4.6	:	Effect of Shaking Speed on the Bioflocculant Production	95
Figure 4.7	:	Effect of Static Condition on the Bioflocculant Production	96
Figure 4.8	:	Carbon Sources Effect on Bioflocculant Production	97
Figure 4.9	:	Effect of Nitrogen Sources on Bioflocculant Production	98
Figure 4.10	:	Time Course of Bioflocculant Production by <i>Bacillus</i> salmalaya 139SI	100
Figure 4.11	:	Determination of Carbohydrate Concentration	101
Figure 4.12	:	Determination of Protein Concentration	102
Figure 4.13	:	Fourier - Transform Infrared Spectroscopy Analysis of Bioflocculant QZ-7	102

Figure 4.14	:	Gel Permeation Chromatography of the Purified Bioflocculant QZ-7	103
Figure 4.15	:	H ⁺ NMR Spectra of Purified Bioflocculant QZ-7	104
Figure 4.16	:	C ¹³ NMR Spectra of Purified Bioflocculant	105
Figure 4.17	:	ESI Spectra of LC/MS Analyses of Bioflocculant QZ-7 Showing Glycoprotein (A), Na^+ , H^+ and NH_4^+ Adducts of Glucose (B), Glucuronic Acid (C) and Rhamnose (D)	106
Figure 4.18	:	SEM Micrograph of Bioflocculant QZ-7 of <i>B. salmalaya</i> Strain 139SI. A. Purified Bioflocculant QZ-7. B. Bioflocculant Aggregation with Kaolin Clay	107
Figure 4.19	:	EDX Analysis of Bioflocculant QZ-7	107
Figure 4.20	:	Thermogravimetric Analysis of the Purified Bioflocculant QZ-7	108
Figure 4.21	:	pH Stability of the Purified Bioflocculant QZ-7	109
Figure 4.22	:	Thermal Stability of the Purified Bioflocculant QZ-7	110
Figure 4.23	:	Bioflocculant QZ-7 Dosage Effect on Flocculating Activity	111
Figure 4.24	:	Effect of Metal Ions on Flocculating Activity of Purified Bioflocculant	112
Figure 4.25	:	Effect of Salinity on Flocculation Performance of Bioflocculant QZ-7	113
Figure 4.26	:	Effect of Bioflocculant Concentration on Turbidity Removal	115
Figure 4.27	:	Effect of Bioflocculant Concentration on TSS Removal	116
Figure 4.28	:	Effect of Bioflocculant Concentration on COD Reduction	117
Figure 4.29	:	Effect of pH on Heavy Metal Adsorption Using 20 mg/L of QZ-7	119
Figure 4.30	:	Effect of Bioflocculant QZ-7 Concentration on the Removal Efficiency of Heavy Metals	121
Figure 4.31	:	Effect of Metal Concentration on the Efficiency of Removal Heavy Metals Using 60 mg/L Bioflocculant	124
Figure 4.32	:	Removal Efficiency of Heavy Metals using Different Concentrations of Bioflocculant QZ-7	127

]	Figure 4.33	:	Synthesis of AgNPs in Nutrient Broth Inoculated with <i>B. salmalaya</i> 139SI	129
]	Figure 4.34	:	Synthesis of AgNPs using Cell-Free Supernatant	129
]	Figure 4.35	:	Synthesis of AgNPs by Purified QZ-7	130
]	Figure 4.36	:	UV-vis Spectra of Synthesized AgNPs	131
]	Figure 4.37	:	FTIR Spectrum of Synthesized AgNPs	132
]	Figure 4.38	:	FE-SEM (a) and EDAX (b) Analysis of Synthesized AgNPs	132
]	Figure 4.39	:	XRD Patterns for the AgNPs Synthesized by Pure Bioflocculant QZ-7	133
]	Figure 4.40	:	Antibacterial Activity of AgNPs Against E. coli ATCC35401, Pseudomonase aeruginosa, Salmonella entertidis ATCC BAA-711 and Staphylococcus aureus ATCC259	134

LIST OF TABLES

Table 2.1	:	Optimum Culture Conditions, Chemical Compositions, Flocculating Activity and Yields of some Bioflocculant	36
Table 3.1	:	Sampling Sites and their GPS Locations	65
Table 3.2	:	Analytical Methods for Major Parameters using Standard Methods	79
Table 4.1	:	Flocculating Activity Values for the Selected Strains	91
Table 4.2	:	Analysis of Variances for the Effect of pH on Bioflocculant Production	92
Table 4.3	:	Analysis of Variances for the Inoculum Size Effect on Bioflocculant Production	93
Table 4.4	:	Analysis of Variances of Different Cultivation Temperature on Bioflocculant Production	94
Table 4.5	:	Analysis of Variances for Effect of Shaking Speed on Bioflocculant Production	95
Table 4.6	:	Analysis of Variances for the Effect of Static Condition on Bioflocculant Production	96
Table 4.7	:	Analysis of Variance for the Effect of Carbon Source on Bioflocculant Production	97
Table 4.8	:	Analysis of Variance for the Effect of Nitrogen Sources on Bioflocculant Production	98
Table 4.9	÷	Effect of Inorganic Salts on the Bioflocculant Production	99
Table 4.10	:	Analysis of Variance for the Effect of Inorganic Salts on the Bioflocculant Production	99
Table 4.11	:	Analysis of Variance for the Time Course of Bioflocculant Production	100
Table 4.12	:	Qualitative Analysis of Bioflocculant QZ-7	101
Table 4.13	:	Analysis of Variance for the Effect of pH on Flocculating Performance	109
Table 4.14	:	Analysis of Variance for the Effect of Temperature on Bioflocculant Performance.	110

Table 4.15	:	Analysis of Variance for the Effect of Bioflocculant QZ-7 Concentration	111
Table 4.16	:	Analysis of Variance for the Effect of Cations on Flocculating Activity	112
Table 4.17	:	Analysis of Variance for the Effect of Salinity on Flocculation Performance	113
Table 4.18	:	The Average Water Quality Characteristic of Langat River	114
Table 4.19	:	Analysis of Variance for the Effect of Pure Bioflocculant QZ- 7 on Turbidity Removal	115
Table 4.20	:	Analysis of Variance for the Effect of Crude Bioflocculant QZ-7 on Turbidity Removal	115
Table 4.21	:	Analysis of Variance For the Effect of Pure Bioflocculant QZ- 7 on TSS Removal	116
Table 4.22	:	Analysis of Variance for the Effect of Crude Bioflocculant QZ-7 on TSS Removal	117
Table 4.23	:	Analysis of Variance for the Effect of Pure Bioflocculant QZ- 7 on COD Removal	118
Table 4.24	:	Analysis of Variance for the Effect of Crude Bioflocculant QZ-7 on COD Removal	118
Table 4.25	:	Analysis of Variance for the Effect of pH on Arsenate Removal Efficiency	119
Table 4.26	:	Analysis of Variance for the Effect of pH on Cupper Removal Efficiency	119
Table 4.27	:	Analysis of Variance for the Effect of pH on Cadmium Removal Efficiency	120
Table 4.28	:	Analysis of Variance for the Effect of pH on Lead Removal Efficiency	120
Table 4.29	:	Analysis of Variance for the Effect of pH on Zinc Removal Efficiency	120
Table 4.30	:	Analysis of Variance for the Effect of pH on Mercury Removal Efficiency	121
Table 4.31	:	Analysis of Variance for the Effect of QZ-7 Concentration on Removal Efficiency of Arsenate	122

Table 4.32	:	Analysis of Variance for the Effect of QZ-7 Concentration on Removal Efficiency of Cupper	122
Table 4.33	:	Analysis of Variance for the Effect of QZ-7 Concentration on Removal Efficiency of Cadmium	122
Table 4.34	:	Analysis of Variance for the Effect of QZ-7 Concentration on Removal Efficiency of Lead	123
Table 4.35	:	Analysis of Variance for the Effect of QZ-7 Concentration on Removal Efficiency of Zinc	123
Table 4.36	:	Analysis of Variance for the Effect Arsenate Concentration on QZ-7 Adsorption Efficiency	124
Table 4.37	:	Analysis of Variance for the Effect of Cupper Concentration on QZ-7 Adsorption Efficiency	125
Table 4.38	:	Analysis of Variance for the Effect of Cadmium Concentration on QZ-7 Adsorption Efficiency	125
Table 4.39	:	Analysis of Variance for the Effect of Lead Concentration on QZ-7 Adsorption Efficiency	125
Table 4.40	:	Analysis of Variance for the Effect Zinc Concentration on QZ-7 Adsorption Efficiency	126
Table 4.41	:	Removal Efficiency of Bioflocculant QZ-7 in Industrial Wastewater Treatment.	127
Table 4.42	:	Efficiency of Bioflocculant QZ-7 In Removal of Pharmaceutical Compounds.	128
Table 4.43	÷	Antimicrobial Activity of AgNPs in Solid Medium	133

LIST OF SYMBOLS AND ABBREVIATIONS

°C	:	Celsius
Da	:	Dalton
μg	:	Microgram
μL	:	Microliter
mg	:	milli gram
mM	:	mille Molar
М	:	Molar
PPT	:	Part Per Thousand
PPM	:	Parts Per Million
Pmol	:	Pico mol
AAS	:	Atomic Absorption Spectroscopy
AgNPs	:	Silver Nanoparticles
BOD	:	Biological Oxygen Demand
BSA	:	Bovine Serum Albumin
CHNS	:	Carbon, Hydrogen, Nitrogen, Sulfur
COD	:	Chemical Oxygen Demand
СРО	:	Crude Palm Oil
DOE	:	Design of Experiments
D_2O	:	Deuterium Oxide
EDA	:	Exploratory data analyse
EDXS	÷	Energy Dispersion X-Ray Spectroscopy
EPS	:	Extracellular Substance
e.g	:	For example
et al	:	And others
EWW	:	Efluent waste water
FFB	:	Fresh Fruit Bunches
FTIR	:	Fourier-Transform Infrared Spectroscopy
H-NMR	:	Proton Nuclear Magnetic Resonance
HPGPC	:	High Performance Gel Permeation Chromatography
HPLC	:	High Performance Liquid Chromatography
ICPMS	:	Inductive Coupled Plasma Mass Spectrometry
IWW	:	Influent Waste Water

LC/MS	:	Liquid Chromatography Mass Spectrometry
MPOB	:	Malaysian Palm Oil Board
NMR	:	Nuclear Magnetic Resonance
NOM	:	Natural Organic matter
NTU	:	Nephelometric Turbidity Units
OD	:	Optical Density
PAA	:	Polyacrylamide
PAC	:	Poly-aluminium Chloride
PDADMA	:	Poly-Diallyl-Dimethyl Ammonium Chloride
PDMDAA	:	Poly-Dimethyl-Allyl Ammonium Chloride
PFC	:	Poly-Ferric Chloride
POME	:	Palm Oil Mill Effluent
PS	:	Polysaccharide
rpm	:	Revolution Per Minute
SEM	:	Scanning Electron Microscopy
SPE	:	Solid- Phase Extraction
i.e	:	That is
TDS	:	Total Dissolved Solids
TEM	:	Transmission Electron Microscopy
TGA	:	Thermo Gravimetric Analysis
TSS	:	Total Suspended Solids
UV Vis	:	Ultraviolet Visible
XRD	÷	X-Ray Diffractometry

LIST OF APPENDICES

Appendix A	:	Screening the Best Strain of <i>B. Salmalaya</i> 139SI for Bioflocculant Production	198
Appendix B	:	Factors Affecting Bioflocculant Production by <i>B. Salmalaya</i> 139SI	199
Appendix C	:	Factors Affecting of Bioflocculant Performance	202
Appendix D	:	Application of Bioflocculant QZ-7 in Treatment of River Water, Industrial Wastewater and Removal Heavy Metals	204
Appendix E	:	Characterisation of the Silver Nanoparticles	206
Appendix F	:	Statistical Analysis	207

CHAPTER 1: INTRODUCTION

1.1 General Overview

Environmental pollution has become one of the major universal problems with its associated weakening effects on economic development (Chen *et al.*, 2018; Pathak *et al.*, 2014). Water pollution is one of the most thought-provoking environmental concerns and has become a global stumbling block for improving the quality of life in many communities (Okaiyeto *et al.*, 2016). Water is a source of life and energy, even though many societies worldwide are suffering from the scarcity of safe water for drinking purposes (Rani *et al.*, 2013).

Innocent development and rapid growing in population have enormously given rise to the risky condition of water pollution and primary harmful environment (Prasertsan *et al.*, 2006). The main sources of these pollutions come from toxic industrial discharge and domestic wastewater effluents, agricultural wastes and untreated sanitary wastewater (Li *et al.*, 2013). The existence of these contaminants in water bodies could be malicious to bio-aquatic life and has also reduced its incompatibility, such as drinkable water sources for domestic utilisation, and a consequence of pollution in aquatic environment is a life-threatening effect on healthful human life (Yang *et al.*, 2012). Water, soil, rice and wheat are contaminated with heavy metals by several sources, such as chemical plants, ceramic factories and plants of recycling materials, because they are heavily dispersed. Besides, advanced transportation and agriculture activities are also a crucial source of heavy metal contamination (Chen *et al.*, 2018).

Environmental pollution with all types of heavy metalhave conferred more consideration in current years because of their high toxicity, bioaccumulation and persistence (Chen *et al.*, 2018; Liu *et al.*, 2017). In recent years, heavy metal contamination has become progressively serious (Yu *et al.*, 2017; Ran *et al.*, 2016). The

essential assessment of health threats to the population is important. Therefore, any adversative effects can be circumvented to the maximum range even though the health threats of heavy metal have been vastly studied (Liu *et al.*, 2017; Cai *et al.*, 2015).

According to the WHO/UNICEF (2017) statement, about 70% to 80% of all diseases in developed countries are related to drinking polluted water, specifically among vulnerable people. When untreated wastewater is released into natural water bodies, it becomes poisonous to marine life and reduces the potable waters for utilisation. For instance, a consequence of these alarming water-borne diseases as well as an increase in the demand for non-toxic water for mutually municipal and industrial utilisation, more consideration was focused on water treatment. Therefore, it is very essential to evaluate water quality on a continuous basis (Yang *et al.*, 2012). Numerous stringent rules were initiated by many countries with respect to the occurrence of pollutants in water to safeguard appropriate treatment of industrial and domestic discharge as well as agricultural wastewater effluents, before their discharges into different water- bodies (Li *et al.*, 2013). Relatively, many water treatment technologies are available and these include coagulation/flocculation, filtration, ion exchange, oxidation processes, adsorption, solvent extraction and electrolysis (Low *et al.*, 2011; Ong *et al.*, 2012).

However, the major obstruction with some of these techniques is limited financial costs in their operations which tend to reduce their capability and effectiveness in application (Li *et al.*, 2013). On the other hand, coagulation/flocculation is the preferred process because it is less costly, effective, not labour-intensive, and requires less skilled personnel (Li *et al.*, 2013; Renault *et al.*, 2009). Moreover, one of the main advantages of the coagulation/flocculation method is the decolourisation of wastewater due to the elimination of dye molecules from the final discharges, and thus removing potential deterioration of dyes which can release toxic chemicals that contain aromatic compounds

(Aboulhassan *et al.*, 2006). Flocculants are naturally occurring organic macromolecules; they are substances with the ability to flocculate suspended solids, cells, and solid colloid particles (Bhunia *et al.*, 2012). Flocculants are widely applied for separation techniques, such as in drinking water purification, wastewater treatment, activated sludge dehydration, downstream processing, and food fermentation (Chen *et al.*, 2017). Shih and Van (2001), recognised that flocculation could be applied as a substitute to filtration and centrifugation for microbial cell separations from broth in food, pharmaceutical and beverage industries. Flocculation is an operative system that is normally applied in wastewater treatment to remove suspended solids and heavy metals (Deng *et al.*, 2003).

Flocculants are typically classified into three groups: organic synthetic flocculants such as polyethylene imine and polyacrylamide byproducts, inorganic flocculants including aluminium sulphate and polyaluminium chloride, and natural flocculants such as chitosan, sodium alginate and bioflocculants (Aljuboori et al., 2013). Organic and inorganic flocculants are referred as chemical flocculants. These chemicals/ synthetic flocculants are commonly applied to remove organic matter from different wastewater because of their high flocculating efficiency and low cost (Zheng et al., 2008). However, several neuropathological studies have showed a relation between residual aluminium in water and the pathogenesis of Alzheimer's disease and dementia (Banks et al., 2006). Ferrite flocculants are also good flocculating agents except for causing an unpleasant metallic taste, malodour and being highly corrosive in nature (Li et al., 2008). On the other hand, polyacrylamides are non-biodegradable, and the carcinogenicity property of the residual acrylamide monomers restricts their industrial applicability (Zhuang et al., 2012). A number of cationic polymers are documented to constitute an important potential hazard to aquatic life, specifically fish and other aquatic lives. Also, the fish gill surfaces have negative charge to which cationic polyelectrolytes can eagerly bind due to electrostatic attraction (Murgatroyd et al., 1996). This effect leads to blockage and

decrease in oxygen transport and produce mucous on the gill surfaces, thereby causing death of the fish which subsequently reduces the supply of healthy fish for human consumption (Narkis & Rebhun, 1975). Being outstanding to numerous problems that are related to the use of chemical flocculants, more attention is diverted to microbial flocculants (bioflocculants).

Bioflocculants are produced by many kinds of microorganisms, such as bacteria, algae and fungi which have been isolated from soil, water and activated sludge (Aljuboori et al., 2013; Czemierska et al., 2017; Tiwari et al., 2015). Different microorganisms (i.e. bacteria, algae, fungi, and actinomycetes) can consume the nutritious substance in the culture medium to synthesise intracellular high molecular weight bioflocculant which can be excreted into fermented broth or on the bacterial cell surface as capsules possessing flocculant properties (Desouky et al., 2008; Gao et al., 2006). Lately, the use of microbial flocculants was activated as a solution to environmental problems that develop from the use of chemical flocculants (Li et al., 2009). Bioflocculants are natural macromolecules produced by microorganisms through their growth, their use has received consideration due to their biodegradability, non-toxicity, safety and eco-friendly characteristics because they lack secondary pollution from intermediate degradation (Ntsangani et al., 2017; Xia et at., 2018). These macromolecules are composed of polysaccharides, protein, glycoproteins, lipids, nucleic acid, lectins and other polymers commonly present in lower concentrations (Giriet et al., 2015; Tang et al., 2014). Extracellular polymeric substance (EPS), are possessed primarily of sugars, protein, glycoprotein and nucleic acids. EPS can be used in processes that correspond to drinking water purification, wastewater treatment, food and fermentation processing as well as in pharmaceutical and cosmetic industries as stabilising, viscosifying and emulsifying agents (Czemierska et al., 2017; Zhong et al., 2018). Also, microbial polymers are important as antibacterial, antiviral, and anti-algal agents and as inducers of microbial accumulation and biofilm formation (Zhong et al., 2018). Distinct conventional flocculants, bioflocculants have the benefit of being biocompatible and environmentally friendly. Until now, they have been employed in several applications, for example, dye wastewater treatment, sludge dewatering and heavy metal removal (Wang et al., 2015; Xia et al., 2018). The flocculating efficiency of a bioflocculant depends uniquely on its chemical structure which can be related to high molecular weight and the types of functional groups present in the molecular chain of such biopolymers (Gao et al., 2006). For example, Zhong et al., (2018), found the novel bioflocculant MBF-15 which is an exo-polysaccharide produced by alkaliphilic, Paenibacillus jamilae. Czemierska et al. (2017) also, stated that the exopolymer from *Rhodococcus rhodochrous*, with a molecular weight of about 1.3×10^3 KDa, constitutes polysaccharide (62.86%) and protein (10.36%). Nwodo et al. (2014) and Li et al. (2010) found that the biopolymers secreted by both Agrobacterium sp. M-503, and a consortium of Cellulomonas and Streptomyces species were predominantly composed of glycoproteins. Meanwhile the bioflocculants synthesised by Bacillus sp. Gilbert (Ugbenyen et al., 2014) and Halomonas sp. AAD6 (Sam et al., 2011) were composed of polysaccharides.

However, the production of bioflocculants is usually subjective by culture media constituents, such as carbon, nitrogen sources, salts as well as factors like pH and temperature of the medium, and aeration level (Lopez *et al.*, 2003; Salehizadeh & Shojaosadati, 2001). Several researchers have been investigated on the synthesis of bioflocculants from different microorganisms by utilising complex media for growth and optimal yields (Piyo *et al.*, 2011). For example, Cosa *et al.*, (2011) found that glucose and peptone were utilised as carbon and nitrogen sources, respectively, with Fe⁺² as an optimal salt for the production of bioflocculants by *Virgibacillus* sp. Rob. *Serratia ficaria* utilises beef extract, urea, lactose and Mg⁺² or Ca⁺² for bioflocculant production (Gong *et al.*, 2008), whereas *Enterobacter cloacae* WD7 favoured sucrose, yeast extract and Ca⁺²

as the salt for bioflocculant production (Prasertsan et al., 2006). Complex media utilisation renders the fermentation process to be economically expensive due to high nutrient costs, such as glucose, sucrose, peptone, yeast extract and salts. There is a need to reduce fermentation costs, whereas increasing bioflocculant yields demands the need for an alternative, but cost-effective, substrates (Fujita et al., 2000). Agricultural industrial wastes, such as sugarcane molasses, fishmeal wastewater and corn-steep liquor could be utilised as an alternative substance for cost-effective production of bioflocculants (Zhuang et al., 2012). The utilisation of organic wastes as substrates in the production media is ecologically promising, sustainable and economically acceptable. Liu et al. (2016) reported that Pseudomonas veronii L918 can transform the hydrolysate of peanut hull into bioflocculants and a harvest of 3.39 g/L bioflocculant MBF-L918 was reached after peanut hull was utilised as a carbon source. Zhou et al. (2003) and Huang et al. (2001) reported on the utilisation of some cost-effective substrates, such as soybean juice and fishmeal wastewater for bioflocculant production. For example, finding by Kurane et al. (1994) showed that two-thirds of the cultivation cost could be saved when yeast extract in the growth medium was replaced by different supplements, such as aquafarm wastewater, bean cake or cattle blood for microbial production of (NOC-1) bioflocculant. The presence of nutrient constituents in brewery wastewater makes it a potential substrate for bioflocculant synthesis by certain microorganisms (Chen et al., 2003). Similarly, the brewery wastewater was used as a carbon source for bioflocculant production by multiple-microorganism consortia (Zhang et al., 2007). High flocculating activity and bioflocculant yields were also achieved when Serratia ficaria and Klebsiella mobilis were cultivated in the presence of dairy wastewater (Gong et al., 2008; Wang et al., 2007). Likewise, the synthesis of a novel bioflocculant REA-11(polygalacturonic acid) by Corynebacterium glutamicum was optimally gained when sucrose and corn steep liquor were used as substrate (He et al., 2004). In contrast, the production of an

exopolysaccharide bioflocculant by *Sorangium cellulosum* was observed in the presence of soluble starch, which is a cheaper substrate than glucose (Zhang *et al.*, 2002). Theconsumption of these cost-effective substrates for bioflocculant production creates a reasonable logic if production costs are to be contained. Furthermore, to reduce the production cost, different wastewaters were used as low-cost substrates, such as potato starch wastes as carbon source (Liu *et al.*, 2015b; Pu *et al.*, 2014), palm oil mill discharge (Aljuboori *et al.*, 2014; Wong *et al.*, 2012), dairy waste effluents (Wang *et al.*, 2007) and chromotropic acid wastes (Zhong *et al.*, 2014). In addition, agricultural wastes, such as corn stover, rice stover are rich in lignocelluloses, in which hydrolysates are applied as low-cost carbon source to produce bioflocculants (Guo *et al.*, 2015b; Wang *et al.*, 2013b). Bioflocculants are documented to have applications in wastewater treatment, for example, in the decolourisation of dyes in solutions (Deng *et al.*, 2005), inorganic solid suspensions e.g. aluminium oxide, bentonite, solid clay, and activated carbon (Yim *et al.*, 2007; Zhou *et al.*, 2003).

Currently, a significant progress is being made by green nanotechnology, in considering environmental benefits, time duration and cost efficiency. Biological synthesis of nanoparticles has a natural tendency to work on different scientific disciplines, including material science and biomedical application (Muthulakshmi *et al.*, 2017). The main advantages of the nanoparticles are that their increase in surface to volume ratio and biocompatibility. The most commonly used synthesis route for nanoparticles is by chemical and physical approach.

1.2 Research Problems

Industrial development has been established as a desirable choice, owing to its impact to economic growth. However, it has extensively increased the level of water contamination, particularly from industrial sources and this has become a crucial environmental concern (Sakar *et al.*, 2006). Water pollution is one of the greatest stimulating issues that is globally distressing the fine quality of life in many countries. The main sources of water pollution are the effluents of untreated sanitary, noxious industrial discharge and domestic and agriculture wastes (Li *et al.*, 2013). The occurrence of contaminated water bodies can be antagonistic to marine life as well as rendering it inappropriate for domestic usage. As a consequence, an increase in waterborne diseases becomes predictable, striking more for non-toxic water for both metropolitan and industrial demands. Therefore, great consideration has been focused on water treatment; consequently, making it essential to estimate water quality on a constant base (Yang *et al.*, 2012).

Chemical flocculants are widely involved in drinking water and wastewater treatment, food and fermentation industries and also downstream processing due to their costeffectiveness and high flocculation competence (You *et al.*, 2008). However, their wide practises have raised significant environmental and health concerns (Mabinya *et al.*, 2012). The use of aluminium as the coagulant in water treatment could cause a higher concentration of aluminium in the treated discharge than in raw water. Besides, aluminium residual in immoderate sludge formed throughout coagulation tends to accumulate in the environment (Ma *et al.*, 2016). For example, a number of studies have presented that aluminium salts are related to Alzheimer's disease (Bank *et al.*, 2006; Arezoo, 2002). In addition, polyacrylamide displays high flocculation activity, while the acrylamide monomer residuals are carcinogenic and neurotoxic to humans (Ruden, 2004; Polizzi *et al.*, 2002). Moreover, acrylamides are not biodegradable and consequently represent an environmental irritant (Lofrano *et al.*, 2013).These expected problems that are associated with chemical flocculants have imposed a search for different flocculants that are eco-friendly and non-toxic (Nwodo *et al.*, 2013). On the other hand, bioflocculants are harmless, biodegradable, neither neurotoxic nor carcinogenic as compared to the synthetic flocculants used in many industrial processes (Li *et al.*, 2009; Liu *et al.*, 2010). However, the high costs associated with their production as well as the corresponding low yields are still the major problems that prevent their industrial usage (Gao *et al.*, 2006; Wang *et al.*, 2007). Therefore, there is a need to screen new microorganisms with high bioflocculant production capacities and also develop several strategies to optimise culture conditions in enhancing bioflocculant yields with improved flocculating activity (Li *et al.*, 2009; Ugbenyen *et al.*, 2012). Biosynthesis of silver nanoparticles by using bioflocculant is used in deactivating *Acidithiobacillus* bacteria which are involved in the generation of acid mine drainage (Natarajan. 2017).

1.3 Hypothesis

It is hypothesised that the bacterial bioflocculant produced by *Bacillus. salmalaya* 139SI can significantly reduce the COD, BOD, TSS and turbidity from surface wastewater and removing heavy metals and pharmaceuticals from wastewater effluent and nanoparticles synthesis.

1.4 Scope of Study

Different flocculants, such as organic high-polymer flocculants, inorganic flocculants and naturally occurring flocculants are applied in wastewater, industrial downstream processes and dredging. Meanwhile organic high-polymer is very effective, a few of them are not simply degraded in nature and some of the monomers obtained from synthetic polymers are deleterious to the human body. In recent years, to resolve these environmental problems, the use of microbial bioflocculants have been anticipated due to their biodegradability, non-toxicity, safe and environmentally friendly, including lack of secondary pollution from intermediate degradations (Ntsangani *et al.*, 2017; Xiang *et at.*, 2018).

1.5 **Objectives of Study**

The aim of this study is to produce a new bioflocculant from *Bacillus salmalaya* 139SI and apply it onsite in water and wastewater treatment. The specific objectives of this study include:

- To determine the optimal medium compositions and cultural conditions for the production of bioflocculant by *B. salmalaya* 139SI strain to enhance the bioflocculant yield.
- 2. To extract, purify and characterise the bioflocculant compound produced by *B*. *salmalaya* 139SI.
- 3. To investigate the potential of bioflocculants produced by *B. salmalaya* 139SI in surface water and industrial wastewater treatment.
- 4. To determine the performance of the bioflocculant in synthesising nanoparticles.

1.6 Significance of Study

The importance and significance of study are:

- 1. Contribution in the investigation of *B. salmalaya* 139SI potential as a novel bioflocculant source and its application in water and wastewater treatment.
- 2. The test the common chemical flocculant substitute by using a new bioflocculant in water treatment to avoid toxic chemical sludge, achieve a lower treatment cost and green environment.
- 3. The empirical of synthesised nanoparticles as an antibacterial agent.

1.7 Thesis Structure

The thesis focuses on the production of a bioflocculant from a selected bacterium, which is *Bacillus salmalaya* strain 139SI, with flocculating potentiality. The scope of the thesis includes factors which affect the production of bioflocculant, extraction, purification and Characterisation of bioflocculant, as well as factors affecting bioflocculant performances and applications.

The following section depicts the flow and outline of the thesis.

Chapter 2 contains a general introduction to different types of flocculants and a review of relevant literature on bioflocculant production, Characterisation of the bioflocculant, bioflocculant performance and applications.

Chapter 3 explains the general material and methods, whereby the experimental research is described and discussed in detail.

Chapter 4 presents the results of the factors affecting the *Bacillus salmalaya* 139SI on bioflocculant production, Characterisation of the bioflocculant and factors affecting bioflocculant performance and bioflocculant applications in river water and wastewater, heavy metal removals and nanoparticle synthesis.

Chapter 5 presents on the overall results and discussions.

Chapter 6 includes the research summary and conclusion with some recommendations for using different methods in water and wastewater treatment and future studies.
CHAPTER 2: LITERATURE REVIEW

2.1 Introduction

Implications for public health have demand alternatives to inorganic and synthetic flocculants. Bioflocculants have been chosen as important candidates. Water is one of the vital sources of life and energy, even though many societies worldwide are suffering from the scarcity of safe water for drinking purposes (Okaiyeto et al., 2016). The bountiful capacity of water on Earth is the sole exclusive factor that distinguishes this planet from others. The significance of water to the existence of natural life is vital for human survival so that the search for water on other planets have grown to be the basic factor to recommend the manifestation of life (Rani et al., 2013). Water occupies 78% of the planet's surface, and until now available water for human use is inadequate. In fact, it is an essential need for human development, health and welfare. Safe drinking water is a globally accepted human right (WHO, 2001) which is registered as one of the 10 targets in the Millennium Development Goals (MDGs), (Rout & Sharma, 2011). The property of water used by humans in certain communities could be considered as an important indicator of the human lifespan superiority surrounded by that environment. Water is extremely necessary for municipal, industrial, agricultural and environmental needs (Okaiyeto et al., 2016). Still, pollutants in water will decrease its effectiveness in utilisation because the undesirable effect of water contamination is threatening humans and the environment.

Water pollution is one of the major thought- provoking environmental concerns and has become an international stumbling block for improving the quality of life in many communities (Chen *et al.*, 2018). Unintended development and rapid growth in population have massively given rise to water contamination risks and prevalent harmful environments (Prasertsan *et al.*, 2006; Okaiyeto *et al.*, 2016). The main sources of water

contamination are the effluent of agricultural and domestic waste, industrial toxic waste and untreated sanitary discharge (Li *et al.*, 2013). The occurrence of these contaminants in water bodies could remain malicious to aquatic biota lifespan and has reduced their inappropriate as drinkable water sources for domestic utilisation, and the consequent of aquatic environment pollution has a serious influence on healthful human life (Yang *et al.*, 2012).

According to WHO/UNICEF (2017), report around 70%–80% of all diseases in developed nation states are related to the consumption of polluted water, particularly amongst susceptible individuals. A report by the World Health Organisation (2015) found that 29% of the worldwide population has lacked safety in managing drinkable water services. 61% of the global population requires safely managed sanitation services. Meanwhile, raw wastewater effluent entering fresh water bodies, become poisonous to aquatic biota and return the unsafe water that is intended for drinking (Okaiyeto *et al.*, 2016). The consequences of this threat are increase in water borne diseases and an increase in the demand for non-hazardous water for public and industrial needs. The impact of this current situation caused an annual death of 361,000 children under five years old from diarrheal diseases due to poor sanitation, poor hygiene or polluted drinking water which corresponds to diseases like cholera, dysentery, hepatitis and typhoid (WHO/UNICEF, 2017).

Significant awareness was concentrated on water treatment; hence, its significant potential to assess water standard on a constant basis (Yang *et al.*, 2012). The most significant is the development of a new technological approach in water and wastewater treatment to contain a stringent environmental rule on the quality of the wastes that are discharged into water bodies (Wong *et al.*, 2006). Several states have established many strict regulations which consider the occurrence of water pollution, and confirm

appropriate treatment of local, industrial wastewater discharge and agricultural wastewater effluents before the effluent enter different water bodies (Bhatnagara & Sillanpaa, 2010; Li *et al.*, 2013).

2.2 Flocculation Process in Water Treatment

The majority of the water treatment stations should have raw water from the tank to enter the first chamber where flocculant substances are added. Then the water is transferred to a sedimentation pond where the flocculating procedure proceeds and suspended particles precipitate (Jarvis *et al.*, 2012). After this stage the clarified water passes through a filtration process preceeding to disinfect for distribution systems (Rong *et al.*, 2013). Coagulation and flocculation are the most important processes for the removal of particulates and natural organic matter in potable water treatment plants (Keeley *et al.*, 2016; Rong *et al.*, 2013).

Flocculants are naturally occurring organic macromolecules; they are substances that can flocculate suspended solids, cells, and solid colloid particles (Bhunia *et al.*, 2012). Flocculants are widely used in separation techniques, such as in drinking water purification, wastewater treatment, activated sludge dehydration, downstream processing, and food fermentation (Chen *et al.*, 2017). Flocculants also play an important role in flocculating and sludge dewatering processes for the removal of solids and reduction of sludge bulk in wastewater treatment (Lee *et al.*, 2015). Moreover, flocculants are applied as alternatives for filtration and centrifugation in the separation of microbial species from fermented food, pharmaceutical and beverage industres (Shih & Van, 2001),

Flocculation is an operative skill that is normally applied in wastewater treatment for removing a number of metal ions as well as suspended particles (Deng *et al.*, 2003). Bestowing to the bio-flocculation mechanism, the bioflocculants can be adsorbed onto the surface of suspended particles, should in adjacent narrowly enclosed to the suspended

particles and its requisite has the essential sufficient attractive force to control the electrostatic repulsive force (Wang *et al.*, 2011).

Furthermore, Lee *et al.* (2012) stated that the effective and prompt flocculation process was influenced by the size of particles, which implicated that the bigger size had a more rapid settling rate. Besides, the designated flocculants have a key effect on the flocculating process capabilities, accumulated particles potency and quantity as well as formation of strong bounds as the result of the flocculating mechanism (Zhang *et al.*, 2014). For instance, the flocculating competence and magnitude of the polyelectrolytebonds are better than that of ferric chloride.

Regardless of the flocculating process peak effectiveness in water treatment, as well as the main weakness of flocculation, it produces minor-flocs through flocculation at low temperature or produces fragile- flocs that can dissolve with the employment of a physical force (Lee *et al.*, 2014). Therefore, it is important to control these problems and expand the flocculating processes to improve its successful applications.

2.3 Classification of Flocculants

Flocculants are applied in various industrial processes, such as purification of drinking water, wastewater treatments, and dredging / downstream processes and in food fermentation industries (Chen *et al.*, 2017). Flocculants are commonly classified as organic, naturally occurring flocculants and inorganic flocculants, as shown in Figure 2.1.



Figure 2.1: Classes of Flocculants.

2.3.1 Organic Flocculants

Organic flocculants are usually applied in several industrial processes as flocculating agents and they comprise of polyethylene amine, polyacrylamide (PAA) and poly-diallyl dimethyl ammonium chloride (PDADMAC) (Lee *et al.*, 2014). As reported by Moussas and Zoulboulis (2009), the acrylamide products are the main type of organic and artificial polymers that are widely applied as flocculation agents due to their cost-efficiency and competence. In nature acrylamide is a white crystalline and a moderately constant monomer that is greatly solvable in water and several organic solvents and it has no scent (Wong *et al.*, 2006). It has a multifunctional group that consists of an amide group and a vinyl carbon-carbon bond with an insufficient double bond that is susceptible to a wide range of chemical reactions (Girma *et al.*, 2005).

However, polyacrylamide (PAA) is a non-ionic polymer, which does not donate extra charges to the flocculating reaction, and thus has no influence on the capability of the inorganic cationic polymers (Okaiyeto *et al.*, 2016). Moreover, the usage of non-ionic organic polymers, such as polyacrylamide might prevail over the extent of the priorly revealed problem encountered with inorganic polymers (Moussas & Zoulboulis, 2009). Besides, these organic polymers are frequently the resultant from non-renewable raw materials or oil-based production (Suopajarvi *et al.*, 2013).

In addition, they generally have a large molecular weight, and contains various polyelectrolyte charges in their molecular sequence which increase their flocculation efficiency (Lee *et al.*, 2014). Huang *et al.* (2014) reported that the volume of sludge produced in wastewater treatment could be condensed by applying artificial polymers like polyacrylamide (PPA) which is not sensitive to pH.

A combination of polyferric sulphate (PFS) with polyacrylamide (PAA) must be present in an improved flocculation activity, while PAA is the excellent flocculating mediator. Synergetic of PAA with PFS would increase the molecular weight of polymers; hence, increasing its flocculation activities as compared to an inorganic polymer (Okaiyeto *et al.*, 2016). Yet, these monomers of the polyacrylamide are not simply decomposed, and thus poised an environmental hazard. In addition, these types of polymer have carcinogenic properties and neurotoxicity. Consequently, these related disadvantages have decreased their usage in many countries (Li *et al.*, 2008).

2.3.2 Inorganics Flocculants

Inorganic flocculants consist of, alum, aluminium chloride, aluminium sulphate polyaluminium chloride, ferrous sulphate and ferric chloride. Meanwhile, the majority of the suspended solids present in wastewater commonly display a negative charge, but when these metals or salts are added to wastewater they can be ionised to form positive charges which could bind with the negatively charged suspended particles (Lee *et al.*, 2014). This type of interaction decreases the surface charge and formation of micro-floc which can create aggregates to form bigger flocs that are easily settled down in solution (Suopajarvi *et al.*, 2013). Amongst these inorganic polymers, polyaluminium chlorides (PAC) are generally applied in drinking water and wastewater treatment. Actually, they are sensitive to pH, ineffective at low temperature, restricted to only limited separation systems and need a larger volume for its flocculating efficiency, and hence produce a bulk capacity of sludge which is activated in wastewater treatment plant systems (Lee *et al.*, 2014; Sharma *et al.*, 2006). Therefore, it is important to investigate the effective technologies that must be rational and operative in excessive sludge recycling, Likewise, many researches have designated that PAC includes aluminium which may spoil potable water and cause severe health complications to users (Banks *et al.*, 2006).

Latest biochemical, neuropathological and epidemiological research were identified as the potential effect of aluminium on the neurological system, as far as aluminium is anxious, the highest brain-health concern is Alzheimer's disease (Lee *et al.*, 2014). Moreover, inorganic flocculants such as ferric polysilicate was discovered to possess smaller molecular weight and flocculation effectiveness as compared to organic flocculants (Moussas & Zouboulis, 2008; Shi & Tang, 2006). Furthermore, despite the flocculation ability of improved ferric polysilicate over ferric sulphate, the addition of polysilicic acid, which have negative charge, can distress the destabilisation capability of the modified flocculants, in the meantime the positive charge on iron classes may be compromised (Moussas & Zouboulis, 2009).

Recently, the awareness of exploiting flocculating compounds in wastewater treatment has attracted more consideration as well as many labour force (Gao *et al.*, 2008). For

example, the greatest flocculation performance was obtained when 1.0 mg/L of PAC and 0.1 mg/L PDMDAAC were used as treatment dose. While the addition of PDMDAAC has improved the natural organic matter (NOM) on removal activities, particularly at low PAC dosages (Shen *et al.*, 2017). For example, the treatment of kaolin suspension or dye solution by using poly-ferric chloride and poly-dimethyldiallyl ammonium chloride (PFC, PDMDAAC), together revealed a better flocculation activity than that of using a single flocculant agent (Gao *et al.*, 2008). Besides, the use of this flocculant compound to the dye solution for textile manufacturing, produced a higher flocculation activity is achieved (Wang *et al.*, 2007). Gao *et al.* (2007) reported that flocculation efficiency was influenced by the proportion of organic flocculants used, as PFC-PDMDAAC which carries more cationic charges as compared to PFC. However, the reduced application of these polymers compound is because they are found merely to be effective in some specific sample treatments which have a higher cationic charge (Moussas & Zouboulis, 2009).

2.3.3 Naturally Occurring Flocculants

Microorganisms produce many polysaccharides that are classified according to their biological functions, such as intracellular storage polysaccharides, extracellular polysaccharides alginate, cellulose, sphingan, (e.g. xanthan) and. capsular polysaccharides (closely interconnected to the outer-surface). Moreover, many bacteria produce a wide-range of exo-polysaccharides which are synthesised by divergent biosynthesis pathways (Jochen et al., 2015). The demand for natural polymers, for example, chitosan, collagen, gelatine and starch, for the treatment of acute and chronic wounds is increasing due to their biocompatibility and biodegradability properties (Zhong et al., 2018).

2.3.3.1 Chitosan

Are hydroxyl (-OH) and amino groups (-NH₂). These two groups possess long-pair electrons that contribute an electron pair to vacant d-trajectories of metal ions; thus, chelating into a chemical compound complex (N-glucosamine) (Okaiyeto *et al.*, 2016). Furthermore, it is a bio-polymer acquired from shellfish sources, a relatively deacetylated polymer induced from the chitin alkaline deacetylation (Lee *et al.*, 2014). Defang *et al.* (2008) found that chitosan is a cationic polymer (polysaccharide) which functions as an artificial polymeric flocculant that could be used in flocculating organic matters in the water and wastewater treatment process ecause they are biodegradable products, environmentally safe and non-corrosive. Szygula *et al.* (2009), reported that chitosan is soluble in diluted organic solvents, while it is insoluble in concentrated organic solvents, as well as water. Chitosan also exists as a solvable cationic polymer, having intensity charge density at acidic pH (Rinaudo, 2006). When chitosan is dissolved in acidic medium, it can synthesise a neutralised amine group through an electrostatic attraction reaction that could eliminate different undesirable heavy metals such as Ag+, Al⁺³, Ca⁺², Cd⁺², Cr⁺², Pb⁺², Hg⁺² and Zn⁺² that prevail in the wastewater (Defang *et al.*, 2008).

Chitosan has a strength electrostatic attraction and strong adsorption due to the presence of amino groups (-NH₂) in the molecular sequence which can protonate with H⁺ in water to make cationic ammonia (NH⁺³). As a result, it could coagulate small suspended particles into a large flocculate, which can be effectively removed by sedimentation out of a solution (Yang *et al.*, 2016). Chitosan was found to be very effective in the removal of the chemical oxygen demand (COD) and suspended solid substance (SS) in water treatment of contaminated water with organic materials (Bolto, 1995; Ishii *et al.*, 2008). Moreover, it has various benefits over the conventional polymeric flocculants that are commonly applied in water treatment. These advantages

comprise: a small dose requirement, quicker residue ingrate, high COD reducing ability, heavy metals and suspended solids. Likewise, it is employed to decrease the production of bulky sludge mass that is commonly produced by inorganic polymer and it does not produce any secondary pollution (Renault *et al.*, 2009). On account of high density of chitosan, it enhances to form larger flocs, which will increase the floc settling volume and decrease the duration of the sedimentation phase (Renault *et al.*, 2009). Even though chitosan is potent in water treatment, it is expensive, and hence its usage effectiveness increases in the overall treatment costs.

2.3.3.2 Sodium Alginate

Sodium alginate is a linear anionic polymer, which is water-soluble and is obtained from the sodium salt of alginic acid and possess a molecular weight of approximately 500,000 (Wu *et al.*, 2012). Sodium alginate was evaluated for its flocculation efficiency in combination with aluminium sulphate as coagulants in the fabric wastewater effluents with artificial dyes which revealed a rugged flocculation efficiency, reduction of colour removal of about 90% and COD of around 80% (Wu *et al.*, 2012). Moreover, sodium alginate was used as the coagulant facilitate which can lead to stronger and larger flocs which could enhance treatment efficiency of the coagulation process (Wang *et al.*, 2014).

2.3.3.3 Tannin

Tannin is an anionic organic substance that are are described as a harmless flocculant which can be practically applied as an alternative for frequent use of polymers in water treatment, owing to its safety to humans, biodegradability and environmental friendly (Ozacar & Sangil, 2000). Tannin is derived from the secondary metabolites of plant tissues, green leaves and fruits of vegetables (Beltran-Herediaa & Sanchez-Martin, 2009). Previously, many investigators have established the flocculation competency of tannin in the removal of colloidal and suspended particles present in potable water after treatment, in addition to the removal of suspended particles from artificial wastewater and the removal of inks, dyes, and colours from industrial wastewater (Roussy *et al.*, 2005). Lee *et al.* (2014) indicated that the combination of tannin with aluminium sulphate as a coagulant to stabilise the negative charge of colloidal particles.

2.3.3.4 Cellulose

Cellulose is an insoluble material that is a major component of the plant cell wall and plant fibres and very abundant in natural polysaccharides acquired from agriculture wastes (Lee *et al.*, 2014). In current years, cellulose has been the theme of studies due to several applications in industries, such as paper, textiles and forestry products (Caciced *et al.*, 2016). Khiari *et al.* (2010) reported that the carboxymethyl-cellulose anionic sodium (CMCNa) is a classical sample of flocculants derived from cellulose, which are designated as environmentally friendly and is used as a composite with aluminium sulphate as a flocculant for the removal of turbidity in potable water. Despite the fact, cellulose is widely known as a plant product, while some bacteria have acquired consideration as a sustainable and alternative source of cellulose by using *Komagataeibacter xylinus* as the representative of bacterial cellulose producer (Caciced *et al.*, 2016).

Plant cellulose and bacterial cellulose have the same chemical formula, but bacterial cellulose has fine physical properties with respect to thermos-stability, mechanical stability, tensile strength, purity, crystal fibrous structure and bio-compatible substance (i.e. biologically non-toxic and no immune system response) (Cacicedo *et al.*, 2016; Zhang *et al.*, 2015). Moreover, Suopajarvi *et al.* (2013) described that an ionised dicarboxylic acid nano-cellulose (DCC) bioflocculant was obtained from cellulose, which

has a strength flocculation property in the prevalence of ferric sulphate in municipal wastewater treatment.

2.3.3.5 Bioflocculant

Recently, natural polymers requirement for different industrial uses havewere directed to an important development in the exopolysaccharide production (EPS). Typically, it is a composite with large molecular weight and has long-chain mix of polymers, consisting of branched repeating units of sugar or carbohydrate derivatives, such as glucose, galactose, fructose, and mannose which are synthesised and secreted by different growing microorganisms (Ismail & Namoothiri, 2010; Sheng *et al.*, 2010).

Microbial flocculation was first described in a yeast strain *levur casseeuse* by Louis Pasteur in 1876. Subsequent, a similar phenomenon was detected by Bordet (1899), in bacterial cultures; vast studies of flocculation mechanism by using Zoogloea- forming bacteria isolated by Butterfield (1935) from activated sludge presented a direct association between the cell aggregation and accumulation of extracellular bioflocculants. The involvement of extracellular produced compounds in microbial flocculation was further confirmed when Tenny and Stunn (1965) suggested that flocculation by microorganisms was mediated by polymers they excreted extracellularly.

According to Aljuboori *et al.* (2015), the use of cell-free supernatant, was surveyed for their ability to disrupt the spread of bacteria. So far, over hundreds of bioflocculantproducing microorganisms are reported in literature with the produced bioflocculants extensively characterised. Among the exopolymeric substance (EPS) discovered in the literature, those that have flocculation property are mainly concerned with the subjects of bioflocculation and this recommends their application in different industrial processes and water treatment. Due to the limits of these organic and inorganic polymers, bioflocculants have acquired vast scientific considerations because of its extraordinary flocculating activities and biodegradability, non-secondary toxic waste, and their deteriorating intermediates are harmless for humans and their immediate environment (Mabinya *et al.*, 2012; Shahadat *et al.*, 2017).

The bioflocculants are biological macromolecules eco-friendly, and were successfully confirmed for wastewater treatment (Pathak et al., 2017). Moreover, polysaccharides produced by cyanobacteria are considered as environmentally friendly, and illustrate an efficient ability to adsorp metals or inorganic substances. Therefore, the type of polysaccharides formulating eco-friendly is suitable for and functional bionanocomposites with different rigid materials (Okajima et al., 2018). Therefore, it has become essential to isolate and detect novel bioflocculant producing microorganisms and determine procedures for their optimum production conditions to recover the bioflocculant yield or by using a consortium microbe to enhance the bioflocculant production (Okaiyeto et al., 2013). Exopolysaccharides (EPS) are generally complex in nature with heterogeneous compounds. Their components, position and locality are dependent on numerous metabolic processes, for instance, modifications in logarithmic growth phase secretion, cell rupture due to cell death and releasing on the cell surface macromolecules, i.e. lipopolysaccharides (LPS) and outer membrane protein and their interaction with the instant environmental conditions (Cristina et al., 2011). Abdel-Aziz et al. (2011) detected the definite enzymes that exist in clusters which regulate the exploitation of nutrient in culture media by microorganism. This was to synthesised bioflocculant with large molecular weight, which secreted or form a capsule on the surface of the microbial cell. EPSs are commonly named as exopolysaccharides due to their position; this is to distinguish them from other classes of polysaccharides that could be made by enclosed cells (Nwodo et al., 2012).

For example, Morris and Harding, (2009) reported that naturally the capsular polysaccharides tremendously provoke the immune system (immunogenic), and might have altered their infrequent biological diversity as a mode of escaping the antibody response to develop/design a vaccine production. Furthermore, they act a major role in the adherence and penetration of the host cells. Exopolysaccharides are the best significant component of biological coagulates and are more effective on the deprivation of organic matters from wastewater, which also contain activated sludge and biofilms after treatment (Martin- Cereced et al., 2001). Exopolysaccharides stimulate the formation of bioflocs by improving the relation between bacterial clusters, diverse bacterial strains, as both inorganic and organic particles. Moreover, their essential starring role is to clamp the microbial cells together (Sheng et al., 2010). They are commonly occurring as a sheet on the surface of the microbial cell, and thus provide protection for the microbial cell against adversative environmental conditions, for example, temperature, oxygen tension, high osmotic pressure and noxious substance. Likewise, Nichols et al. (2005) reported that EPSs might contribute to the adsorption of heavy metals as well as avert desiccation under definite environmental conditions.

Recently, bioflocculants are produced by many kinds of microorganism, such as algae, actinomycetes, bacteria and fungi which were found and isolated from different habitats and marine environment (i.e. soil, water, activated sludge) which sustain a fertile biodiversity of microorganisms, remaining largely unexplored (Aljuboori *et al.*, 2013; Cosa & Okah, 2014; Czemierska *et al.*, 2017; Kumari *et al.*, 2014; Nwodo *et al.*, 2014; Okaiyeto *et al.*, 2015).

2.4 Composition Analysis and Flocculating Potential of some Bioflocculants

Many researches have publicised that the majority of bioflocculants produced are either functional proteins or functional carbohydrates (Huang *et al.*, 2005), for example, the EB-EPS bioflocculant secreted by Enterobacter sp. whereby Fourier-transform infrared (FTIR) spectrometry analysis of the purified EB-EPS showed the occurrence of carboxyl, hydroxyl and amide groups. Chemical analysis showed that the purified EB-EPS contain 88.7% w/w of carbohydrate and 11.3% w/w of protein (Muthulakshmi et al., 2017). Muthulakshmi's et al. (2017), study showed and identified that the bioflocculant, named B1-EPS produced by Bacillus sp., comprise of carbohydrate (92.8% w/w) and protein (21.8% w/w). The bioflocculant PSK1 produced by *Bacillus aryabhattai* is mainly composed of glycoprotein which consist of glucose and rhamnose with some amino acids, such as arginine and phenylalanine. FTIR spectrum analysis of PSK1 indicated the presence of amino and hydroxyl groups (Ayat et al., 2017). Czemierska et al. (2016), found that the bioflocculant secreted by Rhodococcus opacus contain polysaccharides (64.6% w/w) and protein (9.44% w/w) and the bioflocculant constituent sugars were glucose, galactose and mannose. Furthermore, the FTIR spectrometry examination of purified bioflocculant from Bacillus pumilus showed the manifestation of amino, hydroxyl and carboxyl groups and the bioflocculant composed of sugar (75.4%), protein (5.3%) and uronic acid (15.4%) (Makapela et al., 2016). Moreover, novel bioflocculant have functional groups, i.e. hydroxyl, amino and carbonyl and its carbohydrate and protein content were estimated to be 57.12% and 31.7% protein, respectively, while the polysaccharide content comprised of four monosaccharides, i.e. D-xylose, D-fructose, maltose and mannose (Sun et al., 2015). The bioflocculant secreted by Bacillus sp. was found to be water soluble and insoluble in organic solution, FTIR spectrum analysis showed that the manifestation of carboxyl, hydroxyl and carbohydrate derivatives in the polymers (Karthiga & Natarajan, 2015). Nwodo et al. (2014), found that the bioflocculant synthesised by a consortium species of Actinobacteria i.e. Cellulomonas and Streptomyces, FTIR spectrum analysis specified the manifestation of carboxyl, hydroxyl and amino groups, with molecular weight 494.81-18,330.26 Da. The chemical analysis

of purified polymers also showed the presence of polysaccharides and protein and uronic acids. Yin *et al.* (2014) established that the novel bioflocculant synthesised by *klebsiella* sp. ZZ-3, revealed that its chemical analysis had carbohydrates (84.6%) and protein (6.1%) with relatively high molecular weight of 603-1820 kDa. The bioflocculant MBF-TG-1 produced by *klebsiella* sp. TG-1 primarily composed of polysaccharides (84.6%) and protein (11.2%) (Liu *et al.*, 2013). Gomaa E.Z. (2012) indicated that the bioflocculant from *Pseudomonas aeruginosa* ATCC-10145, composed of sugars (89% w/w) and protein (27% w/w) and the infrared spectrometry analysis revealed that the purified exopolymer comprised of hydroxyl, carboxyl, amino and sugar derivative groups.

He *et al.* (2010), indicated that the bioflocculant HBF-3 secreted by *Halomonas* sp. V3a was isolated from deep sea was composed of carbohydrates, comprising of neutral sugar (20.6% w/w), uronic acid (7.6% w/w), amino sugar (1.6% w/w) and sulphate group (5.3 % w/w), and with molecular weight of 595KDa. Likewise, bioflocculant MBFA9 secreted by *B. mucilaginosus* composed of carbohydrates containing primarily of neutral sugar (47.4% w/w), uronic acid (19.1% w/w) and amino sugar (2.7% w/w) and the FTIR spectrometry analysis showed the occurrence of hydroxyl and carboxyl groups as the essential function moieties (Deng *et al.*, 2003). Feng and Xu (2008) found that the acidic bioflocculant synthesised by *Bacillus* sp. BF3-3 contained polysaccharides (66.1% w/w) and protein (29.3% w/w). The produced bioflocculant by *Aspergillus parasiticus* mainly composed of carbohydrate (76.6% w/w) and protein (21.6% w/w) with an average molecular weight of 3.2×10^5 Da (Deng *et al.*, 2005).

This type of bioflocculant display a high flocculating efficiency against kaolin clay suspension, and it has the capability to flocculate dye from synthetic suspension. Deng *et al.* (2005) described that the manifestation of amino, carboxyl, hydroxyl, and amide functional groups in the molecular sequence of bioflocculants were the major functional

moieties, which played an essential part in the flocculating processes of suspended particles because these functional groups were responsible for adsorption sites, whereby the suspended particles could be bound. The exopolymers secreted by Bacillus genus and isolated from a Qatari ecosystem, were applied against kaolin suspension, indicating a flocculating activity of about 85% at 20 mg/L concentration (Desouky et al., 2008). The flocculating activity of the bioflocculant secreted by *B. mucilaginosus* by using kaolin clay suspension was found to be 99.6% at a concentrated dose of 0.1mg/L (Deng et al., 2003). Similarly, bioflocculant secreted by Vagococcus sp. W31 mainly composed of polysaccharide, with molecular weight of 2×10^6 Da, with high flocculation activity in a wide-range of pH (7-11) at a prerequisite dosage of 25 mg/L (Gao et al., 2006). Moreover, the bioflocculant derived from Corynebacterium glutamicum was found to be a polysaccharide as well as it revealed thermo-stability in an acidic range of the pH 30 to pH 6.5 (He et al., 2004). The flocculation effectiveness of the bioflocculant was comparatively higher as compared to artificial flocculants. These characteristics designate its application in the decolourisation of molasses effluent waste. The FTIR spectrum analysis presented the manifestation of carboxyl and hydroxyl groups in the molecular sequence (He et al., 2010).

However, some of these bioflocculant composed higher quantity of protein than polysaccharides. For example, the intracellular bioflocculant called MBF-W6 synthesised by *C. daequense* W6, is mostly composed by protein (32.4%) and polysaccharides (13.1%) (Liu *et al.*, 2010). The bioflocculant synthesised by any microorganisms are a pre-determining factor that affects its flocculation competence (Gao *et al.*, 2006).

2.5 The Functional Groups in Bioflocculant Molecule

The functional group in bioflocculant molecule plays very a significant role in the flocculation process. Therefore, studies on bioflocculant production and Characterisation

investigated the manifestation of functional groups in bioflocculant molecules to recognise and determine their flocculation performances and mechanism. For example, many studies exhibited the presence of carboxyl, hydroxyl, amino and methoxyl groups in bioflocculant molecules were identified to contribute to the flocculation of colloids, such as those detected in polyelectrolytes (Li *et al.*, 2009; Salehizadah & Shojaosadati, 2001). According to Zheng *et al.* (2008) finding the presence of hydroxyl (OH), carboxyl (COOH) groups in the bioflocculant contents and OH⁻, H⁺, groups on the surface of the constituent part can perform hydrogen bonds when the bioflocculant chains are attached to the surface of particles.

2.6 Bioflocculation Mechanisms

Investigation on the flocculation mechanism of bioflocculants discovered the flocculation performance of bioflocculant in flocculation process. Many researches presented the components comprised of bioflocculants which regulated the flocculation mechanism and capability (He *et al.*, 2010; Zhang *et al.*, 2010). However, the main bioflocculation mechanism could be divided into: charge neutralisation and polymer bridging.

2.6.1 Charge Neutralisation

Charge neutralisation is one of the major bioflocculant mechanisms where the negative charge of impurities, either particles or colloids, are neutralised by positively charged bioflocculants (Lian *et al.*, 2008). The occurrence of amino groups in bioflocculant molecules could be simply protonated and result in a positive charge (Deng *et al.*, 2005). Consequently, electrostatic interaction between the positively charged bioflocculant and negatively charged colloids might produce electron attraction and charged neutralisation of the colloid surface, controlling the formation of flocs and decrease the electrical attraction between them (Figure 2.2). Based on the above mechanism, this bioflocculant

was classified as cation-independent bioflocculant, including bioflocculants produced by *Klebsiella* sp. ZZ-3(Yin *et al.*, 2014), *Solibacillus silvestri* W01 (2013). *Pseudomonas aeruginosa* ATCC-10145 (Gomaa. 2012), *B. subtilis* (Bhunia *et al.*, 2012), *B. mojavensis strain* 32A (Elkady *et al.*, 2011), *Methylobacterium* sp. (Ntsaluba *et al.*, 2011), *B. licheniformis* (Shih *et al.*, 2010), *Chryseobacterium daeguense* W6 (Liu *et al.*, 2010), *B. mucilaginosus* (Lian *et al.*, 2008), *Klebsiella* sp. (Sheng *et al.*, 2006).



Figure 2.2: Charge Neutralisation Flocculation Mechanism (Reproduced with permission from the Library, Bhavnagr University, India)

2.6.2 Polymer Bridging

Biopolymer bridging suggested the flocs formed primarily by cation mediated bridges between the bioflocculant molecules and kaolin particles (Sobeck & Higgins, 2002). The cation could prompt the bioflocculation by neutralisation and destabilisation of residual negatively charged carboxyl groups of the bioflocculant, establishing bridges which bind kaolin particles to each other (Prasertsan *et al.*, 2006). Figure 2.3 shows the polymer bridging flocculation mechanism. Based on this mechanism, this type of bioflocculants is classified as cation-dependent bioflocculant. Many studies presented that bioflocculants are produced by *Actinomycete*, *Streptomyces* sp. hsn06 (Li *et al.*, 2017), *Bacillus* sp. AEMREG4 (Ntsangani *et al.*, 2017). *Bacillus* sp. (Okaiyeto *et al.*, 2016), *B. amyloliquefaciens* ABL19 (Ogunsade *et al.*, 2015), *Arthobacter* sp. B4 (Li *et al.*, 2014). *Klebsiella* sp. TG-1 (Liu *et al.*, 2013), *Klebsiella mobilis* (Wang *et al.*, 2007), Enterobacter cloacae WD7 (Prasertsan et al., 2006), Enterobacter aerogenes (Lu et al., 2005), Sorangium cellulossum NUST06 (Zhang et al., 2002b), Enterobacter sp. BY-29 (Yokoi, 1997), Bacillus subtilis (Yokoi et al., 1996) were cation-dependent bioflocculants.



Figure 2.3: Bioflocculant Bridging Flocculation Mechanism (Reproduced with permission from the Library, Bhavnagr University, India)

2.7 Factors Affecting Bioflocculant Production

Bestowing to the presented literature study, the production of microbial bioflocculants was extremely affected by the composition of the production medium and some other physicochemical parameters (Fang *et al.*, 2013). Besides, these discoveries indicated that the bioflocculant production was affected by many factors that comprised of the culture medium ingredients in growth addition, environmental conditions (He *et al.*, 2004). The effects of the nutritional constituents have been widely examined for the production of bioflocculants (Abdel-Aziz *et al.*, 2011).

The influence of the essential factors, such as the initial pH of the production medium, carbon, nitrogen source, inoculum size, culture time, ionic strength, metal ion, shaking speed, and incubation temperature, greatly affect the bioflocculant production (He *et al.*, 2004). Remarkably, in the bioflocculants production, the microorganisms can consume uncommon carbon sources as nutrients besides the production of other bacterial

secondary metabolites. For example, *Rhodococcus* sp. is efficiently consumed by the alkanes in biosurfactant, to produce bioflocculants (Pathak *et al.*, 2017). Also, some studies illustrated that the potential of microbes to utilise petroleum hydrocarbon during its metabolic pathway to produce bioflocculants (Pathak *et al.*, 2015).

2.7.1 Effect of Carbon and Nitrogen Sources

Many researches have recognised the implication of carbon and nitrogen sources, which played a crucial role in improving the bioflocculant production by many microorganisms (Salehizadeh & Yan, 2014). For example, Makapela et al. (2016) found that the maltose was the sole carbon source, as well as nitrogen sources, such as urea, yeast extract and ammonium sulphate and initial pH 7, were favoured by Bacillus pumilus for bioflocculant production. Meanwhile, the best preferred carbon, nitrogen sources for consortium culture of Bacillus sphaericus F6 and Rhizobium radiobacter F2 were found to be glucose and urea for producing a novel bioflocculant CBF (Li et al., 2016). Nwodo et al. (2014) found that the consortium of Cellulomonas and Streptomyces species for bioflocculant production, sucrose, peptone and magnesium chloride were favoured as carbon and nitrogen sources, giving flocculating activity of 91% and 82%, respectively. Lee et al. (2014), stated that B. licheniformis X14 favoured sucrose, starch and ethanol as the preferred carbon sources for the production of bioflocculant ZS-7, while ammonium chloride was the favoured choice as the nitrogen source. Sheng et al. (2006) found that maltose and urea are the most preferred carbon and nitrogen sources for bioflocculant production by Klebsiella sp. Cosa et al. (2013b) detected that tryptone and sodium carbonate were the best appropriated nitrogen source for Oceanobacillus sp. to produce bioflocculants. Tryptone was chosen as the organic nitrogen source for the bioflocculant production via Chryseobacterium daeguense W6 (Liu et al., 2010). Whereas, the organic carbon sources, such as glucose, lactose, maltose, sucrose and xylitol are all appropriate

substrates for *Paenibacillus polymyxa* BY-28 to produce the bioflocculant (Gong *et al.*, 2011).

The novel bioflocculant REA-11 synthesised by *Corynebacteria glutamicum*, sucrose was utilised as a carbon source and composite nitrogen source containing corn steep liquor and urea as documented by (He *et al.*, 2004). Cosa *et al.* (2013a) reported that glucose was favoured as carbon source among other sources examined for the production of bioflocculant by *Virgibacillus* sp. Whereas, as presented glucose, glycerol, corn starch, and sucrose. were suitable substrates for *Aspergillus parasiticus* to produce bioflocculant, performing a good flocculating activity of over 80% at three days of fermentation (Deng *et al.*, 2005). The optimum carbon source for bioflocculant production by *Solibacillus silvestris* W01 was prevailed after maltose was used as the carbon source in the cultivation medium (Wan *et al.*, 2013).

For instance, *Sorangium cellulosum* utilised soluble starch as a carbon source to sustain the optimum bioflocculant production, while the addition of 3 g/L glucose as an enhancement, completely inhibited cell binary fusion and synthesis of the bioflocculant (Zhang *et al.*, 2002). But, the lactose, glucose, and fructose were not appropriate for *B. licheniformis* to produce bioflocculant, while the simultaneous occurrence of numerous carbon sources, such as citric acid, glutamic acid, and glycerol in the production fermentation media hadenhanced cell growth and bioflocculant yield (Shih *et al.*, 2010). While, bioflocculant yield was increased by the optimised strain of *Penicillium* sp. HHE-P in the fermented medium, consisted of glucose and yeast extract.

Glucose was the greatest preferable carbon source for bioflocculant production by most microorganisms, but the higher cost of glucose increased the production cost (Liu & Chen, 2010). For instance, while molasses was replaced by glucose, flocculation efficiency against kaolin suspension achieved over 90%, a strong significant indication

confirmation has revealed that different bacterial strains could exploit as either organic, inorganic nitrogen source or their mixture sources for producing bioflocculants (Gong *et al.*, 2008).

The production of exopolysaccharides by *Rhizobium radiobacter* SZ4S7S14 was optimised for bioflocculant yield enhancement, where the D-Mannose and yeast extract were utilised as carbon, nitrogen sources (Rasulov *et al.*, 2017). However, when peptone merged with inorganic sodium nitrate, it was found to be the most appropriate nitrogen source for *Aspergillus parasiticus* for bioflocculant synthesis (Deng *et al.*, 2005).

However, when combined with $(NH_4)_2SO_4$, no bioflocculant was produced. Deng *et al.* (2005) found that the organic nitrogen sources, were preferred and had enhanced bioflocculant yield in many bacterial strains. For instance, when both beef extract and urea was utilised as a nitrogen source by the strain of S-14, it was found to be the most appropriate for the production of bioflocculants. Meanwhile, strain TJ-1 was successfully capable of consuming peptone, beef extract, and yeast extract as a nitrogen source. However, when the peptone was used as the sole organic nitrogen, the source was found to increase the bioflocculant yield and was the cheapest substrate (Xia *et al.*, 2008)., Cosa *et al.* (2013a) also examined the efficiency of a composite nitrogen source, containing of yeast extract, urea and $(NH_4)_2$ SO4 to enhance the optimum production of bioflocculant by *Virgibacillus* sp. Likewise, Gong *et al.* (2008) reported that the combined nitrogen source, such as beef extract and urea revealed a considerable enhancement in production of bioflocculant by *Serratia ficaria* among others that were examined.

Correspondingly, the bioflocculant was produced by mixing strains of *Acintobacter*, *Agrobacterium* and *Enterobacter* species, in which the medium were merged with a yeast extract complex and ammonium sulphate as the nitrogen source (Kurane & Matsuyama, 1994). The sodium nitrate and peptone were found to be the most favoured nitrogen

source among other sources examined by *A. parasiticus* for the production of bioflocculant (Deng *et al.*, 2005). Likewise, a study by Li *et al.* (2013) confirmed that the peptone was found to be the most preferable nitrogen source for the *Paenibacillus elgii* B69 to produce bioflocculant among other assessed sources.

Microorganism	Source	Carbon source	Nitrogen source	Chemical composition	Flocculating activity (%)	Yield	Reference
Achromobacter xylosoxidans	Activated sludge	Sucrose	Yeast extract +urea	Glycoprotein	83.3	5gL-1	Subudhi <i>et al.</i> (2016).
Aeromonas sp.	Activated sludge	Corn flour	Soyabean flour	Polysaccharide	49.34	NA	Li et al. (2008).
Aeromonas sp.	Activated sludge	Glucose	Peptone	Polysaccharide	92.4	2.25	Li <i>et al</i> . (2007).
<i>Arthrobacter</i> sp. Raats	Freshwater	Lactose	Urea	Glycoprotein	87.5	NA	Mabinya <i>et al.</i> (2012).
Aspergillus parasiticus	NA	Starch	Pepton+sodium nitrate	Glycoprotein	98.1	NA	Deng <i>et al.</i> (2005)
Aureobasidium pullulans	NA	Sucrose	NaNO ₃	Polysaccharide	12.5	NA	Ravella <i>et al.</i> (2010).
Bacillus aryabhattai	Soil	Glucose	Yeast extract	Glycoprotein	92.8	~6g/l	Ayat <i>et al.</i> (2017).
Bacillus sp. AEMREG7	Marine	Glucose	Urea + yeast extract + (NH ₄) ₂ SO ₄	Glycoprotein		1.6	Okaiyeto <i>et al.</i> (2015b).
Bacillus sp. AEMREGH	Thyume river	Starch	Yeast extract	Glycoprotein	76	NA	Nozipho <i>et al.</i> (2017).
Bacillus alvei NRC-14	Soil	Chitosan	Yeast extract	Polysaccharide	98	10	Abdel-Aziz <i>et al.</i> (2011).
Bacillus clautti	Brewery wastewater	Glucose	NA	Glycoprotein	88.67	NA	Adebayo, T.B & Adebami, G.E. (2014).
Bacillus sp. Gilbert	Marine	Sucrose	Ammonium chloride	Polysaccharide	91	NA	Piyo <i>et al.</i> (2011).

Table 2.1: Optimum Culture Conditions, Chemical Compositions, Flocculating Activity and Yields of some Bioflocculant.

Microorganism	Source	Carbon source	Nitrogen source	Chemical composition	Flocculating activity (%)	Yield	Reference
Bacillus licheniformis	Contaminated LB medium	Sucrose	Yeast extract + urea	Glycoprotein	700 U/ml	2.94	Xiong <i>et al.</i> (2011).
Bacillus licheniformis X14	Soil	Glucose	NH4 Cl	Glycoprotein	99.2	NA	Li et al. (2009).
Bacillus sp.	Maya Marine	Glucose	Ammonium nitrate	Glycoprotein	95.6%	. NA	Ugbenyen & Okoh. (2013).
Bacillus subtilis	Soil	Cane molasses	Yeast extract	Polysaccharide	NA	4.92	Abdul-Razack <i>et al.</i> (2014).
Bacillus subtilis F9	Wastewater sludge	Sucrose	Peptone	Glycoprotein	NA	2.32	Giriet <i>et al.</i> (2015).
Bacillus toyonesis strainAEMREG6	Marine	Glucose	NH ₄ NO ₃	Glycoprotein	89.5	3.2	Okaiyeto <i>et al.</i> (2015a).
Bacillus velezensis 40B	Brackish water	Glucose	Yeast extract	Glycoprotein	99.9	3.54	Zaki <i>et al.</i> (2013).
Brachybacterium sp.	Freshwater	Maltose	Urea	Glycoprotein	91.2	NA	Nwodo <i>et al.</i> (2013a).
Brachybacterium sp.	Freshwater	Maltose	Urea	Glycoprotein	87.8	NA	Nwodo <i>et al.</i> (2013a).
<i>Cellulomonas</i> sp. Okoh	Freshwater	Glucose	(NH ₄) ₂ SO ₄	Glycoprotein	86.3	4.47	Nwodo & Okoh. (2013).
Chryseobacterium daeguense W6	Backwashing sludge	Glucose	Trypetone	Glycoprotein	96.9	NA	Liu <i>et al.</i> (2010).
<i>Citrobacter</i> sp. TKF04	Soil	Propionic acid and acetic acid	Yeast extract	Glycoprotein	85	0.2	Fujita <i>et al.</i> (2000).

Table 2.1, continued.

Microorganism	Source	Carbon source	Nitrogen source	Chemical composition	Flocculating activity (%)	Yield	Reference
<i>Cobetia</i> sp. OAUIFE	Marine	Glucose	Urea+Yeast extract+(NH ₄) ₂ SO ₄	Glycoprotein	92.78	NA	Ugbenyen <i>et al.</i> (2012).
Coryneobacillus glutamicum NA	NA	Corn steep liquor	Urea	NA	520U/mL	NA	He et al. (2004).
Enterobacte aerogenes	Soil	Fructose+Glucose	Urea+Yeast extract + (NH ₄) ₂ SO ₄	Glycoprotein	80	1.3	Lu <i>et al.</i> (2005).
Enterobacter cloacae WD7	Activated sludge	Glucose or sucrose	(NH4)2 SO4	Polysaccharide	105	2.27	Prasertsan <i>et al.</i> (2006).
<i>Halobacillus</i> sp. Mvuyo	Marine water	Glucose	Ammonium chloride	Glycoprotein	93	0.34	Cosa <i>et al.</i> (2012).
<i>Halomonas</i> sp. Okoh	Marine	Glucose	Urea	Polysaccharide	95	NA	Mabinya <i>et al.</i> (2011).
Halomonas sp.	Marine sediment	Glucose	Urea	polysaccharide	95	NA	Mabinya <i>et al.</i> (2011).
Klebsiella mobilis	Soil	Dairy wastewater+ Ethanol		Polysaccharide	95.4	2.58	Wang <i>et al.</i> (2007).
Klebsiella pneumoniae	Human saliva	Glucose	Peptone	Glycoprotein	96.5	4.7	Luo <i>et al.</i> (2014).
Klebsiella pneumoniae	Sputum	Glucose	Urea + Yeast extract+(NH ₄) ₂ SO ₄	Glycoprotein	98	NA	Zhao <i>et al.</i> (2013).
Klebsiella sp.	Activated sludge	Glucose	Yeast extract + Urea	Polysaccharide	86.5	3.52	Yang <i>et al.</i> (2012).
<i>Klebsiella</i> sp.	Activated sludge	Glucose	Yeast extract + Urea	Polysaccharide	86.5	1.8	Yang <i>et al.</i> (2012).

Table 2.1, continued.

Table 2.1, continued.								
Microorganism	Source	Carbon source	Nitrogen source	Chemical composition	Flocculating activity (%)	Yield	Reference	
Klebsiella sp. TG-1	Wastewater	Sucrose	Beef extract	Glycoprotein	86.9	NA	Liu <i>et al.</i> (2013).	
Klebsiella sp. ZZ-3	Sludge	Glucose	NaNO ₃	Glycoprotein	92.6	0.126	Yin <i>et al.</i> (2014).	
Methylobacterium sp.	Freshwater	Glucose	Urea+Yeast extract+(NH ₄) ₂ SO ₄	Glycoprotein	95	8.203	Ntsaluba <i>et al.</i> (2013).	
Methylobacterium sp.	Freshwater	Glucose	Peptone	Polysaccharide	72	NA	Ntsaluba <i>et al.</i> , (2011).	
<i>Micrococcus</i> sp. Leo	Marine	Glucose	Urea+Yeast extract	Glycoprotein	87.5	0.738	Okaiyeto <i>et al.</i> (2014).	
Paenibacillus elgii B69	Soil	Sucrose	Peptone+Yeast extract	Polysaccharide	87	25.63	Li et al. (2013).	
Paenibacillus mucilaginosus	Soil	Sucrose	Yeast extract	Polysaccharide	97	NA	Tang <i>et al.</i> (2014).	
Paenibacillus polymxya BY-28	Soil	Sucrose	Bean cake powder	Glycoprotein	99	NA	Gong <i>et al.</i> (2011).	
Penicillium purpurogenum	NA	Glucose	Yeast extract	Polysaccharide	96	6.4	Liu &Cheng. (2010).	
Rhodococcus erythropolis	Activated sludge	Livestock wastewater	NA	Glycoprotein	87.6	1.6	Peng <i>et al.</i> (2014).	
Serratia ficaria	Soil	Lactose	Yeast extract	Polysaccharide	95.4	NA	Gong <i>et al.</i> (2008).	
Solibacillus silvestris	Marine	Maltose	Yeast extract	Glycoprotein	90	1.7	Wan <i>et al.</i> , (2013).	

Table 2.1, continued.

Table 2.1, continued.								
Microorganism	Source	Carbon source	Nitrogen source	Chemical composition	Flocculating activity (%)	Yield	Reference	
Sorangium cellulosum	NA	Soluble starch	NaNO ₃	Glycoprotein	96.6	17.5	Zhang <i>et al.</i> (2002).	
Streptomyces and Brachybacterium species	Freshwater	Glucose	NH4 NO3	Polysaccharide	63.7	3.02	Nwodo & Okoh. (2014).	
Virgibacillus sp.	Marine	Glucose	Urea+ (NH ₄) ₂ SO ₄ +Yeast extract	Polysaccharide	91.8	2.43	Cosa <i>et al.</i> (2013a).	
Vagococcus sp.	Wastewater	Glucose	Urea+Yeast extract+(NH ₄) ₂ SO ₄	Polysaccharide	86.5	2.3	Gao <i>et al.</i> (2006).	

Table 2.1, continued.

2.7.2 Effect of Temperature

Temperature has a significant influence on cultivation and production of bioflocculant in microorganisms, as the enzymes in control of bioflocculant synthesis are activated at an optimal temperature (Li *et al.*, 2009; Zhang *et al.*, 2007). After the literature review, the study documented that the optimum temperature for the bioflocculant production differed at therange between 25 °C –37 °C (Salehizadeh & Shojaosadati, 2001). The optimal cultural temperature and shaking speed was 30°C, at 140 rpm for 24 h, were favoured for the mixed culture of *Bacillus sphaericus* F6 and *Rhizobium radiobacter* F2 to produce novel bioflocculant CBF with flocculating activity reached 98.52% (Li *et al.*, 2016). Nakata and Kurane (1999) found that the bioflocculant synthesised by *Citrobacter* sp. TKF4 was cultured at 30 °C temperature, which had a crucial effect on bioflocculant production. Meanwhile the optimal enzymatic reactions were frequently achieved at optimal temperature for microbial growth. Giriet *et al.* (2015), confirmed that *Bacillus subtilis* F9 produced bioflocculants with tremendous flocculating activity at optimum temperature 40 °C.

2.7.3 Effect of Inoculum Size

The inoculum size among many physiological properties plays an important role in metabolism processes, as well as a substantial effect on microbial cell growth and the synthesis of organic compounds (Gong *et al.*, 2008). Consequently, small inoculum size might extend the stationary growth phase. However, larger inoculum size can extremely overlap the function of the microbial multiplication, and thus reduce the production of the bioflocculant (Li *et al.*, 2009). For example, the inoculum size of 1% (v/v), was found to be the optimal size for X14 and support the adjustment of strain X1 for the production medium, reducing the logarithmic growth phase which activate the bioflocculant ZS-7

production (Li *et al.*, 2009). Also, 2% (v/v) inoculum volume of *Oceanobacillus* sp. Pinky was found to be the optimum value of production of bioflocculant (Cosa *et al.*, 2013b).

Similarly, the study showed that the best favourable inoculum volume was obtained at 2% (v/v) of *Micrococcus* sp. Leo for the bioflocculant production (Okaiyeto *et al.*, 2014). Besides, 3% (v/v) inoculum size of *Bacillus* sp. Gilbert, was favoured for bioflocculant production (Ugbenyen *et al.*, 2014). Moreover, the optimal culture condition, i.e. inoculum size of 4% (v/v) *Bacillus pumilus* strain ZAP028 was appropriate for the bioflocculant production (Makapela *et al.*, 2016). Similar result of Ntsangani *et al.* (2017), found that the optimum inoculum size of *Bacillus* sp. AEMREG4 was found to be 4% (v/v) for bioflocculant production and the flocculating activity achieved 80%. In addition, the optimum cultivation condition as inoculum size at 5% (v/v) was preferred by *Bacillus* sp. W7 for bioflocculant production (Okaiyeto *et al.*, 2016)

2.7.4 Effect of Initial pH

The initial pH is one of the parameters of the production medium that performed and the most significant role in bioflocculant production as well as flocculation activities (Zheng *et al.*, 2008). It also regulates the oxidation-reduction potential and electrical charge of the microbial cells, which may possibly affect the absorption of nutrients and enzymatic reaction in the fermentation medium (Salehizadeh & Shojaosadati, 2001).

The initial pH 7 of fermentation medium was found to be the optimal pH for bioflocculant production by *B. pumilus*, with high bioflocculating activity (Makapela *et al.*, 2016). For example, the optimised pH 7 was found to be the most favourable condition for bioflocculant production by *Halomonas* sp. Moreover, a similar study indicated that pH 7 was more appropriate for the production of bioflocculant by *Halobacillus* sp. Mvuyo (Cosa *et al.*, 2012; Mabinya *et al.*, 2011).

Moreover, Okaiyeto *et al.* (2016) found that the *Bacillus* sp. isolated from marine environment favoured a slightly acidic condition of fermentation medium (pH 6) for bioflocculant production and flocculating activity of 94.9 % was stretched at three days of cultivation. A different study showed that that *Aspergillus parasiticus* favoured acidic condition of fermentation medium for bioflocculant production (Deng *et al.*, 2003). Whereas, *Klebsiella* sp.TG-1 chose alkaline condition of pH 8 for the production of bioflocculant (Liu *et al.*, 2013). An extremely pH stable and more efficient bioflocculant was produced by *Rhodococcus erythropolis* by using a combination of pre-treated livestock and sludge wastewater as a cheap medium with the supplementation of Na₂ HPO₄ (Peng *et al.*, 2014).

2.7.5 Effect of Agitation Speed on Bioflocculant Production

The shaking or agitation speed regulates dissolved oxygen concentration that stimulates enzymatic reactions and facilitates nutrient absorption (Lopez *et al.*, 2003). Optimum shaking speed for the bioflocculant production by *Bacillus licheniformis* X14 was found between 140 rpm–160 rpm. Even so, the change in shaking speed prerequisite for various microorganisms can be the effect on the different oxygen demands at various microbial growth phases (Li *et al.*, 2009).

Similarly, research indicated that the difference in oxygen demand depends on the growth stage of BPMs and found that the demand of oxygen in the early growth phase was low which then increased as the *Bacillus licheniformis* entered the exponential and stationary phases. In contrast, Xia *et al.* (2008) found that the production of bioflocculant by *Proteus mirabilis* was significantly decreased at shaking speeds of lower or higher than 130 rpm. This may due to the oxygen demand of *P. mirabilis* which was limited to the oxygen concentration at 130 rpm shaking speed. The highest yield of bioflocculant

produced by *Klebsiella mobilis* was 2.58 g/L at optimal shaking speed at 120 rpm (Wang *et al.*, 2007).

2.7.6 Effect of Metals on the Bioflocculant Production

The bioflocculant production is affected via the chemical nature of metal ions supplemented in the production medium (Li *et al.*, 2009). For example, the production of the bioflocculant by *Flavobacterium* sp. was enhanced with the addition of Ca^{+2} , Mn^{+2} and Ba^{+2} , respectively, but inhibited by the presence of Mg^{+2} (Gonzalez & Hu, 1991). The cations, such Na⁺, Ca⁺², Mg⁺² and Fe⁺², also had no effect on production of bioflocculant ZS-7, whereas Cu⁺² extremely inhibited cell growth (Li *et al.*, 2009). For example, the production of bioflocculant by *Serratia ficaria* was activated by adding K⁺, Ca⁺², Ba⁺² and Fe⁺³ but was inhibited by the presence of Mg⁺² and Cu⁺² (Gong *et al.*, 2008).

Furthermore, the production of bioflocculants by *Proteus mirabilis* was influenced by the presence of Ca^{+2} , Mg^{+2} and Fe^{+3} , while the same study stated Na^+ , K^+ and Fe^{+2} have no evident correspondence with the production of bioflocculant, but Al^{+3} cation inhibits its production (Xia *et al.*, 2008). The cell growth and bioflocculant production of *Paecilomyces* sp. was induced by Ca^{+2} , while Cu^{+2} and Fe^{+2} ions were found to inhibit the cell growth (Tang *et al.*, 2014).

2.8 Factors Affecting Flocculation Efficiency of Bioflocculants

The flocculation performance of bioflocculants is influenced by chemical and physical factors, pH, bioflocculant concentration, temperature, metal ions and some other factors.

2.8.1 Effect of pH on Flocculating Efficiency

The flocculation performance of bioflocculants is commonly affected by the pH value of the suspended solution. Many studies investigated the effect of pH on the flocculating efficiency of bioflocculants to determine the optimal range. Some study concluded, that at acidic pH, both bioflocculant and kaolin particles were conceivably adsorbed to hydrogen ions (H⁺), which destabilised the formation of complexes between bioflocculant molecules and kaolin particles intermediated by Ca⁺². Likewise, hydroxide ions (OH⁻) interfered with the coagulation of the bioflocculant molecules and kaolin particles at high alkaline pH values, causing lower flocculation efficiency (Prasertsan *et al.*, 2006).

For instance, the maximum flocculation efficiency of bioflocculant produced by *B*. *firms* and *B. licheniformis* was observed at optimum pH that ranged from 7–9 (Karthiga & Natarajan, 2015). Makapela *et al.* (2016) indicated that the bioflocculant from *B.pumilus* was found to be thermos-stable with maximum flocculating activity in a wide range of pH 3–pH11. The determined flocculating efficiency of PSK1 bioflocculant was documented at pH 2.0, 94.56% with kaolin clay suspension concentrated at therange of 5 to 9 g/L (Ayat *et al.*, 2017).

Moreover, the flocculating activity of the produced bioflocculant CBF-F26 by a consortium strain of *Bacillus sphaeicus* F6 and *Rhizobium radiobacter* F2 was found to be the highest at neutral pH and weak alkaline conditions (Wang *et al.*, 2011). Meanwhile, the highest flocculating efficiency of the bioflocculant produced by *Enterobacter cloacae* WD7 was at pH 6.0 (Prasertsan *et al.*, 2006). A similar finding revealed that the bioflocculant was produced by *Aspergillus parasiticus* at a range pH 5– pH 6. Meanwhile, a higher pH may reduce cell growth and bioflocculant production, as well as its flocculation efficiency.

However, at low pH it was found that fungus growth and production, were significantly enhanced, as well as increasing of the bioflocculant yield in the fermentation broth medium (Deng *et al.*, 2005). The bioflocculant from a novel strain SW-2 sustain its flocculating activity with kaolin suspension over pH range of 3–9 and 20 °C – 80° C

(Zhong et al., 2014). Li et al. (2010) found that the flocculating efficiency of the bioflocculant synthesised by Agrobacterium sp. M-503 was appropriate at range pH 7pH12. Bioflocculant **REA-11** produced by Corynebacterium glutamicum CCTCCM201005 was found quite thermo-stable at a moderate acidic range of pH 3pH6.5 (He et al., 2004). While, the flocculation activity of the biopolymer produced by Gyrodinium impudicum KG03 was identified to be suitable at a range of pH 3 – pH 6 with the optimal efficiency documented at pH 4 (Yim et al., 2007). Besides, bioflocculant WF-1 secreted by E. aerogenes, was found to be stable and capable of flocculating the strong alkaline Trona suspension at a pH of over 13 efficiently (Lu et al., 2005). While other bioflocculant showed their flocculating activity between pH 3 - pH 8, such as B. licheniformis CGMCC2876 (Xiong et al., 2010). Moreover, the bioflocculant IH-7 synthesised by Aspergillus flavus, was found to have good flocculating activity and stable in a wide range of pH and temperature (Aljuboori et al., 2013).

2.8.2 Effect of Bioflocculant Concentration on Flocculation Efficiency

Many studies found that the flocculating efficiency increased as the concentration of bioflocculants increased to the optimal level, and thus high concentration of bioflocculant might involve more neutralisation site, strong bridging and optimum flocculating efficiency than the bioflocculant with a lower concentration or low molecular weight (Li *et al.*, 2008; Zhang *et al.*, 2010).

For example, the flocculating efficiency of bioflocculant secreted by *Bacillus* sp. F19 was achieved at above 90% at concentration range of 1–20 mg/L at pH 3 and the maximum flocculating efficiency was reached at bioflocculant dosage of 2 mg/L (Zheng *et al.*, 2008). Meanwhile, the flocculating activity of bioflocculant from *Sorangium cellulosum* was found to require 20 mg/L – 100 mg/L of bioflocculant concentration to reach over 90% in the presence of cations. Whereas, the flocculating efficiency of

bioflocculant produced by *Pseudoalteromonas* sp. increased as the bioflocculant concentration increased from 2 mg/L - 10 mg/L, and the highest flocculating activity was achieved at 10 mg/L of bioflocculant (Li *et al.*, 2008). Moreover, the maximum flocculating efficiency of bioflocculant produced by *E. cloacae* WD7 was detected at a dosage of 2 mg/L (Prasertsan *et al.*, 2006). However, the effect of bioflocculant concentration on flocculating efficiency is dependent on the type and concentration of cations if the bioflocculant is cation-dependent. Therefore, because more concentration of cation-dependent bioflocculants contain more carboxyl groups that need high cation concentration to neutralise the charge of the bioflocculant and kaolin particles (Wu & Ye, 2007).

2.8.3 Effect of Temperature on Flocculating Efficiency

Many researches have deliberated that the chemical composition of bioflocculant has a crucial role in the thermo-stability of bioflocculants. Bioflocculants compose only peptide or protein in its structure, which is commonly sensitive to heat, but those components of polysaccharides are thermo-stable. For example, bioflocculant produced by *B. licheniformis* CGMCC 2876 was found to be very stable in a wide range of temperature of 4 °C to 80°C and the flocculating activity kept over 90% was constant when the bioflocculant sample was heated for 30 min at 80 °C (Xiong *et al.*, 2010). The polysaccharides ofbioflocculant from *B. mucilaginosus* had flocculating activity of 85% to 89% at a temperature that ranged from 23 °C to 70°C, while decreasing significantly over 70 °C to 90°C (Lian *et al.*, 2008). Also, the bioflocculant WF-1 (polysaccharides) secreted by *E. aerogenes*, had its flocculating activity decreased by up to 22% when the bioflocculant was exposed to heat for 15 min at 100°C and decreased by 45% after 50 min (Lu *et al.*, 2005).
Furthermore, Hetero-polysaccharide bioflocculant (pKr) synthesised by *Kocuria rosea* BU22S exhibited strong flocculating activity at wide range of temperature from 0 °C to 100°C and pH 2 to pH 11 (Chouchane *et al.*, 2017). Bioflocculant secreted by *B. aryabhattai* strain PSK1 showed maximum flocculating activity of 92.8% at 50°C, while it was found thermally stable up to 80°C (Abd El-Salam *et al.*, 2017). Moreover, bioflocculant produced by *Bacillus* sp. was found to be thermostable with approximately 65.6% of residual flocculating activity reserved after heat exposure of the bioflocculant for 25 min at 100°C (Ugbenyen *et al.*, 2013).

2.8.4 Effect of Metal Ions on Flocculating Efficiency

Cations perform a dynamic role in bio-flocculation processes, in which they can enhance the flocculation efficiency via neutralisation and stabilisation of the negative charge residual /net surface charge of the functional groups on the bioflocculant; hence, stimulating the formation of bridges between the bioflocculant and suspended particles (Wu & Ye, 2007). According to Wang *et al.* (2011), cations have an energetic role in enhancing the adsorption of suspended particles on bioflocculant in reducing the distance among them and expanding the electrostatic attraction between the suspended particles and the bioflocculant molecules.

Furthermore, Prasertsan *et al.* (2006) mentioned that the carboxylate functional groups present in the bioflocculant chain are involved in the adsorption of the cation ions, which can form complexes between kaolin particles and bioflocculants. Li *et al.* (2016) indicated that bioflocculant active components from *Shinella albus* xn-1 comprised accumulated double bonds and triple bonds, metal ions, temperature and algal pH exhibited high influences on the flocculation efficiency. However, highest flocculating activity reached 85.65%. The maximum flocculating efficiency of exopolymer R-202 was detected in the presence of 10 mM solutions of Ca⁺², Mg⁺², Fe⁺² and Na⁺, but this effect depended on the type of ions and increased in the order - Na⁺ < Ca⁺² < Mg⁺²< Fe⁺². Meanwhile Fe⁺² ions showed the highest flocculating activity (Czemierska *et al.*, 2017). Bioflocculant produced by *Bacillus* spp. UPMB13 activity was affected by monovalent and divalent cations synergistic, including Na⁺, Ca⁺², and Mg⁺², while Fe⁺² and Al⁺³, causing inhibiting effects on flocculating activity. Divalent cations were revealed as the preferred cation source to improve flocculation efficiency (Zulkeflee *et al.*, 2012). The highest flocculating activity of the bioflocculant was produced by aquatic bacteria *Oceanobacillus*sp. Pinky, was revealed by using cations such as aluminium chloride and calcium chloride (Cosa *et al.*, 2013b).

When the bioflocculant is protein in nature, that is rich in amino acids, and contain carboxylate functional groups might provide negative charges of the suspended particles, which activates cations to enhance neutralisation, and create bridges between the bioflocculant and suspended particles (Cosa *et al.*, 2013a). The occurrence of extra functional groups, such as carboxyl groups on the bioflocculant molecule, can act as cation binding sites for the cations, and thus, by adding of these cations to a suspension might enlarge the floc size, causing better sedimentation (Li *et al.*, 2007). The flocculation competence of protein bioflocculants produced by *Alcaligenes cupidus* and *Rhodococcus erythropolis* revealed that it was more efficient to add Al⁺³ and Ca⁺², respectively (Banks *et al.*, 2006).

Moreover, the flocculating efficiency of the bioflocculant MBF3-3 secreted by *Bacillus* sp. was found to be more appropriate with the addition of the metals, such as K⁺, Na⁺ Mg⁺², Ca⁺² and Al⁺³ ions, but become suppressed by the addition of Fe⁺³ ions (Feng & Xu. (2008). A similar result was detected by Zheng *et al.* (2008) found that flocculation efficiency was totally inhibited by the addition of Fe⁺³ to the bioflocculant produced from *Bacillus* sp. F19 and kaolin suspension. Divergent to the above finding, the flocculation

efficiency of bioflocculant p-KG03 was enhanced with the addition of Fe⁺³ with analogous results displayed by the bioflocculant produced from *Enterobacter* sp. BY-29 (Wu & Ye, 2007). However, the novel bioflocculants produced by *Bacillus* sp. F19, *Citrobacter* sp. TKF04 and *Gyrodinium impudicum* KG30 did not require metal ions for their flocculation efficiency (Zheng *et al.*, 2008).

The flocculating activity of bioflocculant produced from haloalkalophilic *Bacillus* sp. was remarkably induced by the addition of divalent cations such as Ca^{+2} , Zn^{+2} and Cu^{+2} , respectively (Kumar *et al.*, 2004). While, flocculating competence of the bioflocculants from *Halomonas* sp. V3a was enhanced in the presence of Ca^{+2} at a wide range of pH 3– pH11, increasing in flocculating efficiency of above 80% against kaolin clay suspension (He *et al.*, 2010). Similarly, the flocculating activity of the bioflocculant secreted by *Serratia ficaria* achieved 95.4% in the presence of Ca^{+2} and Mg^{+2} , respectively, at pH 5– pH 7 range against kaolin suspension.

2.9 Cost-Effective Substrates for Bioflocculant Production

In recent years, bioflocculants have acquired vast scientific and biotechnological consideration because of their biodegradability, the non-hazardous nature of their decomposing products and prospect demand (Nwodo *et al.*, 2013). To reduce the production cost, challenges were engaged to search for lower-cost substrates. Some considerable determination to reduce the relative cost of producing bioflocculants at industrial large scale was to use economical substrates (Fujita *et al.*, 2000). Low-priced substrates were used for bioflocculant production (Zhuang *et al.*, 2012). For instance, activated sludge was applied as a raw substrate for the bioflocculant production (Guo *et al.*, 2014; Peng *et al.*, 2014). Moreover, different industrial wastes were utilised as a cheap carbon source substrate to reduce production cost, such as potato starch waste effluent (Pu *et al.*, 2014; Liu *et al.*, 2015b), palm oil mill effluent (Wong *et al.*, 2012; Aljuboori

et al., 2014), and brewery effluent wastewaters (Zhang *et al.*, 2007). The comparison in costs of commonly utilised substrates, such as glucose, sucrose, maltose, fructose and galactose, sustain an adverse impact on manufacture budgets, and thus confine the marketing potential of these bioflocculants (Mabinya *et al.*, 2012). Moreover, determinations have also concentrated on the isolation of new microbial species for novel bioflocculant production competences in exploiting economical substrates and improving the production medium ingredients and fermentation parameters condition that is required to enhance bioflocculant yield (Sathiyanarayanan *et al.*, 2013).

2.9.1 Molasses as Substrate

Molasses is a secondary product derived from sugar-cane manufacturing process that contains about 50% (w/w) of total carbohydrate, nitrogenous compounds and vitamins (Moosavi-Nasab et al., 2010). Therefore, it has tremendous properties that can be utilised as a potential-substrate for microbial growth and bioflocculant production (Zhuang et al., 2012). Besides, the sugar-cane molasses is a strong liquid with some distinctive properties, including, high concentration of biochemical oxygen demand (BOD) range of (40000-60000 mg/L) and chemical oxygen demand (COD) concentrations range (80000 mg/L –120000 mg/L), this effluent fluid requires treatment before discarding to avoid environmental contamination. The hydrolysates of agricultural wastes, rich in polysaccharides, were used to produce biological products (Zheng et al., 2005). It supports the production of bioflocculants by Klebsiella sp. after neutralisation without the addition of any nutrient supplements, which makes the commercial production considerably cost effective. Many bioflocculant-producing microbes discovered previously exploited carbohydrate-rich compounds as an energy source and the major source of carbon (Ugbenyen et al., 2012). For example, molasses is a low-cost substrate that can be recycled as an enriched medium through various microorganisms for exopolymer compounds production (He et al., 2004). Similarly, molasses was used as a

production medium for *Penicillium* sp. HHE-P7 to produce bioflocculant, and its highest flocculating efficiency was achieved after three days of cultivation (Liu *et al.*, 2010). Sam *et al.* (2011), stated that pre-treated molasses was used as fermentation substrate in the bioflocculant production by halophilic bacteria. According to Mao *et al.* (2010) *Pseudomonas alcaligenes* PS-25 and *Pseudomonas fluorescens* C-2, were cultivated for three days in molasses substrate for bioflocculant production. However, the bioflocculant REA-11 from *Corynebacterium glutamicum* CCTCC M201005 was enhanced by the addition of glucose, sucrose and fructose, but sucrose was found to be more favoured as the sole carbon source, due to its low-cost and high bioflocculant yield (He *et al.*, 2004). Therefore, the capability of microorganisms to utilise sucrose may give rise to potentiality to consume molasses as a carbon source on large-scale production; hence, is promising for producing bioflocculant commercially.

2.9.2 Palm Oil Wastewater

The palm oil industry is one of the most important industries in Malaysia with a production of 20.7 million tonnes of crude palm oil (CPO) from an oil palm planted area of 5.39 million hectares in 2014 (Kamalrudin and Ramli, 2014). Although, such production has resulted in a large amount of palm oil mill effluent (POME), evaluated at approximately three times the quantity of CPO. One tonne of the fresh fruit bunches (FFB) processed, an approximate of 0.67 tonnes of POME are produced. Therefore, in 2014, about 67.28 million tonnes of POME were produced. POME is a highly polluted wastewater since it has high biological oxygen demand (BOD) and chemical oxygen demand (COD) (Nurul *et al.*, 2014). Also, POME has a variation of microbial classes as indicated by the BPM. Furthermore, palm oil biomass waste contains high organic components, such as cellulose, hemicellulose, lignin and extractive (Hashim *et al.*, 2011; Tsai *et al.*, 2009)

2.10 Application of Bioflocculant

Recently, the investigation of potentiality extracellular polymeric substance (EPS) exploitation was extremely amplified due to its various distinctive properties that recommend its possibilities, applications in many industrial processes (Elkady *et al.*, 2011). For example, EPS are applied in many industrial processes, due to their unique bio-physicochemical properties, such as, in the manufacturing of pharmaceuticals, cosmetics, detergents, food additives as thickening, emulsifying and stabilising agents, textiles, adhesives as well as brewing (Kunmani *et al.*, 2011; Mishra and Jha, 2013). In addition, EPS can function as antioxidant, natural immunomodulatory, a drug carrier agent in wastewater treatment, oil recovery, dredging, in different downstream processes, bioflocculant and heavy metal removal (Wang *et al.*, 2008).

Lin and Zhang, (2004) indicated that some of these biopolymers have been designated to have anti-inflammatory, anti-viral and anti-tumour properties, which function as inducers for colony stimulating factors and interferon. Among recognised bioflocculant, polysaccharides attract the consideration of scientists in flocculating processes, mostly in drinking water purification (Raza *et al.*, 2011). Bioflocculants are also widely applied in wastewater treatment, e.g., in the treatment of dye solutions and inorganic solid suspension, such as activated carbon, aluminium oxide, solid clay, and bentonite (Deng *et al.*, 2005; Shih *et al.*, 2001; Yim *et al.*, 2007). Sathiyanarayanan *et al.* (2013) reported that bioflocculant production rich in polysaccharide is not type specific, due to each microbial strain of the same species may be producing different types of carbohydrates in the fermentation medium accompanied by many biological functions. In addition, as polysaccharides have a hydroxyl group, with a hemiacetal reducing mechanism, with other capability and suitability that could have crucial roles in the reduction reactions (Mata *et al.*, 2009).

2.10.1 Treatment of Wastewater

2.10.1.1Removal of COD and Suspended Solids from Wastewater

Several studies had proved the effectiveness of bioflocculants in the removal of COD, suspended solids, heavy metals and humic acids from wastewater discharged to streams, microorganisms, latex particles and separation of oil from oil-water suspension suspended solids, COD, humic acids, heavy metals from waste streams, microorganism, latex particles and separation of oil from oil-water suspension and fine coal procedures (Zemmouri et al., 2011). The majority of the bioflocculant documented in literature have shown satisfaction in flocculating efficiency against kaolin suspension. However, most bioflocculant display different flocculation capabilities for further suspended materials in aqueous solution. Likewise, bioflocculant rich in carbohydrate showed an exceptional flocculation efficiency in removing suspended solids from starch wastewater (Deng et al., 2003). The bioflocculant from Aspergillus latus has the potential to flocculate oil emulsion (Kurane et al., 1991). Whereas, bioflocculant from Paenibacillus elgii B69 strain display a good flocculating capability in treating various wastewaters which comprise colour removal (88%), turbidity reduction (83%) and COD reduction (68%) (Li et al., 2013). Also, the bioflocculant from Serratia ficaria was found to have significant flocculating activity against kaolin suspension as well as presenting high flocculation efficiency in different wastewater (Gong et al., 2008). For example, river water was treated with bioflocculant SF-1 secreted by Serratia ficaria, and the flocculation efficiency removal has shown 90.4% of colour removal, 87.1% of COD reduction, and 84.2% of turbidity removal (Gong et al., 2008). Similarly, bioflocculant SF-1 thatwas used to treat brewery wastewater, revealed efficiency in removing turbidity (91.8%), COD (80.7%). Meanwhile for meat processing, wastewater, the turbidity reduction was 93.7%, COD removal was 76.3% and colour removal was 64.1% of soy sauce brewery waste-water (Gong et al., 2008). The bioflocculant MBFA9 from Bacillus mucilaginosus showed a high flocculating efficiency of suspended solids with a reduction percentage of 85.5% and 68.5% of COD (Deng *et al.*, 2003). Furthermore, bioflocculant MBF-6 secreted via *Klebsiella pneumoniae* YZ-6, which was isolated from human saliva, had been used for flocculating at various wastewaters, such as dairy, sugar, textile and the brewery industry effluents (Zhengshan *et al.*, 2014). The highest flocculating activity was detected in sugar effluent wastewater, whichexhibited maximum removal of COD (77.8%) and BOD (80.7%), while the reduction of suspended solids was about 78.6% (Zhengshan *et al.*, 2014). However, the bioflocculant produced by *Bacillus mucilaginosus* had a BOD reduction percentage of 42.3% and 74.6% of the COD for domestic wastewater, whereas the removal efficiency of the brewery wastewater standard was 77.4% of the BOD and 70.5% of COD. In addition, its removal efficiency of BOD and COD from pharmaceutical wastewater was 41.7% and 66.2%, respectively (Lian *et al.*, 2008).

2.10.1.2 Drug Removal

Pharmaceuticals are biologically active and persistent constituents, which are identified as a permanent threat to environmental safety (Santos *et al.*, 2010). Figure 2.4 showed.



Figure 2.4: Representation for Source and the Fate of Pharmaceutical in the Environment (Reproduced with permission from Elsevire)

Pharmaceuticals are applied widely in human and veterinary medicine to prevent diseases and as well as growth promoters in livestock and agriculture such as fish farming. After utilisation, pharmaceuticals might be conjugates of sulphuric and glucuronic acid (Heberer *et al.*, 2002; Nikolaou *et al.*, 2007). The use of pharmaceuticals and personal care products is strongly increasing, and these are classified as organic micro-contaminants that are relatively recalcitrant and slowly bio-accumulating and produce permanent damage to the ecosystem in time (Xing *et al.*, 2013).

Jelic *et al.*, (2011) reported that some pharmaceuticals are residues in wastewater. This showed that even if it has a good reduction percentage level achieved in the aqueous phase, for example, comparison of influent streams and effluent wastewater concentration cannot imply degradation of similar degree. Generally, the removal of most of the substance is inadequate and the improvement of the wastewater treatment which can precede treatment of the produced sludge which is necessary to prevent the initiation of

these micro-contaminants in the environment (Jelic *et al.*, 2011). Similarly, some of the common pollutants, such as pesticides, cleansing agent, petroleum among other drugs, are constantly distributed, even at low extent, could influence by increasing the toxicity even without persistence estimate (Santos *et al.*, 2010). The most evident pathway for environmental pollution of pharmaceutical is by the untreated discharge of feaces and urine, while another anthropogenic mechanism must be expected. For example,

- Household disposal, any topical formulation or unused medicine is discarded via the sink/toilet or through waste collection before they are actually reserved to landfill sites, where they might appear as a terrestrial ecosystem pollutant. Instead, they probably leak into adjacent water bodies (Bound *et al.*, 2005).
- Diagnostic composites, such as X-ray contrast media are directly discarded in its natural form.
- 3. Metabolism post-ingestion, whereby many drugs are metabolised as the organism attempts to transform hydrophobic compound into more simple excreted polar residues (Timbrell. 2002).
- Impacts due to anthropogenic activities, for example, sewage treatment plant sludge, which can convey non-suspected drugs and is commonly employed as a fertiliser on agricultural fields (Topp *et al.*, 2008).

Veterinary medicines are excreted in urine and faeces by animals before being spread onto land through feed application as fertilisers. Besides the potential for direct soil contamination, there is also a risk of heavy run-off. Therefore, possibly contaminate both the surrounding surface and groundwater (Kay *et al.*, 2005). Another important source of environmental contamination of pharmaceuticals is the effluents of pharmaceutical production plants (Larsson *et al.*, 2007). Removal efficiencies in WWTPs depend on various factors, such as composite physicochemical properties, the environment conditions e.g. sunlight intensity and temperature and the type of treatment process applied, the operational conditions of the treatment process, such as temperature of operation, redox conditions, solids retention time and hydraulic retention time as well as the phase of the activated sludge used in the plant system (Castiglioni *et al.*, 2006; Suárez *et al.*, 2008; LeMinh *et al.*, 2010). So, removal efficiencies can vary significantly from plant to plant and within a plant at different time periods (Vieno *et al.*, 2007). Sulfamethoxazole is a medicine that is generally used in animal food and animal droppings that contaminate water bodies. The bioflocculant synthesised by *Klebsiella* sp. displayed significant adsorption ability for sulfamethoxazole in aqueous solution; under optimised conditions (Xing *et al.*, 2013).

2.10.1.3 Heavy Metal Removal

The heavy metal contamination is a serious environmental issue, and the use of bioflocculants was shown to perform a significant role in the removal of heavy metals from contaminated water. The bioflocculants possess various anionic groups that help in binding metal ions. The metal adsorption by bioflocculant depends on several factors, such as pH, initial metal concentration, temperature, bioflocculant dosage, charge density and type of conformation of polymer with adsorbed ions (Gomma, 2012; Lin & Harichund, 2011a, 2011b; Morillo *et al.*, 2006; Salehizadeh & Shojaosadati, 2003; Wang *et al.*, 2013b). Several industrial manufacturing procedures evolved in the discharge of heavy metals into aquatic ecosystems. This has been demanded suitable consideration due to the adverse effects of these heavy metals in the marine environment (Salehizadeh & Shojaosadati, 2003). Heavy metals, for example, arsenic, lead, mercury, copper, cadmium, nickel, zinc and chromium are the main noxious waste of fresh water reservoirs because of their hazard, non-biodegradable, and nature persistence (Azmi *et al.*, 2017; Kuniawan *et al.*, 2002). The development industries are the key source of heavy metals,

leading such chemicals into different environmental parts, including water, soil, air and biosphere. For example, sea foods and vegetables will easily absorb heavy metals due to their high solubility in the aquatic environment. Therefore, they can accumulate into aquatic biota and the human body by means of the food chain (Arezoo et al. 2017). Moreover, they can combine with sulfhydryl groups of protein, inhibiting the enzyme activity, the existences of accumulation of heavy metals in polluted ecosystem threaten both human and livestock health and merit distinctive responsiveness (Cobbing, 2008; Cristina et al., 2011). Therefore, the strong affinity of arsenate to methanetheiol (mercaptan) groups present in living organism, for example, amino acids, peptides and proteins containing some enzymes, which can cause severe toxicity to humans (Altumn et al., 2014). Copper frequently originates in high concentration near minefield and disposal, waste sites which originate from electroplating and metal finishing industries. Copperiedus causes itching and dermatisation, hand plane warts (keratinisation) and allergic contact dermatitis in human beings (AL- Asheh et al., 2003). Aydin et al. (2008) reported that the inhalation of copper spray can cause lung cancer among unprotected personnel. On the other hand, lead is one of the most dangerous heavy metals since it gets into the human body. Zhu et al. (2008) indicated that upon its dissemination into the body, lead causes severe damages on the physiology of the organs. For instance, it can impair the erythrocytes and reduce their capability of oxygen transport to the biological human system. Also, its effect can harm the kidneys, hearing and nervous system. Grandjean, and Landrigan. (2014) demonstrated that lead compound is identified as an enzyme inhibitor and metabolic poison, and they showed that foetuses and young children are more subjected to poisonous lead ions that can threaten their health.

Cadmium is presented into the water from the various manufacturing processes such as: mining, metal plating, phosphate fertilizers, nickel batteries, composite industries, sewage sludge, dyes and stabilisers (Zhu *et al.*, 2008). The destructive effects of cadmium comprise serious and chronic metabolic syndromes, such as emphysema, hypertension, renal damage, itai-itai disease and testicular atrophy. Therefore, it is necessary to remove heavy metal polluted wastewaters before they are discharged into the aquatic environment (Kurniawan. 2002).

Environmental system as abiotic methods, have been applied for the removal of heavy metals from polluted locations, for instance, activated carbon adsorption, reverse osmosis, chemical precipitation and ion exchange. The escalating catastrophe of heavy metal contamination of water, soil and some other sediment has setthe search for changes to remove these pollutants as priority. Many researches on heavy metal removal from wastewater and petroleum had highlighted the development of substance which can undergo improved selectivity and affinity and volume for targeted metals (Parirandeh et al., 1998). The treatment of industrial wastewater effluent contaminated with noxious heavy metals is very important from the perspective of environmental contamination control (Guangyu & Thiruvenkatachari, 2003). The EPS produced by B. subtilis exhibited chromium (VI) adsorption of 48% (Chug et al., 2016). The bioflocculant from P. aeruginosa showed the adsorption ability for Hg⁺ (89.09%) and Cu⁺² (87.39%) at a bioflocculant concentration of 20 mg/L; Cd ⁺² (79.93%) and Pb ⁺² (79.7%) at 40 mg/L and Zn^{+2} (80.5%) and Arsenate (72.92%) at 60 mg/L (Gomma, 2012). Many strains that were screened from Pseudomanas sp. Paenibacillus sp. and Herbaspirillium sp. have the capability and tolerate the adsorption of heavy metals. For example, the bioflocculants from *Pseudomonas* sp. showed that over 90% of Pb^{+2} and 78% of Hg^+ was adsorbed. Another example, the efficient removal of bioflocculant from Paenibacillus sp. CH11 showed a significant removal of about 90% of Cd⁺² (Lin & Harichund, 2011a). Similarly, a finding showed that the lead (Pb^{+2}) tolerant Achromobacter sp. revealed that it can tolerate up to 1500 mg/L of Pb⁺². It has revealed tremendous lead adsorption from effluent and water and showed 95% flocculating efficiency (Batta et al., 2013) As reported by

Geddie and Sutherland (1993), bacteria producing compounds, such as cell wall components and extracellular polysaccharide (EPS) that can adsorb heavy metals and EPS has been recognised to show a significant role in monitoring heavy metal pollution in the sewage treatment processes. Ozdemir et al. (2003) reported that the EPS secreted by different microorganisms documented in the literature was found to be acidic polysaccharides with various carboxyl functional groups that carried negative charges that can adsorb the metal ions. The bioflocculant from Bacillus firmus can adsorb approximately 98.3% of Pb^{+2} , 74.9% of Cu^{+2} and 61.8% of Zn^{+2} from synthetic wastewater (Salehizadeh & Shojaosadati, 2003). Also, the maximum removal efficiency of bioflocculant secreted by Paenibacillus validus MP5 for heavy metals was revealedas Zn⁺²(27%), Ni⁺² (16%), Cd⁺² (15%), Cr⁺² (9%) and Pb⁺² (7.5%) (Rawat & Rai, 2012). Meanwhile, the bioflocculant from P. elgii showed the highest removal efficiency for aluminium (A1⁺³) at 72% and good removal efficiency of Pb⁺² (60%), Cu⁺² (53%) and of Co^{+2} (49%). Aluminium (Al⁺³) has multi-valent ions, so enhanced affinity binding to the EPS. As indicated, since the bioflocculant have a strong capability for interaction with heavy metals, they are suggested as surface active agents for the removal turbidity, heavy metal and bacterial population from their industrial wastes (Lin & Harichund, 2012). Consequently, the investigation of novel technologies for the treatment of industrial sewage has become significant. Bioflocculant MBF4-13 has shown removal efficiency of dichromate ion (Cr₂O₇⁻²) about 69.3% and the Nickel (Ni⁺²) 19.2%, because the bioflocculant MBF4-13 primarily composed of polysaccharide that comprised hydroxyl groups in the molecular sequence which could simply form hydrogen bonds with $Cr_2O_7^-$ ², which cause the highest removal efficiency of Ni⁺² (Gao *et al.*, 2009). Moreover, the novel bioflocculant MBF-TG-1 from *Klebsiella* strain TG-1 has a flocculating activity of almost 86.9% for Trona suspension. Many designated researches had revealed that bioflocculant from *Paenibacillus* have the capability of removing heavy metals from

water (Mokaddem *et al.*, 2009; Rawat & Rai, 2012). Recently, studies have described that these bio-polymers display a significant potentiality to adsorb heavy metals because of the presence of functional groups, such as carboxyl, amino, hydroxyl, sulfhydryl, phosphoryl and phenolic groups (Chug *et al.*, 2016).

2.11 Synthesis of Nanoparticles

The area of nanotechnology research is emergent as a progressive technology, that is interdisciplinary with biology, chemistry, physics, material science and medicine. There are a huge number of chemicals, physical, biological and hybrid methods available to synthesise different types of nanoparticles (NPs) (Gudikandula & Maringanti, 2016). Meanwhile physical and chemical methods are more prevalent in the synthesis of nanoparticles, but the use of toxic chemicals greatly limits their biomedical applications, especially, in clinical fields. Therefore, biogenic synthesis of nanoparticles offers an attractive alternate to chemical synthesis methods. Various hazard free, eco-friendly methods of synthesis of silver nanoparticles are in operation (Gudikandula & Maringanti, 2016).

Silver nanoparticles have various applications, for instance, they can be used as spectral coating for solar energy absorption and intercalation material for electrical batteries, optical receptors, catalysts in chemical reaction, for bio-labelling and as antimicrobials (Mokhtari *et al.*, 2009). Furthermore, synthesis of silver nanoparticles by *Pseudomonas fluorescens* was identified and the synergistic effect of the biosynthesised nanoparticles with common antibiotics concerning microbial drug resistance was detected (Marsili & Das, 2016). Maharani *et al.* (2016) detected the cytotoxicity effect of biosynthesised silver nanoparticles by *E. coli.* as well as their potential for anticancer application. Also, the silver nanoparticles are used to control mosquito larvae responsible for the transmission of dengue, malaria, Zika virus and other serious diseases. Moreover,

due to their distinguished antimicrobial competence, silver nanoparticles can be applied as free radical scavengers (Marsili & Das, 2016). In addition, the maximum antibacterial activity of silver nanoparticles synthesised by biological and chemical method was revealed in Staphylococcus aureus and E. coli. (Gudikandula & Maringanti, 2016). Raveendran et al. (2013) found that Halomonas maura produce a good extracellular acidic bioflocculant, comprising of glucose, galactose, mannose and glucuronic acid was applied to synthesise Mauran/Chitosan nanoparticles in the size range of between 30 nm and 200 nm. The bioflocculant contains polysaccharides are considered favourable applicants for the synthesising of nanoparticles and essential role as reducing and stabilising agents (Salehizadeh et al., 2012; Raveendran et al., 2013). The bioflocculant from B. subtilis was applied to synthesise silver nanoparticles in the size range of 60 nm with stability of 5 months (Sathiyanarayanan et al., 2013). Zaki et al. (2014) found that the bioflocculant secreted by B. mojavensis strain 32A produced silver nanoparticles of 7 nm-72 nm size. Mokhtari et al. (2009), found that the bioflocculant synthesised by Klebsiella pneumonia, effectively synthesise consistently scattered silver nanoparticles with a uniform size and shape in the range of 1nm–6 nm with an average size of 3 nm.

2.12 Conclusion and Future Prospects

Synthetic chemical flocculants are effective at flocculating colloids and are broadly applied in several industrial processes. Because of their adverse health influence and the environmental hazards related with chemical flocculants, microbial flocculants have acquired enormous scientific and biotechnology planning due to their safety and biodegradable characteristics. Aquatic habitation, which provide a very rich biodiversity of marine microorganism remain under discovered for this determination, and thus bear tremendous potential as reservoirs of novel bioflocculant producing organisms. Meanwhile many bioflocculants were recounted in literature, their large-scale production is quite insufficient by high production cost, low flocculation efficiency and low yield. Optimisation of culture medium components and fermentation condition parameters are the most significant approaches to increase on bioflocculant yield and flocculation efficiency. Still, the high cost of production media components, would make it very favourable to exploit cost effective substrate for their large scale production in bioflocculant industry. Moreover, the application of consortium microorganism for bioflocculants production might keep the enhanced flocculation efficiency and larger bioflocculant yields than pure strains is an essential. Moreover, some cation independent bioflocculants were recognised and reported, and thus registered in literature. Consequently, more research studies are necessary to explore a novel cation independent bioflocculant with higher flocculation efficiency.

CHAPTER 3: MATERIALS AND METHODS

3.1 Experimental Design and Sampling Sites

The overall experimental design of the production of bioflocculant named QZ-7 from *B. salmalaya* 139SI, characterisation and application are illustrated in this chapter. Its flocculation performance and its application in surface water and industrial wastewater treatment and synthesis of nanoparticles are shown in Figure 3.1.

Influent and effluent wastewater samples was collected from a wastewater treatment plant as shown in Table 3.1, using a homemade glass sampler. Then, the samples were kept in clean 1Liter amber glass bottles. The bottles were stored in an icebox at 0°C while the samples were transported to the laboratory. Then the samples were micro- filtered using 1.2 μ m glass fiber filter (Whatman, Maidstone, UK). The filtered samples were preserved by addition of 1g/L of sodium azide to prevent microbial degradation (APHA 2005).

Sample source	Place	GPS location		
River	Kaiana	2°59'40"N 101°47'02"E		
Water	Kajalig			
Ethanol mill	Parlia Malaysia	6°47'74'77"N 100°26'0712"E		
wastewater	Ferris: Malaysia			
Sugar mill	MSM Parila SDN PHD	5°37'4302"N100390499"E		
wastewater	WISIWI FEITIS SDIN.BHD			
Rubber mill	Shorubber SDN.BHD	6°44'07''N 100°22'834''E		
wastewater	Perils Malaysia			
Hospital	University of Malaya	3°06'46"N 101°39'11"E		
wastewater	Oniversity of Malaya			

Table 3.1: Sampling Sites and their GPS Locations.



Figure 3.1: Overall experimental design.

3.2 Chemicals and Reagents

Chemical material used in this study are ammonium chloride, potassium carbonate, potassium dihydrogen phosphate anhydrous, magnesium sulphate, sodium chloride and sodium hydroxide (Friendemann Schmidt), arsenate, copper chloride, cadmium chloride, lead acetate, carbazole, yeast extract, zinc sulphate and Folin-Ciocaltens phenol reagent (Merk, Germany), Di potassium hydrogen orthophosphate anhydrous (Fisher, UK), glucose (Systerm), sucrose (UNIVAR, Australia), maltose, lactose and starch (Sigma), potassium dichromate (R & M Chemical, UK). The European Pharmacopoeia (EP) reference standard materials for mefenamic acid, ibuprofen, and perindopril as well as caffeine, diclofenac, simvastatin, and nifedipine were purchased from Sigma-Aldrich (Schnelldorf, Germany). The purity of all standards used in this study was ≥98%. The HPLC-grade ethanol, methanol and acetonitrile, formic acid, ammonium formate, acetic acid, and ammonium acetate (LiChrosolv) were supplied by Fisher Scientific (Loughborough, UK). The Ultra-pure water was prepared from a Milli-Q water purification system (MA, USA). All other reagents were of analytical grade.

For composition of the used Media, there are different types of media were used for screening and production of the produced bioflocculant. These media are cultivation media, seed media and production media.

1. The cultivation media for slant and subculture such as Brain Heart Infusion Agar (BHIA) has approximative formula g/L; brain heart infusion 6.0g, peptic digest of animal tissue 6.0g sodium chloride 5.0g, dextrose 3.0g, pancreatic digest of gelatine agar 14.4g, disodium phosphate 2.5g and agar 15.0g and the final pH 7.4 ± 0.2 .

- 2. The seed medium has constituents in g/L as follows; glucose 10g (C source), yeast extract 1.5g and urea 1.5g (N source), KH₂PO₄ 0.1g, NaCl 0.1g and MgSO₄,7H₂O 0.2g. The pH was adjusted to 7.0 ± 0.2 .
- Production medium has components in g/L as follows; sucrose 20 g (C source), yeast extract 1.5g and urea 1.5g (N source), KH₂PO₄ 0.1g, K₂HPO₄
 0.1g, NaCl 0.1g and 0.2g MgSO₄.7H₂O 0.2g. The initial pH was adjusted to 7.0 ± 0.2 (Xiong *et al.*, 2010).

3.3 Cultivation and Isolation of the Bacteria

The *B. salmalaya* strain 139SI was obtained from the Molecular Bacteriology and Toxicology laboratory at the University of Malaya. The bacterial isolate was originally isolated from soil samples in a private farm located at 2.99917 °N and 101.70778 °E in Selangor, Malaysia. The bacterium was identified as *B. salmalaya* strain 139SI and deposited in Gen Bank KM0511837 (Ismail & Dadrasnia, 2015). The selected strain was streaked on blood agar medium plates and incubated at 37 °C for 18-24 h. The bacterial colonies were 2-3 mm in diameter, large and white-grey with a rough and irregular edge and shows strong β - haemolytic activities were obtained and sub cultured on slant tubes of brain heart infusion agar. The subcultures were incubated under aerobic conditions at 37°C for 24 h. This bacterial strain was consistently cultivated on nutrient agar and preserved in glycerol solution (20% w/v) suspended at -80 °C.

3.4 Screening for Bioflocculant-Producing Bacteria

Fifteen colonies from fresh culture of *B. salmalaya* strain 139SI were inoculated into 15 set of McCartney bottles each contains 10 mL seed medium. The pre- cultured bottles were incubated at 37 °C and shaken at 150 rpm for 24h. Subsequently, 2% of each bacterial culture broth was seeded into 100 mL of fermentation medium. The fermentation

bottles were kept incubated at 37 °C for 24 h with shaking at 150 rpm. For bacterial cell separation, the cell-free supernatant was harvested via centrifugation at 4000 rpm for 30 min (Yang *et al.*, 2015). The flocculating activity was measured in order to select the best strain to produce the bioflocculant.

3.5 Determination of Flocculating Activity

To select the best strain that produced bioflocculant, flocculating activity was determined from the cell-free supernatants. The flocculating activity was investigated using a suspension of kaolin clay. The suspension was prepared by mixing 4.0 g kaolin clay in 1.0 L of distilled water (Kurane & Nohata, 1994). A mixture of 95 mL of kaolin suspension with 3 mL of 1.0% calcium chloride (CaCl₂) solution and 2.0% (v/v) of cellfree supernatant was prepared. The mixed solution was vigorously agitated and left to settle at room temperature for 5 min. The optical density (OD₅₅₀) of the obtained clarified solutions determined via spectrophotometric 550 (UV-1700 was at nm spectrophotometry, SHIMADZU). A control sample was prepared in the same way, except the cell-free supernatant was replaced with unfermented broth media. The flocculating activity was calculated using the following expression (Kurane & Nohata, 1994):

Flocculating activity (%) =
$$\frac{Ac-Bs}{Ac} \times 100$$
 (3.1)

Where A_c and B_s represent the OD of the control and real samples, respectively.

3.6 Optimisation of Cultural Conditions of *B. salmalaya* 139SI for Bioflocculant Production

The bioflocculant production is affected through many factors, such as the constituents of the culture medium and culture conditions (He *et al.*, 2004; Nakata & Kurane, 1999).

The effects of the key factors, such as cultivation time, initial pH of the production medium, carbon source, nitrogen source, cultivation temperature, shaking speed, static conditions, metal salts and inoculum size, were investigated in the optimisation of the culture condition of *B. salmalaya* 139SI strain for bioflocculant production. Then the flocculating activities which indicate the highest level of bioflocculant production was measured as mentioned in section 3.5. All the experiments were conducted in triplicate.

3.6.1 Effect of Temperature, Static Conditions and Shaking Speed on Bioflocculant Production

The *B. salmalaya* 139SI strain was inoculated into seed media and incubated at 37° C on a shaker at 150 rpm for 24h. From fresh culture of 2% v/v, was inoculated into several sets of 200 ml bottles containing 50 ml of production medium, then incubated at different cultivation temperatures were investigated, i.e. 25, 30, 35, 40 and 45°C on shaking 150 rpm for 144 h. Also, the shaking speeds were investigated for different speeds such as, 100, 120, 140, 160,180, 200 and 220 rpm, respectively. While, for the static condition of the inoculated production medium bottles were kept in static condition (without aeration). The cell free supernatant was obtained by centrifuge at 4000 rpm for 30 min to separate the cells. Then the flocculating activity was checked as mentioned in section 3.5, using Equation 3.1.

3.6.2 Effect of pH

The effect of pH on the production medium was determined at different pH value ranging from 3-11 adjusted by using 1 N HCl and 1 N NaOH solutions as needed A fresh culture of 2% (v/v) *B. salmalaya* 139SI strain selected was inoculated into the prepared medium, incubated for 144h at 35.5°C, and shaken 160 rpm. The flocculating activity was examined using kaolin clay to check the optimal pH required for the bioflocculant

production as indicated above (Aljuboori *et al.*, 2013). The flocculating activity was checked, using Equation 3.1.

3.6.3 Effect of Inoculum Size

The influence of the inoculum volume for bioflocculant production by *B. salmalaya* 139SI strain was examined because different inoculum sizes exert certain effects on the bioflocculant production and cell mass growth. The inoculum sizes used were 0.1%, 0.5%, 1%, 2%, 5%, 10% and 15% (Aljuboori *et al.*, 2013), whereas, the cultivation condition obtained were applied and the samples kept at shaken 160 rpm, 35.5°C and pH 7 for 144 h. The flocculating activities were checked, using Equation 3.1.

3.6.4 Effect of Cultivation Time on Bioflocculant Production

The cultivation condition was displays as pH 7, temperature 35.5°C, shaking speed 160 rpm and inoculum size 5% v/v, for 144 h were applied. The flocculating activities were calculated, using Equation 3.1.

3.6.5 Effect of Carbon and Nitrogen Sources on Bioflocculant Production

Bioflocculant production by microorganisms is significantly influenced by the carbon and nitrogen sources (Xia *et al.*, 2008). These parameters were assessed on bioflocculant production according to the protocol of Lachhwani (2005). The production media were prepared in separate flasks. Fresh bacterial suspension of 5% (v/v) was inoculated into the prepared medium. The media were supplemented with 10 g/L of different carbon sources such as glucose, sucrose, maltose, lactose, starch and a mixture of glucose and sucrose. The cultivation condition was displays as pH 7, temperature 35.5°C, shaking speed 160 rpm and incubated for 144 h were applied. To determine the influence of nitrogen source on bioflocculant production 1.5 g/L of each nitrogen sources was integrated into the fermentation medium in separate bottles, and similar cultivation condition as above were applied. Flocculation activities were calculated according to Lachhwani (2005), using Equation 3.1.

3.6.6 Effect of Various Salts on Bioflocculant Production

To study the effect of different salts sources on bioflocculant production, were carried out by one at a time method where various metal sources such as calcium chloride ferric chloride, copper sulphate, potassium chloride, magnesium sulphate, aluminium sulphate, dipotassium phosphate, disodium phosphate and monopotassium phosphate were added to the fermentation medium at final concentration of 0.25g/L, cultivation condition was displays as pH 7, temperature 35.5°C, shaking speed 160 rpm and incubated for 144 h were applied. Flocculating activity was determined according to Lachhwani (2005), using Equation 3.1.

3.6.7 Time Course Assay for Bioflocculant Production by *B. salmalaya* 139SI

Seed culture was prepared by inoculation 5% (v/v) bacterial suspension in 50 mL of enriched medium followed by overnight incubation at 35.5°C and 160 rpm. For optical density (OD600) test, sterile saline water was used to dilute the fermented broth to 0.1% (Cosa *et al.* 2012). In one litre of production medium, the optimized bacterial strain suspension was inoculated and incubated at 35.5 °C under shaking at 160 rpm for 144h. A 10 mL aliquot of the sample were taken periodically at timed intervals of 24 h, and 5 mL of the fermented broth was centrifuged. The obtained supernatant was used for determination of bioflocculant activity, according to Kurane and Nohata. (1994). The rate of bacterial growth was monitored by bacterial count using the standard plate method and the OD₆₀₀. In addition, the pH and flocculating activity were determined during the study.

3.6.8 Extraction and Purification of Bioflocculant

At the end of fermentation period, the culture was subjected centrifugation for 15 min at 3500 rpm in order to separate pelleted bacterial cells. The extracted supernatant was mixed with one volume (v/v) of sterile distilled water, followed by centrifugation for 15 min at 3500 rpm to remove indissoluble materials. Furthermore, the supernatant was mixed with two volumes of cold ethanol (1:2). The sample was thoroughly mixed with a stirrer and allowed to stand at 4°C for 12 h. Subsequently, the precipitate was extracted. The obtained crude polymer was dissolved in sterile distilled water. The solution sample was then mixed with chloroform and n-butyl-alcohol in proportion (5:2 ratios, v/v) with stirring and kept to stand at room temperature overnight. The upper surface portion was separated and subjected to centrifugation at 3500 rpm for 15 min to obtain a pure bioflocculant (Gao *et al.*, 2006). The purified supernatant was concentrated at 40 °C. To recover the precipitate, two volumes of ethanol water to obtain a pure bioflocculant (2.72 g/L) (Cosa *et al.*, 2011), and coded as bioflocculant QZ-7.

3.7 Analysis and Characterisation for Purified Bioflocculant QZ-7

3.7.1 Phenol-Sulfuric Acid Method for Total Carbohydrates

The phenol-sulfuric acid is a colorimetric method used to determine the total carbohydrate concentration in a bioflocculant or in any other biochemical compounds. It detects all the types of carbohydrates such as monosaccharides, disaccharides, oligosaccharides and polysaccharides.

To measure the carbohydrate concentration a calibration curve of glucose was first prepared at six different concentrations, and the absorbance was measured using UV-vis spectrophotometer at 490 nm wavelength. 0.05 mL of phenol (80%) was added into glucose standard and sample solution tubes and mixed well by a vortex. Then 5 ml of H_2SO_4 were added rapidly into each tube in order to get a good mixing and mixed again by vortex. All the tested tubes kept standing for 10 min and then placed in a 25°C water for another 10 min before the absorbance reading (Chaplin & Kennedy, 1994). The concentration of total sugar in bioflocculant samples was calculated by the standard curve equation.

3.7.2 Bradford Method for Total protein

The Bradford is a colorimetric assay for total protein determination that implicates the binding of Coomassie Brilliant Blue to protein. This method is faster, involves less mixing step, heating is not required, and more stable colorimetric response than other protein assays.

Different concentrations of Bovine Serum Albumin (BSA) were prepared as the standard solution in order to get the protein standard curve. The unknowns' protein concentration samples were diluted to obtain between $10 \ \mu g - 100 \ \mu g$ of protein. 5 mL of dye reagent was added into standard and sample tubes, then mixed well and allowed to stand for 5 min to complete the reaction. As a final point, the absorbance of the standard and sample were measured by spectrophotometer at 595 nm and the concentration of total protein in bioflocculant sample was calculated by the standard curve equation (Bradford, 1976).

3.7.3 Carbazole Assay for Uronic Acid

Uronic acid occurs in polysaccharides of Bacteria, fungi and plants cells. Carbazole assay is the fast method used for detecting and quantifying free and polymeric uronic acids according to Chaplin and Kennedy (1994).

Chemical reagents were prepared for the test including:

- 0.95g of sodium tetraborate decahydrate dissolved in 2 mL of hot water and 98mL of ice-cold concentrated sulphuric acid was added carefully with stirring. Then the reagent kept in refrigerator
- 2. 125 mg of carbazole dissolved in 100 mL of absolutes ethanol.

Initially, different concentrations of D (+) Glucuronic acid were prepared by using distilled water as a standard solution. All the standard, samples and control (125 μ L) cooled in ice bath, and 1.5 mL of an ice-cold reagent A was added with mixing in the ice bath. Then the mixture was heated at 100°C for 10 min, followed by rapid cooling in the ice bath. 50 μ L o reagent B was added and mixed well and then the mixture was reheated at 100°C for 15 min. Finally, the mixture was rapidly cooled to room temperature and the absorbance at 525 nm was obtained (Chaplin & Kennedy, 1994). The concentration of uronic acid in bioflocculant sample was calculated by the standard curve equation.

3.7.4 Fourier-Transform Infrared Spectroscopy (FTIR)

The purified bioflocculant QZ-7 was further subjected to Fourier-transform infrared (FTIR) spectroscopy (Perkin Elmer spectrum 400. USA). The purified bioflocculant QZ-7 was blended with potassium bromide (KBr) powder and compressed into disc to obtain translucent pellets for FTIR analysis. The background reference compound used was the pelleted form of potassium bromide. Infrared absorption spectra were recorded with PerkinElmer spectrum 400. The spectral resolution and wave number accuracy were 4000 -400 cm^{-1} under ambient conditions (Rasulov *et al.*, 2016c; Ugbenyen & Okoh, 2013)

3.7.5 Nuclear Magnetic Resonance (NMR)

The proton NMR spectrum was used to inspect the manifestation of chemical composition in the bioflocculant. The column purified bioflocculant was prepared by

dissolving 5 mg into 1mL deuterium oxide (D_2O), and kept shaking it overnight at 50 x g. Then the solution was transferred into the NMR tube, capped and subjected to ¹H-MNR spectra were plotted in 0-20 ppm at (500 MHz) or (400MHz) Bruker Advance-2 t (NMR BCX 400MHz, Jeol. Japan)

3.7.6 Determination Molecular Weight of Purified Bioflocculant QZ-7

The molecular weight (MWs) of QZ-7 was determined by high- performance gel permeation chromatography (HPGPC) coupled to refractive index (RI detector, Shimadzu, Japan), with a TSK G4000PWx1 column operated at 40°C. The column was calibrated by using dextran standards. The mobile phase was deionized- distilled (DDI) water at flow rate of 0.5 mL/min. Before injection, the sample was filtrated through 0.45µm filter (Chen *et al.*, 2017). The following regression equation was obtained

$$Log (mass) = K_1 T + K_2$$
(3.2)

Where mass (Da) and T (min) are the molecular mass and retention time of the samples, respectively, and K_1 and K_2 are constants.

3.7.7 Liquid Chromatography Mass Spectrometry (LC/MS)

To confirm the chemical composition of purified bioflocculant QZ-7 was determined by LC-MS analysis (Agilent Technology. Germany). For this, 10 mg of purified QZ-7 was completely dissolved in HPLC grade water and employed for LC-MS analysis (Nwodo *et al.*, 2014).

3.7.8 Scanning Electron Microscopy (SEM)

The SEM remarks was examined. Small amount of purified bioflocculant QZ-7 powder was spread and fixed on the iron stub. The fixed sample was scanned at 2kV using scanning electron microscopy (SEM, HITACHI- SU8220. Japan) (Rasulov *et al.*, 2017)

3.7.9 Energy Dispersion X-ray (EDX) Spectroscopy

The quantitative elemental analysis of the bioflocculant QZ-7 were conducted by fixing 5 mg of QZ-7 on copper stuff by an X-ray detector of SEM-EDX (HITACHI-SU8220. Japan)(Rasulove *et al.* 2017), and was examined with Aztec: 22650005761133539 software. Which revealed the weight and atomic (%) of different elements present in the sample.

3.7.10 Thermogravimetric Analysis (TGA)

About 10 mg of purified QZ-7 was examined using a TGA analyzer (Perkin Elmer TGA 4000. USA) at temperature of 50-900°C and at a heating degree of 10°C/min in a constant nitrogen gas flow rate of 20 mL/min.

3.8 Factors Affecting the Flocculation Performance of Purified Bioflocculant QZ-7

The followings will describe the material and methods of the assays of the parameters tested for the determination of the effect on bioflocculant QZ-7 performance, such as QZ-7 concentration, pH, temperature, cation ions, salinity and initial kaolin on the flocculating efficiency of QZ-7 were investigated (Gomaa *et al.*, 2012).

3.8.1 Effect of Bioflocculant QZ-7 Concentration on the Flocculation Efficiency

Optimal concentration of QZ-7 was investigated. Jar-test was used for the amplification of kaolin clay suspension (4 g/L) at neutral pH 7. Different concentrations of pure bioflocculant were used ranging from 0.25 to 10 mg/mL (1.0 mg/L.) The samples of kaolin clay suspension mixture containing 100 mL (4 g/L, pH 7), 0.25 to 10 mg/mL of bioflocculant concentration and 3 mL of 1% CaCl₂ were stirred vigorously and kept standing for 2-5 min. Calcium ions were added to neutralize the negative repulsion

changes to enhance and facilitating flocculation process. Optical density (OD) of the clarified solution was measured at 550 nm. Experimental control was also organised, and bioflocculant QZ-7 solution was switched with distilled water. Flocculation activity was estimated according to the following Equation 3.3 (Gao *et al.*, 2009).

Flocculation activity (%) =
$$\frac{F_1 - F_2}{F_1} \times 100$$
 (3.3)

Where F_1 and F_2 are the control of the sample and optical densities (OD) at 550 nm, respectively.

3.8.2 Effect of pH on the Flocculating Efficiency of Purified Bioflocculant QZ-7

To evaluate the effect of pH on flocculating activities of QZ-7, pH of the kaolin suspension was adjusted to pH 2, 3, 4, 5, 6, 7, 9 and 11 with 1M HCl or 1M NaOH, the samples were kept standing at 4°C for 1 day (He *et al.*, 2004). Bioflocculant QZ-7 a 2 mg/mL was dissolved in 10 mL of deionized water. The samples were mixed together and stirred for 2 min at 220 rpm, then another round of agitated for 10 min at 100 rpm and kept standing for 5 min. The purpose of using a wide range of pH is to determine the condition that enhances the flocculation process to occur with the aid of the bioflocculant and to obtain an optimal range as where it might perform. The measurement of flocculation activities was determined according using Equation 3.3 (Gao *et al.*, 2009).

3.8.3 Effect of Temperature on the Flocculation Efficiency of Bioflocculant QZ-7

Purified QZ-7 2 mg/mL was dissolved in 10 mL of distilled water to determine the optimum temperature for bioflocculant QZ-7 performance. The sample were tested at temperature of 25, 30, 35, 40, 45, 55, 65, 75, 85 and 100°C for 60 min in a water bath (He *et al.*, 2004). The kaolin suspension (4g/L) was adjusted at pH 7, the mixed solution was

stirred 2 min at 220 rpm, and agitated again agitated at 100 rpm for 10 min, and kept to stand for 5 min. The flocculation efficiency was determined using above Equation 3.3.

3.8.4 Effect of Cations on the Flocculating Efficiency of Bioflocculant QZ-7

The influence of cations on the flocculation activity of the QZ-7 was studied. Calcium chloride (CaCl₂) solution formerly used as activated substance was replaced by different metal salts solution of 1% (v/v), of monovalent such as KCl, NaCl, LiCl, divalent as MnCl₂, MgCl₂, and trivalent AlCl₃, FeCl₃ (Manivasagan *et al.*, 2015). Flocculating efficiency were measured using Equation 3.3.

3.9 Applications of Bioflocculant QZ-7

The surface water and different industrial wastewaters such as sugars mill, palm oil, rubber effluent and hospital wastewater were selected for the experiments. The initial chemical oxygen demand (COD), total suspended solid (TSS) and turbidity for surface water (river water). Also, COD and biological oxygen demand (BOD) for the filtered raw wastewater were measured according to standard method for water and wastewater examination as shown in Table 3.2, (APHA. 2005).

Parameters	Method reference	Wave-length	Range	Unit
pН	APHA 4500	-	-	-
COD	APHA 5220 C	425nm	0-1500	mg/L
BOD	APHA 2210B			mg/L
NH3-N	APHA 4500	425nm	0-2.5	mg/L
Colour	APHA2120 B	455nm	0-500	*Pt-Co
Conductivity, EC	APHA2510 B			μS/cm
TSS			0-750	mg/L
TDS	APHA 2540 D			mg/L

Table 3.2: Analytical Methods for Major Parameters using Standard Methods.

*Platinum-Cobalt

3.9.1 Surface Water Treatment

Water samples were collected from Langat River, kajang, Selangor, Malaysia. A sets of 100 mL of raw water were prepared to examine different concentration of crude and pure bioflocculant, for example, 0.5, 1, 2, 4, 6, 8 and 10 mg/L, respectively. The Jar test experiment was carried out in accordance with description of Wang *et al.* (2010). Each water sample was mixed with 1% (v/v) CaCl₂ and with different dose of bioflocculant, and the mixture were vigorously agitated at 200 rpm for 2 min and slowly stirred at 50 rpm for 5 min. The samples were kept standing for 25 min, and then the supernatant was extracted for analysis. The measurement of turbidity, COD and TSS were determined according to the standard methods for examination of water and wastewater (APHA. 2005)

Removal Efficiency (%) =
$$\frac{\text{Co} - \text{C}}{\text{Co}} \times 100$$
 (3.4)

Where C_0 is the initial value and C is the value after the flocculation treatment.

3.9.2 Removal of COD from Wastewater with Bioflocculant QZ-7

Approximately 0.2 g of the prepared bioflocculant QZ-7 was added to 100 mL of filtered wastewater. The physio-chemical characteristics of the raw wastewater (Appendix D, Table 4). The system was agitated with a magnetic starrier at room temperature for 2 min at 200 rpm and for 10 min at 50 rpm. The samples were left to stand for 15 min and clarified through a 0.45 µm membrane filter paper. The final COD was determined by the closed reflux colorimetric method (APHA. 2005). The COD removal rate was calculated as follows:

Removal rate =
$$(CODi - CODf)/CODi \times 100$$
 (3.5)

Where R is the removal rate in %, CODi is the COD concentration of the wastewater before treatment, and CODf is the COD concentration of the wastewater after treatment.

3.9.3 Removal of BOD from Wastewater with *B. salmalaya* strain 139SI

The removal of BOD from wastewater with *B. salmalaya* strain 139SI was carried out using two diluted samples. The initial dissolved oxygen (DOi) of the first sample was measured using a DO meter. The other sample was inoculated with 10% of bacterial suspension and incubated in a BOD incubator for 5 days. The final dissolved oxygen (DOf) was measured after 5 days. The BOD value was calculated using Equation 3.6 (APHA. 2005), and the BOD removal percentage was calculated using the following equation.

$$BOD (mg/L) = DOi - DOf$$
(3.6)

3.9.4 Heavy Metal Adsorption from Aqueous Solution

Removal of heavy metals using bioflocculant was measured according to the work Lin and Harichund (2011). The metal salts used comprised copper sulphate, lead acetate, sodium arsenate, zinc sulphate, cadmium chloride (Sigma Co). The effect of different initial concentrations of heavy metals (20, 40, 60, 80, and 100 mg/L), bioflocculant concentrations (20, 40, 60, 80 and 100 mg/L) and pH value (3, 4, 5, 6, 7, 9 and 11) on the metal adsorption were examined. Bioflocculant QZ-7 solutions (5mL) were added through a dialysis tube in flasks having accurate 100 mL of each metal-salts solution and kept for 2 h at room temperature under shaken at 100 rpm. Next, 2 mL of each the solution was filtered through an Amicon filter (Centrifree) and then acidified within 1% of nitric acid solution for residual metal determination. The metal amounts removed from the samples tested, that is the bounded polymers were measured previously, whereas, those that remained after 2 h were detected using Inductively Coupled Plasma Mass Spectrometry (AAS; Model AA 6300, Shimadzu, Japan); removal percentage rate for each element was calculated (Gourdon *et al.*, 1990). Controls, 5 mL de-ionized water were prepared in the dialysis tube per different metal-salt solutions. Removal ratio (R) is expressed Equation 3.7.

Removal ratio =
$$\frac{\text{Ci} - \text{Ce}}{\text{Ci}} \times 100$$
 (3.7)

Where Ci and Ce are original and equilibrium metal concentrations removed, respectively.

3.9.5 Removal of Heavy Metals from Wastewater using Bioflocculant QZ-7

Wastewater from rubber industry was selected for the experiments. Removal efficiency analysis was performed as stated by the standard methods for examination of water and wastewater (APHA, 2005). Filtered wastewater samples of 100 mL were poured into a dialysis tube, and different bioflocculant concentration 20, 40 and 60 mg/L, were added and stirred with a magnetic stirrer for 15 min at 100 rpm, slowly stirred for 5 min at 50 rpm and kept standing to settle for 15 min. The volume of supernatant was filtered through a filter paper and used for heavy metals analysis before and after bioflocculant treatment through inductively coupled plasma mass spectrometry (AAS; Model AA 6300, Shimadzu, Japan); metal removed percentage rate of each element was also calculated (Gourdon *et al.*,1990). Controls, 5mL deionized water were prepared in the dialysis tube per different metal-salt solutions. Removal ratio (R) is expressed in Equation 3.7.

3.9.6 Removal of Pharmaceuticals from Wastewater using Bioflocculant QZ-7

3.9.6.1 Preparation of Standard Solutions

The individual stock standard solution of pharmaceuticals (1000 μ g/mL) was prepared by dissolving an appropriate amount of each analytical standard for pharmaceuticals in methanol. Each stock solution was then stored in a clean 15-mL amber glass vial with fluoropolymer-lined cap in a freezer at -15°C. A mixed standard solution (1 μ g/mL) was prepared from the stock solutions with a concentration of 1000 μ g/mL by mixing 10 μ L of each stock solution in a 10-mL volumetric flask with methanol. A series of calibration standard solutions was prepared by an appropriate dilution of the mixed standard solution in (2:1) water and methanol at pH 10. The water and methanol mixture was adjusted to pH 10 using ammonium hydroxide (2.0 M) and formic acid (2.0 M).

3.9.6.2 Sample Collection and Preparation

Influent and effluent wastewater was collected from a wastewater treatment plant at Sungi Boloh Hospital Malaysia using a homemade glass sampler. Then, the samples were kept in clean 1Liter amber glass bottles. The bottles were stored in an icebox at 0°C while the samples were transported to the laboratory. Then the samples were μ filtered using 1.2 μ m glass fiber filter (Whatman, Maidstone, UK). The filtered samples were preserved by addition of 1g/L of sodium azide to prevent microbial degradation (APHA, 2005).

3.9.6.3 Solid-Phase Extraction (SPE)

Effluent and influent wastewater were collected from hospital water treatment plant around Kuala Lumpur. After water sampling, 1 g of sodium azide and 50 mg of ascorbic acid were added to the water samples to preserve the samples and prevent microbial degradation. All water samples were filtered through Whatman GF 6 filter, then the pH of the samples was adjusted to 7.0 with 2.0 M NaOH and 2.0 M HCl solutions. HLB SPE
Tube from Supelco (3 cc, 60 mg) (Bellefonte, USA) was used for the solid phase extraction (SPE). The extraction volumes were 100, 250 mL for sewage influent and effluent, respectively. The solid phase adsorbent was pre-conditioned with 5 mL of methanol and 3 mL of non-contaminated tap water (pH adjusted to 7.0). The samples were introduced to the cartridge using a manifold which holds multiple cartridges and combined with the large volume sampler. The sample flow rates in the system were 2, 5, and mL min⁻¹ for sewage influent and sewage effluent water, respectively. After sample loading, the solid phase was washed with 2 mL of 5% methanol in ultra-pure water at pH 10. The cartridges were then dried for about 30 min and subsequently, the pharmaceuticals were eluted with 1×1 mL of METH and 2×1.5 mL of ACN-METH with 2% ammonium hydroxide. The extracts were evaporated to near dryness under the vacuum at 50 °C and then 0.5 mL of (2:1) water and methanol at pH 10 were added. The extracts were stored at -18°C until analysis (Kafeenah *et al.*, 2018).

3.9.6.4 UPLC-ESI-MS/MS Analysis

The LC analysis was performed using an UPLC Agilent 1290 Infinity system (Agilent Technologies, Germany) consisting of a binary pump, vacuum degasser, an autosampler, and a thermostatic column oven. The chromatographic separation was achieved with Accucore Polar Premium LC column (100 mm \times 2.1 mm, particle size 2.6 µm, supplied by Thermo Scientific (Loughborough, UK). The injection volume was set at 1 µL with the column temperature 30 °C. The column was eluted for the new method with methanol (eluent B) and NH4Ac 5 mM/HAc in HPLC water (eluent A) at pH 4.6 at a flow rate of 0.1 - 0.2 mL/min. Where ammonium format and formic acid (NH4Fc/HFc) in HPLC water at pH 3.2 were used as (eluent A) in the positive ionisation mode method. Ammonium carbonate at pH 7 was used with the negative mode ionisation method (Kafeenah *et al.*, 2018).

3.9.6.5 Mass Spectrometry

For detection and quantification, the Agilent Technologies 6490 triple-quadrupole mass spectrometer (Agilent Technologies, Singapore) was connected to UPLC via an electrospray ionisation (ESI) interface. It is equipped with the Agilent Jet Stream system AJS ESI as an ionisation source. The capillary voltage was 3 kV while the nebuliser pressure was 45 psi for both modes. Nitrogen gas was used as desolvation and nebulising gas at a flow rate of 14 L/min and temperature 200°C. In addition, the mass hunter software was used for instrument control, peak detection, and integration. To increase sensitivity, selectivity, and data acquisition, a multiple reaction monitoring modes (MRM) were used both in the negative and positive mode simultaneously with a dwell time of 0.2 s (Kafeenah *et al.*, 2018).

3.10 Statistical Analysis

Statistical analyses for the data collected were analysed using SPSS statistical version 25 software. A minimum of the three replicates of treatment were made to detect variability and to avoid biases. Exploratory data analyses (EDA) with descriptive statistical as mean and standard deviation were determined for pattern trend and observation. Significant differences were analysed through analysis of variance ANOVA for the basis of making conclusion and predication (Zhu *et al.*, 2012). One-way ANOVA, A multiple comparison post-hoc tests was used for the determination of differences in each assay for cases of equality of variance assumed or not assumed, respectively. Posthoc test results were used as the basis for determination of the highest flocculating activity achieved in the respective assays. Statistical significance was defined as a *p*-value < 0.05 mean value are shown for 3 individuals' \pm SD.

3.11 Synthesis of Silver Nanoparticles (AgNPs) by using Bioflocculant

The synthesis of AgNPs using bioflocculant QZ-7, its characterisation and application was carried out in nutrient broth medium, using cell-free supernatant and purified bioflocculant.

3.11.1 In nutrient broth medium

The nutrient broth was supplemented with 1.0% of glucose since the bioflocculant producing bacteria need glucose as carbon source and 3 mM of AgNO₃, 2.0% of 24 h fresh culture broth of *B. salmalaya* 139SI strain was inoculated to the medium and the mixture was incubated in darkness at 35.5°C at 140 rpm for 7 days (Zaki *et al.*, 2014).The extracellular synthesis of AgNPs was observed by visual check-up of the change in nedium colour from a clear yellow to brown. The control was prepared without addition of AgNO₃.

3.11.2 Using Cell-Free Supernatant

The basal bioflocculant producing medium containing (g/L): glucose, 20.0; yeast extract, 0.5g; ammonium sulphate, 0.2g and urea, 0.5g; K₂HPO₄ , 5.0g; KH₂PO₄ , 2.0g and MgSO₄ , 0.2g, was inoculated with 2.0% of fresh culture broth of *B. salmalaya* 139SI, then incubated at 35.5°C for 24 h. The broth was centrifuged at 4.000 rpm for15 min; about 10% of the cell-free supernatant was added to 100 mL of 3 mM AgNO₃ solution. The bottle was incubated in darkness on a shaker 120 rpm at 37°C (Elkady *et al.*, 2011). The spectra were taken at every 24 h interval between 200 to 600 nm using UV-vis spectrophotometer. Subsequently, the peaks are achieved, the AgNPs having liquid was kept in oven at 60°C for overnight until dried.

3.11.3 Using Purified Bioflocculant QZ-7

Prepared concentration of 3mM AgNO₃ was added to100 mL of 10% bioflocculant QZ-7 solution in 200 mL bottles. The bottles were incubated in darkness at 37°C for 48 hrs at 140 rpm (Zaki *et al.*, 2014). The AgNPs synthesis was monitored as described above. The control was prepared without addition of AgNO₃.

3.12 Characterisation of Silver Nanoparticles (AgNPs)

3.12.1 UV-vis Spectrophotometry

The primary screening of AgNPs production was visual observation for a colour change from yellow to brownish, 5 mL of the sample was withdrawn and centrifuged at 5000 rpm for 10 min. The spectra of supernatant were conducted using a UV-Vis 1800 spectrophotometer in the range of 200-600 nm. Quartz cuvettes with optical path length of 10 mm were used in the measurement. 1mL of synthesized AgNPs was withdrawn and the absorbance was measured. The synthsized AgNPs was obtained by centrifugation at 10.000 rpm for 10 min. The pellet samples were dried at 60 °C in vacuum oven for overnight.

3.12.2 Fourier-Transform Infrared Spectroscopy (FT-IR)

The presence of functional groups in the synthesized AgNPs were determined using FT-IR spectra (Perkin Elmer spectrum 400. USA). The dried pellet of silver nanoparticles was subjected to FT-IR spectral characterisation in the frequency rang of 4000-400cm⁻¹ and at resolution ratio of 1cm⁻¹ (Rasulov *et al.*, 2016c; Rasulov *et al.*, 2017).

3.12.3 X-ray Diffractometry (XRD)

The dried powder of AgNPs was subjected to powder XRD using Empyrean. Cu-K radiation diffracted intensities were recorded from 10 and 80° in 2θ angles. XRD approves

the nature of powder material, which ever crystalline or amorphous. The AgNPs mean crystallite size was measured by the Debye-Scherre equation 3.8 (Monshi *et al.*, 2012)

$$\mathbf{D} = \mathbf{K}\lambda/\beta\,\cos\theta\tag{3.8}$$

K – Scherrer constant, related to crystalline shape, λ –Radiation wavelength, β – Full width at half maximum of the diffraction peak, and θ – Bragg's angle.

3.12.4 Field Emission Scanning Electron Microscopy (FE-SEM) and EDAX Analysis

The FE-SEM and EDAX were used to study the morphology, size, and elemental composition of the biosynthesized AgNPs. The surface imaging technique, detection of particles shape, size, surface morphology and size distribution of nanoparticles by using the instrument - FE-SEM (JEOL JEM-1230, Japan), and EDAX (INCApenta FETX3 OXFORD (Lawrence & Prakash, 2019). For this, AgNPs were suspended into water; a drop of it was placed on the copper grid and kept to dry at room temperature.

3.13 Antibacterial Activity

The antimicrobial activity was tested in both, solid and liquid media. As such liquid AgNPs were used in this study with slight modification in the method described by Wei *et al.* (2012). At present, 20 μ L of AgNPs was added to tubes containing 10 mL of Muller-Hilton broth inoculated with 10 μ L of inoculum of pure isolates of *Escherichia coli* ATCC35401, *Salmonella enteritidis* ATCCBAA-711, *Pseudomonas aeruginosa* and *Staphylococcus aureus* ATCC2592 incubated at 37 °C at 120 rpm. The absorbance was measured at 600 nm at every 3 h interval up to 12 h and then 24 h of incubation to check the growth. On solid medium, well diffusion method was employed. Muller-Hilton agar plates were streaked with the test organisms with top agar, wells were made after solidification and different concentrations of AgNPs such as 20 μ L, 40 μ L, 60 μ L and 80 μ L

were pipetted into wells (i.e. 35.2, 70.4, 105.6 and 140.8 μ g). The plates were incubated for 24 h at 37°C and the zone of inhibition was measured.

CHAPTER 4: RESULTS

4.1 Introduction

In the present study, the production of the bioflocculant produced by *B. salmalaya* 139SI was investigated to determine the optimal culture medium composition and environmental condition required. Various factors influencing the production of bioflocculant, such as initial pH of culture medium, inoculum size, culture temperature, carbon and nitrogen sources, metal salts, shaking speed, static condition and time course were investigated. Chemical composition and properties of the bioflocculant were determined. Also, the application of bioflocculant in surface water and industrial wastewater treatment were investigated. Furthermore, the bioflocculant was tested to synthesized nanoparticles.

4.2 Selection of Bacterial Strain for Bioflocculant Production

4.2.1 Morphological Identification

Figure 4.1a, shows the *Bacillus salmalaya* strain 139SI colonies were large and whiteopaque with a rough and irregular edge. The colonies were 2-3 mm in diameter and shows strong β -haemolytic activity after 18-24 h of incubation at 37°C on 5% v/v sheep blood agar as showed in Figure 4.1b.



Figure 4.1: a. Front View of Pure Colony of *B. salmalaya* 139SI Strain; b. Back View of β -haemolytic *B. salmalaya* 139SI Activity.

4.2.2 Microscopic Examination

Figure 4.2, shows the *Bacillus salmalaya* strain 139SI under the microscope a gram positive, short rods in chains and the formation of endo-spore.



Figure 4.2: Gram Stain of B. salmalaya 139SI.

4.3 Screening Strain for Bioflocculant Production

The bacterial strain has been screened for bioflocculant production. From 15 strains appendix A-1, 5 strains have the highest flocculating activity as presented in Table 4.1.

Strain code no	Flocculating activity (%)	Standard deviation
BS* 139SI-1	67.5	0.65
BS 139SI-5	54.2	0.77
BS 139SI-7	83.3	0.75
BS 139SI-8	72.2	1.95
BS 139SI-13	63.4	0.45

 Table 4.1: Flocculating Activity Values for the Selected Strains.

*BS means *Bacillus salmalaya* strain

4.4 Optimisation of Bioflocculant Production by *B. salmalaya* 139SI

4.4.1 Effect of pH on Bioflocculant Production

The inceptive pH of the fermentation media has a direct affected the bioflocculant production. Figure 4.3, and Table 4.2, shows the optimum pH value and the significant differences (p<0.05), between different pH value for the bioflocculant production.

Table 4.2: Analysis of Variances for the Effect of pH on Bioflocculant Production.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	2.011	7	0.287	1607.644	0.000
Within Groups	0.003	16	0.000		
Total	2.013	23			



Figure 4.3: Effect of pH on Bioflocculant Production.

4.4.2 Effect of Inoculum Size on Bioflocculant Production

The influence of inoculum volume of bioflocculant production was determined according to zhang *et al.* (2008), using bacterial inoculum size ranging from 0.1 to 15 % (v/v). Figure 4.4, showed the differences effect in inoculum volume at certain change on

the cell mass and flocculation activity values. While, table 4.3, prevailed that there is a significant difference (p<0.05), of different inoculum size.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	3042.443	6	507.074	547.177	0.000
Within Groups	12.974	14	0.927		
Total	3055.417	20			

Table 4.3: Analysis of Variances for the Inoculum Size Effect on Bioflocculant

 Production.



Figure 4.4: Inoculum Size Effect on Bioflocculant Production.

4.4.3 Effect of Temperature on Bioflocculant Production

The influence of temperature was used to investigate its effect on *B. salmalaya* 139SI for bioflocculant production. Figure 4.5, show that, the flocculating activity of bioflocculant was found to be 81.93%, when the culture temperature was at 35.5°C, which was a best flocculating activity in the experiment.





Table 4.4, shows there is significant differences (P < 0.05), for the effect of different temperature degrees on bioflocculant production.

Table 4.4: Analysis of Variances of Different Cultivation Temperature on Bioflocculant

 Production.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	5778.758	6	963.126	133.644	0.000
Within Groups	100.893	14	7.207		
Total	5879.651	20			

4.4.4 Effect of Shaking Speed on Bioflocculant Production

The effect of shaking speed on bioflocculant production is shown in Figure 4.6. The optimum shaking speed was found to be 160 rpm, and flocculating activity was 83.6



Figure 4.6: Effect of Shaking Speed on the Bioflocculant Production.

Table 4.5, showed that there is significant difference (p<0.05), between different shaking speed.

Table 4.5: Analysis of Variances for Effect of Shaking Speed on Bioflocculant

 Production.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1203.763	6	200.627	601.881	0.000
Within Groups	4.667	14	0.333		
Total	1208.430	20			

4.4.5 Effect of Static Condition on Bioflocculant Production

When flocculating activity of the cultures under non-shaking condition it was found that the strain of *B. salmalaya* 139SI exhibited much less flocculating activity as shown in Figure 4.7.



Figure 4.7: Effect of Static Condition on the Bioflocculant Production.

Table 4.6, presented that there is a significant difference (p<0.05), of the effect of static

condition.

Table 4.6: Analysis of Variances for the Effect of Static Condition on Bioflocculant

 Production.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1730.198	5	346.040	377.727	0.000
Within Groups	10.993	12	0.916		
Total	1741.191	17			

4.4.6 Effect of Carbon Source on Bioflocculant Production

The effect of carbon sources includes sucrose, glucose, lactose, maltose, fructose, and starch was evaluated for the production. The results of distinct carbon sources on the bioflocculant synthesis by *B. salmalaya* strain 139SI are shown Figure 4.8. The highest 90.1% and lowest 28.9% flocculating activities were obtained with a mixed carbohydrate (glucose and sucrose), source and starch with (1:1) mixing ratio, respectively.



Figure 4.8: Carbon Sources Effect on Bioflocculant Production.

As shown in Table 4.7, there is a significant difference (p<0.00), between different carbon source.

Table 4.7: Analysis of Variance for the Effect of Carbon Source on Bioflocculant

 Production.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	2.229	5	0.446	241.474	0.000
Within Groups	.022	12	0.002		
Total	2.251	17			

4.4.7 Effect of Nitrogen Sources on Bioflocculant Production

Figure 4.9, showed the highest flocculating activity which obtained after 72 h of cultivation with mixture nitrogen such as yeast extract and urea 72.2%, whereas, as a single nitrogen source such as urea 67.7%, yeast extract 64.53%, while the lowest activity was obtained with peptone 42.5%. However, the inorganic nitrogen shows 58.86% of flocculating activity.

Table 4.8, revealed that there is a significant difference (p<0.00), between different nitrogen source.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	0.647	5	0.129	11.955	0.000
Within Groups	0.130	12	0.011		
Total	0.777	17			

Table 4.8: Analysis of Variance for the Effect of Nitrogen Sources on Bioflocculant

 Production.



Figure 4.9: Effect of Nitrogen Sources on Bioflocculant Production.

4.4.8 Effect of Different Inorganic Salts

The effect of various salts on bioflocculant production were examined. Table 4.9. Show that the flocculating activity was nourished by several salt such as CaCl₂, KH₂P₄, K₂HPO₄, KCl, FeCl₃ and Na₂HPO₄, but were adversely affected by Al₂ (SO₄)₃ and CuSO₄ Table 4.10, shows there is significant differences (p< 0.00), of the different of inorganic salts.

Salts	Flocculation activity (%)	Standard deviation
CaCl ₂	76.8	1.7
KC1	73.0	1.6
FeCl ₃	55.6	1.4
CuSO ₄	40.2	1.2
$Al_2(SO_4)^3$	42.2	1.8
KH ₂ PO ₄	75.7	1.7
K ₂ HPO ₄	73.1	1.3
Na ₂ HPO ₄	63.1	1.7
MgSO ₄	67.0	1.5
0.1%K ₂ HPO ₄ +0.5% KH ₂ PO ₄	91.8	1.5

 Table 4.9: Effect of Inorganic Salts on the Bioflocculant Production.

Table 4.10: Analysis of Variance for the Effect of Inorganic Salts on the Bioflocculant

 Production.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	6957.467	9	773.052	280.430	0.000
Within Groups	55.133	20	2.757		
Total	7012.600	29	<u>~</u>		

4.5 Time Course of Bioflocculant Production by *B. salmalaya* 139SI

Optimal culture conditions were monitored for bioflocculant production by *B.* salmalaya 139SI. Figure 4.10, shows the time course assay of bioflocculant production. The production of bioflocculant at the early growth phase of *B. salmalaya* 139SI was reflected by the flocculating activity achieved through cultivation period. Results depicted in Table 4.11, prevailed that there is a significant difference (p<0.00), of the cultivation condition.



Figure 4.10: Time Course of Bioflocculant Production by Bacillus salmalaya139SI.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	11248.807	7	1606.972	6536.837	0.000
Within Groups	3.933	16	0.246		
Total	11252.740	23			

4.6 Characterisations of Bioflocculant QZ-7

The composition of bacterial bioflocculant plays role in their flocculating activities (Wu & Ye, 2007).

4.6.1 Qualitative Chemical Analysis of the Bioflocculant QZ-7

Based on qualitative analysis of bioflocculant QZ-7, depicted in Table 4.12, presenting a clear indication that the bioflocculant QZ-7 composed of polysaccharides, protein and uronic acid was found.

Reaction type	Analytical method	Occurrence
• -	Anthrone test	+
Polysaccharide	Phenol-sulphric acid test	+
•	Molish test	+
Ductoin	Bradford test	+
Protein	Folin-Lowry test	+
Uronic acid	ronic acid Carbazole assay	

Table 4.12: Qualitative Analysis of Bioflocculant QZ-7.

4.6.2 Quantitative Chemical Analysis of the Bioflocculant QZ-7

On the basis of the quantitative analysis of the produced bioflocculant QZ-7. Figure 4.10, and Figure 4.11 showed the standard curve of the total carbohydrate and protein, were determined to be 79.08% and 15.4%, respectively.



Figure 4.11: Determination of Carbohydrate Concentration.



Figure 4.12: Determination of Protein Concentration.

4.6.3 Fourier-Transform Infrared Spectroscopy (FTIR)

Infrared spectrophotometry was used to analyze the purified bioflocculant QZ-7 as illustrated in Figure 4.13. Clear absorption peaks were revealed at spectrum of wave number 3420.56, 2929.82, 2437.35, 2176.08, 2073.19, 1658.90, 1432.97, 1187.49, 1109.66, 924.94, 618.50, 535.79, and 476.32 cm⁻¹.



Figure 4.13: Fourier-Transform Infrared Spectroscopy Analysis of Bioflocculant QZ-7.

4.6.4 Molecular Weight Analysis

The HPGPC spectrum of the purified bioflocculant QZ-7 exhibited a symmetrical and sharp peak in the retention time of 6.53 and 9.95 (Figure 4.14). The molecular mass–retention time equation in accordance with the calibration curve was expressed as follows:

Log (molecular weight) = -0.1368T + 8.3496

The average weight of the bioflocculant was calculated to be 5.13×10^5 Da, which was much higher than the weight of other bioflocculants (Yang *et al.*, 2015).



Figure 4.14: Gel Permeation Chromatography of the Purified Bioflocculant QZ-7.

4.6.5 H-NMR Spectrum

The spectral analysis of bioflocculant QZ-7 shown in Figure 4.15, 4.16 revealed the presence of hydroxyl, methyl, amino and carbonyl groups.



Figure 4.15: H⁺ NMR Spectra of Purified Bioflocculant QZ-7.



Figure 4.15: C¹³ NMR Spectra Of Purified Bioflocculant.

4.6.6 Liquid Chromatography Mass Spectrometry (LC-MS)

Through the LC/MS based analysis, adducts of Na+, NH4+ and H+ ions were identified for the sugar monomers depending on fragmentation and elution time in present case. Figure 4.17, shows the presence of carbohydrate such as glycoprotein (A) at 741 m/z-745 m/z at retention time 3.34. While, Na+ adducts for glucose (B) appeared as (Glc+Na) + (182.96 m/z), glucuronic acid (C) at 212.85 m/z, at retention time 1.653, and rhamnose (D) as (Rha+Rha+Na) + (m/z 354.3) at retention time 1.576 and 1.529 min, respectively.









Figure 4.17: ESI Spectra of LC/MS Analyses of Bioflocculant QZ-7 Showing Glycoprotein (A), Na⁺, H⁺ and NH4⁺ adducts of Glucose (B), Glucuronic Acid (C) and Rhamnose (D).

4.6.7 Scanning Electron Microscopy (SEM) Imaging

The morphological surface structure of QZ-7 were illuminated prior to and after flocculation process with kaolin clay particles. As shown in Figure 4.18A, the QZ-7 has

a greyish colour with a condensed crystalline brick shaped structure. Whereas, Figure 4.18B, showed the flocculation of the bioflocculant QZ-7 with kaolin clay.



Figure 4.16: SEM Micrograph of Bioflocculant QZ-7 of *B. salmalaya strain* 139SI. A. Purified Bioflocculant QZ-7. B. Bioflocculant Aggregation with Kaolin Clay.

4.6.8 Energy Dispersive X-ray (SEM- EDX)

The elemental analysis of bioflocculant QZ-7 on the basis of SEM-EDX testing as shown in Figure 4.19, the existence of C, O, N, P and S in this macromolecule as 55.74%, 42.74%, 0.54%, 0.93% and 0.06%, respectively.



Figure 4.19: EDX Analysis of Bioflocculant QZ-7.

4.6.9 Thermogravimetric (TGA) Determination of the Bioflocculant QZ-7

Bioflocculant QZ-7 was tested to explain its property comportments while exposed to a different heat degree. For this reason, to assist us to recognise its pyrolysis property after exposed to a very elevated temperature. Results as depicted in Figure 4.20, approximately 12.14%, 23.76% in weight loss of QZ-7 at 100°C, 200°C, respectively, and a 45.28% weight loss of QZ-7 at 200-450°C.



Figure 4.20: Thermogravimetric Analysis of the Purified Bioflocculant QZ-7.

4.7 Factors Affecting of Bioflocculant QZ-7 Performance

4.7.1 Effect of pH on the Flocculation Efficiency of QZ-7

As shown in Figure 4.21, bioflocculant QZ-7 was found to be a fairly steady at wider range of pH range (5-7), and above 70 % flocculating activity was accomplished at that pH series.



Figure 4.17: pH Stability of the Purified Bioflocculant QZ-7.

As depicted in Table 4.13, there is significant difference (p<0.000), of different pH value.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	16405.203	7	2343.600	2014.895	0.000
Within Groups	18.610	16	1.163		
Total	16423.813	23			

Table 4.13: Analysis of Variance for the Effect of pH on Flocculating Performance.

4.7.2 Effect of Temperature on the Flocculation Efficiency of QZ-7

The influence of temperature on the flocculation activity of the purified bioflocculant QZ-7 was investigated. Figure 4.22, showed that more than 80% of flocculating efficiency was sustained in the temperature series of 25- 45°C, also the flocculating efficiency above 70% was found to be a constant at 55-65 °C. Moreover, above 60% of flocculating efficiency was obtained at 75-85 °C.



Figure 4.18: Thermal Stability of the Purified Bioflocculant QZ-7.

As shown in Table 4.14, the treatment analysis for the effect of different temperature degrees on flocculating activity revealed that there are significant differences (p<0.00).

 Table 4.14: Analysis of Variance for the Effect of Temperature on Bioflocculant Performance.

	Sum of	Degrees of	Mean	F-ratio	P-value
	Squares	Freedom	Square		i value
Between Groups	12932.466	8	1616.558	983.215	.000
Within Groups	29.595	18	1.644		
Total	12962.061	26			

4.7.3 Effect of Bioflocculant QZ-7 Concentration on the Flocculation Efficiency

Figure 4.23 showed the flocculating activity increased dramatically as the bioflocculant QZ-7 concentration increased from 0.25 to 2 mg/mL. The highest flocculating efficiency of 93.6% was obtained at 2 mg/mL of bioflocculant QZ-7. Also, QZ-7 show a good flocculating activity of 77.23% at 1mg/L, 73.2% at 0.5mg/mL, 84.2% at 4 mg/mL and over 64.93% at 6 mg/mL.



Figure 4.19: Bioflocculant QZ-7 Dosage Effect on Flocculating Activity.

Table 4.15, shows the result obtained is in accordance with the reported findings in the flocculation performance of different concentration of bioflocculant has significant differences (p<0.00).

Table 4.15: Analysis of Variance for the Effect of Bioflocculant QZ-7 Concentration.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	12241.856	8	1530.232	834.503	0.000
Within Groups	33.007	18	1.834		
Total	12274.863	26			

4.7.4 Effect of Cations on the Flocculation Efficiency of QZ-7

Figure 4.24 shown the monovalent and divalent cations, including Al^{+3} were facilitating the flocculating activity to a good point matched with the trivalent Fe⁺³ cations. The maximum flocculating activity 92.9 % was detected with Ca⁺², followed by Al^{+3} (83.3%), Mn^{+2} (75.6%), K^{+} (71.6%), Mg^{+2} (71.4%), Na^{+} (67.7%) and Li⁺ (64.0 %).



Figure 4.20: Effect of Metal Ions on Flocculating Activity of Purified Bioflocculant QZ-7.

The statistical analysis presented in Table 4.16, indicated that there is a significant difference (p<0.000) between different cations.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	11327.947	7	1618.278	1016.387	0.000
Within Groups	25.475	16	1.592		
Total	11353.422	23			

Table 4.16: Analysis of Variance for the Effect of Cations on Flocculating Activity.

4.7.5 Effect of Salinity on Flocculation Efficiency of QZ-7

Figure 4.25. Different concentrations of NaCl salt were tested to examine the effect of salinity on bioflocculant QZ-7 flocculation performance. As presented at 1% and 5% concentration of NaCl the bioflocculant QZ-7 maintained over 90 % of flocculating efficiency. While 10% (w/w) of NaCl in kaolin suspension has no effect on the flocculating efficiency of bioflocculant QZ-7.





The statistical analysis of variance of the effect salinity concentration of flocculating activity is depicted in Table 4.17, revealed that there is significant difference between different concentration of salt 1-30% (w/w) found to be (p<0.000).

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	18648.423	6	3108.071	1049.224	0.000
Within Groups	41.472	14	2.962		
Total	18689.895	20			

Table 4.17: Analysis of Variance for the Effect of Salinity on Flocculation Performance.

4.8 Application of Bioflocculant QZ-7

In this study the application of bioflocculant QZ-7 in water treatment and industrial wastewater were investigated by using surface water, sugar mill, rubber wastewater and hospital wastewater. Bioflocculant QZ-7 produced by *Bacillus salmalaya* 139SI shown different levels of COD, TSS, TDS, Turbidity and BOD removals, heavy metal adsorption. Differences in the attraction of metals for bioflocculants are due to attractive

interaction, charge density and polymers conformation natures with adsorbed ions (Morillo *et al.*, 2006). The metal biosorption mechanism and kinetics influenced by experimental conditions mainly, pH medium, bioflocculant concentrations and initial heavy metal concentrations (Converti *et al.*, 2006).

4.8.1 Surface Water Treatment by Bioflocculant QZ-7

The physical and chemical characteristic of Langat River are shown at Table 4.18.

Table 4.18: The Average Water Quality Characteristic of Langat River.

Parameter	Values
pH at 25°C	6.72
Conductivity (ms/cm)	25.5
Chemical Oxygen Demand (mg/L)	67.4
Total suspended solid (mg/L)	194.6
Turbidity NTU	178.5

4.8.1.1 Removal of Turbidity

The flocculation performance of purified QZ-7 was shown in Figure 4.26. The turbidity removal of purified QZ-7 was higher than crude bioflocculant especially at low bioflocculant concentration. Purified QZ-7 concentration of 5 mg/L, reduced the turbidity of river water from 178.5 NTU to 5.2 NTU and the turbidity was 96.8%. The crude bioflocculant at 5 mg/L dosage produced water with final turbidity of 35.6 NTU with removal efficiency 83.3%.





As depicted in Table 4.19 and Table 4.20 there is a significant difference (p<0.000),

between different concentration of pure QZ-7.

Table 4.19: Analysis of Variance for the Effect of Pure Bioflocculant QZ-7 onTurbidity Removal.

	Sum of	Degrees of	Mean Square	F ratio	P voluo
	Squares	Freedom		1'-1 atio	I -value
Between Groups	2813.491	5	562.698	677.950	0.000
Within Groups	9.960	12	.830		
Total	2823.451	17			

Table 4.20: Analysis of Variance for the Effect of Crude Bioflocculant QZ-7 on Turbidity Removal.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1733.765	5	346.753	209.730	0.000
Within Groups	19.840	12	1.653		
Total	1753.605	17			

4.8.1.2 Total Suspended Solid Treatment

The following Figure 4.27, show the final TSS concentration removal of water treated by purified and crude bioflocculant QZ-7. In general, the TSS removal efficiency was gradually increased as the bioflocculant concentration increased. The TSS removal efficiency of purified bioflocculant QZ-7 was found a significant at concentration range of 0.5 to 7 mg/L. In addition, 5 mg/L of purified bioflocculant produced water with final TSS of 4.53 mg/L. (97.8%).



Figure 4.23: Effect of Bioflocculant QZ-7 Concentration on TSS Removal.

The result obtained by statistical analysis in Table 4.21 and Table 4.22 found that there is significant difference (p<0.000) between pure and crude bioflocculant on TSS removal.

Table 4.21: Analysis of Variance for the Effect of Pure Bioflocculant QZ-7 on TSSRemoval.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P- value
Between Groups	732.375	5	146.475	135.995	0.000
Within Groups	12.925	12	1.077		
Total	745.300	17			

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	965.433	5	193.087	184.969	0.000
Within Groups	12.527	12	1.044		
Total	977.960	17			

Table 4.22: Analysis of Variance for the Effect of Crude Bioflocculant QZ-7 on TSSRemoval.

4.8.1.3 Chemical Oxygen Demand (COD) Removal

Results showed in Figure 4.28, that the COD reduction was increased dramatically as the bioflocculant QZ-7 concentration increased from 0.5 to 5 mg/L. The highest reduction of COD was 91.2% of optimum bioflocculant concentration at 5 mg/L, then the COD reduction was dropped as the purified bioflocculant QZ-7 concentration increased from 5-7 mg/L.





Based on the statistical analysis of the effect of pure and crude bioflocculant QZ-7 on COD removal are presented in Table 4.23 and Table 4.24, revealed that there is significant difference (p<0.000), between the two types of QZ-7.

Table 4.23: Analysis of Variance for the Effect of Pure Bioflocculant QZ-7 on CODRemoval.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	3540.608	5	708.122	505.946	0.000
Within Groups	16.795	12	1.400		
Total	3557.403	17			

Table 4.24: Analysis of Variance for the Effect of Crude Bioflocculant QZ-7 on CODRemoval.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1905.352	5	381.070	330.090	0.000
Within Groups	13.853	12	1.154		
Total	1919.205	17			

4.8.2 Removal of Heavy Metals using Bioflocculant

4.8.2.1 Effect of pH on the Heavy Metal Adsorption.

Figure 4.29, showed that the highest adsorption 89.3% of As, 81.3% of Zn^{+2} , are reported at pH 7, whereas, the highest adsorption 76.1% of Cu^{+2} , 77.9% of Pb^{+2} and 68.7% of Cd^{+2} were reported at pH 9 and the highest adsorption 68.4% of Hg^{+2} was at pH 5.



Figure 4.24: Effect of pH on Heavy Metal Adsorption using 20 mg/L of QZ-7.

Regarding to the analysis of variance of the effect of different pH value on arsenate removal efficiency is presented in Table 4.25. It is revealed that the different pH value has a significant difference (p<0.00) on as removal.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	2555.898	6	425.983	274.912	0.000
Within Groups	21.693	14	1.550		
Total	2577.591	20			

Table 4.25: Analysis of Variance for the Effect of pH on Arsenate Removal Efficiency.

As depicted in Table 4.26, there is a significant difference (p<0.000) between the different pH value on cupper removal.

Table 4.26: Analysis of Variance for the Effect of pH on Cupper Removal Efficiency.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1141.283	6	190.214	205.056	0.000
Within Groups	12.987	14	.928		
Total	1154.270	20			

Table 4.27, shows that there is a significant difference (p<0.000) between the different pH value on cadmium removal.

Table 4.27: Analysis of Va	riance for the Effect of p	H on Cadmium Removal	Efficiency.
----------------------------	----------------------------	----------------------	-------------

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	738.030	6	123.005	118.600	.000
Within Groups	14.520	14	1.037		
Total	752.550	20			

Table 4.28, presented the statistical treatment of the effect of different pH value on lead removal efficiency. It is revealed that there is a significant difference (p<0.000) between the groups.
	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1707.656	6	284.609	687.779	0.000
Within Groups	5.793	14	0.414		
Total	1713.450	20			

Table 4.28: Analysis of Variance for the Effect of pH on Lead Removal Efficiency.

As depicted in Table 4.29, showed that there is a significant difference (P<0.000) between the different pH value on the zinc efficiency removal. However, multiple comparison analysis presented in appendix F23, found that no significant difference (p>0.05), have been detected between pH value such as pH 3 and 11, pH 6 and 9, and pH 7 and 9.

Table 4.29: Analysis of Variance for the Effect of pH on Zinc Removal Efficiency.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1040.038	6	173.340	131.603	.000
Within Groups	18.440	14	1.317		
Total	1058.478	20			

The result obtained by statistical analysis presented in Table 4.30, showed that there is a significant difference (p<0.000), between the different pH value of mercury removal efficiency. Moreover, the multiple comparison shown in appendix F24, revealed that there is no significant difference (p>0.05) between the following pH 3 and 6, pH 4 and 5, 7 and 9, and pH 6 and 7.

Table 4.30: Analysis of Variance for the Effect of pH on Mercury Removal Efficiency.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1067.392	6	177.899	29.260	.000
Within Groups	85.120	14	6.080		
Total	1152.512	20			

4.8.2.2 Effect of the Bioflocculant QZ-7 Concentration on Heavy Metals Adsorption

The results obtained in Figure 4.30 showed that an exceed 50% of heavy metals were removed at 20 mg/L - 100 mg/L of bioflocculant QZ-7 concentration. Whereas, the highest metal removal was achieved at 60 mg/L of bioflocculant concentration, as 84% of As, 80.7% of Cu^{+2} , 78.5% of Zn^{+2} and 77.4% Cd^{+2} and Pb⁺² 72.43% at 100 mg/L.





As shown in Table 4.31, there is a significance difference (p<0.000) of different QZ-7 concentration. Besides, in accordance with the multiple analysis in appendix F26, revealed that there is no significant difference (p>0.05), between 40 mg/L and 80 mg/L and 100 mg/L.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	740.568	5	148.114	35.041	0.000
Within Groups	50.723	12	4.227		
Total	791.291	17			

Table 4.31: Analysis of Variance for the Effect of QZ-7 Concentration on Removal

 Efficiency of Arsenate.

Table 4.32, showed that there is a significant difference (p<0.000), between the different concentration of QZ-7 on cupper removal. However, as revealed in multiple comparison appendix F25. That there is no significant difference between 20 mg/L and 40 mg/L, also between 60 mg/L and 80 mg/L or 100 mg/L.

Table 4.32: Analysis of Variance for the Effect of QZ-7 Concentration on Removal

 Efficiency of Cupper.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1959.031	5	391.806	263.448	0.000
Within Groups	17.847	12	1.487		
Total	1976.878	17			

Table 4.33, showed that there is a significant difference (p<0.000), between the different concentration on cadmium removal efficiency. Whereas, the multiple comparison appendix F27, showed that there is no significant difference (p>0.05), between 20 and 60 or 80 mg/L, or 40 and 60 mg/L and 80 and 100 mg/L.

Table 4.33: Analysis of Variance for the Effect of QZ-7 Concentration on Removal

 Efficiency of Cadmium.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1728.051	5	345.610	187.483	0.000
Within Groups	22.121	12	1.843		
Total	1750.172	17			

Table 4.34, presented that there is a significant difference (p<0.000), of different concentration on Pb^{+2} removal efficiency. While, the multiple comparison in appendix F28, indicated that there is no significant difference between 40 mg/L and 80 mg/L or between 60 mg/L and 100 mg/L.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	983.998	5	196.800	95.405	0.000
Within Groups	24.753	12	2.063		
Total	1008.751	17			

Table 4.34: Analysis of Variance for the Effect of QZ-7 Concentration on RemovalEfficiency of Lead.

Table 4.35, indicated that there is a significant difference (p<0.000) of different QZ-7 concentrations on the removal efficiency of zinc. Moreover, based on the multiple comparison analysis presented in appendix F29, revealed that there is no significant difference (p>0.05) between 20 mg/L and 40 mg/L, 60 mg/L and 100 mg/L of QZ-7 concentrated on the zinc removal. Also, between 40 mg/L and 80 mg/L or 100 mg/L.

Table 4.35: Analysis of Variance for the Effect of QZ-7 Concentration on Removal

 Efficiency of Zinc.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	3389.997	5	677.999	299.776	0.000
Within Groups	27.140	12	2.262		
Total	3417.138	17			

4.8.2.3 Effect of Heavy Metals Concentration on the Adsorption using of Bioflocculant QZ-7

Results presented in Figure 4.31 shows that the heavy metals adsorption exceeded 60%, with the increased of it's an initial concentration of 10 mg/L to 100 mg/L. For example, indicated the maximum removal achieved 94.3% of 60 mg/L As and the highest removal of Cu^{+2} 85.2%, Cd^{+2} (83.15%), Zn^{+2} (84.5) and Pb⁺² (82.76%), of 100 mg/mL.



Figure 4.26: Effect of Metal Concentration on the Efficiency of Removal Heavy Metals using 60 mg/L of Bioflocculant QZ-7.

As presented in Table 4.36, there is a significant difference (p<0.000) of the different concentration of arsenate. Whereas, the multiple comparison analysis in appendix F30, showed that there is no significant difference (p>0.05), between 40 mg/L of arsenate and 80 mg/L or 100 mg/L, also 60 mg/L and 100 mg /L.

Table 4.36: Analysis of Variance for the Effect of Arsenate Concentration on QZ-7

 Adsorption Efficiency.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	897.752	5	179.550	91.711	.000
Within Groups	23.493	12	1.958		
Total	921.245	17			

Table 4.37, showed that there is a significant difference (p<0.000), of the cupper concentration on the adsorption of QZ-7 performance. Furthermore, multiple comparison analysis in appendix F31, revealed that there is no significant difference (p>0.05), of 20 mg/L of cupper and 40 mg/L, or between 60 mg/L and 80 mg/L, 100 mg/L.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1137.910	5	227.582	100.779	0.000
Within Groups	27.099	12	2.258		
Total	1165.009	17			

Table 4.37: Analysis of Variance for the Effect of Cupper Concentration on QZ-7Adsorption Efficiency.

Table 4.38, showed that there is a significant difference (p<0.000), of cadmium concentration on the adsorption of QZ-7 performance. Despite the fact that, the results revealed in multiple comparison appendix F32, there is no significant difference (p>0.05) between 20 mg/L of cadmium and 60 mg/L, also between 40 mg/L and 60 mg/L or 80 mg/L.

Table 4.38: Analysis of Variance for the Effect of Cadmium Concentration on QZ-7

 Adsorption Efficiency.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	1411.147	5	282.229	65.704	.000
Within Groups	51.545	12	4.295		
Total	1462.692	17			

Table 4.39, there is a significant difference (p<0.000), of lead concentration on the QZ-7 adsorption performance. Although, multiple comparison analysis in appendix F33, indicated that there is no significant difference (p>0.05), of 10 mg/L lead concentration and 40 mg/L or 60 mg/L, or between 20 mg/L, 60 mg/L, and 80 mg/L and 100 mg/L.

Table 4.39: Analysis of Variance for the Effect of Lead Concentration on QZ-7Adsorption Efficiency.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	563.465	5	112.693	26.520	0.000
Within Groups	50.991	12	4.249		
Total	614.457	17			

Table 4.39, shows that there is a significant difference (p<0.000) of zinc on QZ-7 adsorption performance. Despite the fact that, the multiple comparison analysis in appendix F34, revealed that there is no significant difference (p>0.05), of 20 mg/L zinc and 40 mg/L, 60 and 80 mg/L, also between 40 mg/L and 80 mg/L and 100 mg/L.

Table 4.40: Analysis of Variance for the Effect of Zinc Concentration on QZ-7Adsorption Efficiency.

	Sum of Squares	Degrees of Freedom	Mean Square	F-ratio	P-value
Between Groups	279.373	5	55.875	17.071	0.000
Within Groups	39.277	12	3.273		
Total	318.650	17			

4.8.3 Treatment of Real Wastewater

4.8.3.1 Removal of Heavy Metals from Industrial Wastewater by Bioflocculant QZ-7

Figure 4.32 showed the efficient removal of arsenate (As), from 71.9% to 89.85% with the increased of the amount of adsorbent concentration, as well as the Zn^{+2} removal was also increased from 63.85% to 77.3%. Whereas, the Cu⁺² slightly removed from 53.2% to 58.4 %.



Figure 4.27: Removal Efficiency of Heavy Metals using Different Concentrations of Bioflocculant QZ-7.

4.8.3.2 Removal of BOD and COD from Industrial Wastewater

The conjugation of COD analysis test with BOD test are useful in sign of toxic

circumstances and the incidence of biological resistance (Poddar & Sahu, 2017).

Demonstern	Removal Efficiency %			
Parameter	Sugar mill	Ethanol	Rubber	
BOD mg/L	92.48	22.15	57.2	
COD mg/L	93	25.8	61.1	
Colour	39.4	0	62.2	
Turbidity (NTU)	90.7	0	56.6	
Total dissolved solid mg/L	88.2	52.1	61.2	
Total suspended solid mg/L	89.69	null	61.5	

Table 4.41: Removal Efficiency of Bioflocculant QZ-7 in Industrial Wastewater

 Treatment.

4.8.3.3 Removal of Pharmaceutical Compounds using Bioflocculant QZ-7

The primary elimination of selected pharmaceuticals as shown in Table 4.42, from

hospital wastewater was determined at the Laboratory.

Table 4.42: Efficiency of Bioflocculant QZ-7 in Removal of Pharmaceutical Compounds.

Pharmaceutical compounds	Removal efficiency %		
Simvastatin	92.45939		
Perindopril	25.49839		
Nifedipine	0		
Diclofenac	4.046219		
Mefenamic acid	13.64833		
Salbutamol	88.69775		
Caffeine	66.52178		
Acetaminophen	69.94135		
Metformin	7.193886		
Ibuprofen	8.773372		

4.9 Synthesis of Nanoparticle

The biopolymers-based synthesis of AgNPs has been carried out since a decade and a variety of polymers viz., algal, bacterial, fungal and plant-based polymers were exploited greatly.

4.9.1 Nutrient Broth Medium

The nutrient broth supplemented with glucose showed synthesis of AgNPs which was primarily detected by visual colour change Figure 4.33 in a bottle on the 3^{rd} day of inoculation with *B. salmalaya* 139SI.



Figure 4.28: Synthesis of AgNPs in Nutrient Broth Inoculated with B. salmalaya 139 SI.

4.9.2 Using Cell- Free Supernatant

In the cell-free supernatant having bioflocculant QZ-7 the colour change was detected after 72 h of the incubation. The bioflocculant QZ-7 reduced AgNO₃ to produce AgNPs, Figure 4.34.



Figure 4.29: Synthesis of AgNPs using Cell-Free Supernatant.

4.9.3 Synthesis of AgNPs using Purified Bioflocculant QZ-7

Figure 4.35 showed the synthesis of AgNPs by using purified bioflocculant QZ-7 after 24h.



Figure 4.30: Synthesis of AgNPs by Purified QZ-7.

4.10 Characterisation of AgNPs.

4.10.1 Characterisation of Ag NPs by UV-Vis Spectroscopy

The primary synthesis of Ag NPs was recognized by UV–vis spectral analysis. In this particular absorption spectrum, a strong, broad peak located at approximately 400-450 nm, was observed for nanoparticles synthesized using the cell free supernatant of QZ-7 (Figure 36). This was revealing of the presence of nanoparticles and was reconfirmed by the manifestation of a similar strong, broad peak after a two-month period.



Figure 4.31: UV-vis Spectra of Synthesized AgNPs.

4.10.2 Characterisation of AgNPs by FTIR

FTIR spectroscopy was used to determine the interaction between glycoprotein and metallic particles. Figure 4.37, shown the reduction of silver ions and stabilisation of silver nanoparticles was reflected as intense absorption bands at 2921.48, 2851.09, 2321.94, 2163.83, 2120.46, 1578.26, 999.07, 478.31 and 463.30 cm⁻¹.



Figure 4.32: FTIR Spectrum of Synthesized AgNPs.

4.10.3 FE-SEM and EDAX Analysis

The FE-SEM and EDAX were carried to study the morphology and elemental composition of biosynthesized AgNPs. The FE-SEM clearly shows that the particles is spherical as shown in Figure 4.38a. The elemental composition of AgNPs is examined by EDAX spectroscopy and is a chemical analysis method combined with FE-SEM Figure 4.38b. (Lawrence & Prakash, 2019).



Figure 4.33: FE-SEM (a) and EDAX (b) Analysis of Synthesized AgNPs

4.10.4 Characterisation of AgNPs by XRD Analysis

As illustrated in Figure 4.39 the XRD spectrum of AgNPs showed 4 intense peaks in rang of 20-80° at 2θ value; particularly at 38.2667°, 44.4964°, 64.5751°,77.5673° and 82.23° with conforming degree of 111, 200, 220, 311 and 222, reflection of metallic silver. This data was found to be in agreement with database of the Joint Committeeon Powder Diffraction Standards (JCPDS File no. 04-0783).



Figure 4.34: XRD Patterns for the AgNPs Synthesized by Pure Bioflocculant QZ-7.

4.11 Antibacterial Activity of AgNPs Synthesized by Bioflocculant QZ-7

The zone of inhibition was measured after 24 h and results were presented (Table 4.43 and Figure 4.40). *E. coli* ATCC35401, *Pseudomonase aeruginosa*, *Salmonella entertidis* ATCCBAA-711 and *staphylococcus aureus* ATCC2592 showed almost consistency in the zone of inhibition.

Table 4.43: Antimicrobial Activity of AgNPs in Solid Medium.

	Zone of inhibition at various concentration of AgNPs (mm)				
Microorganisms	20 µL	40 µL	60 µL	80 µL	
E. coli ATCC35401	15	18	23	24.5	
S. enteritidis ATCC BAA-711	16.5	21	24.5	26	
P. aregonosa	16	19	22	22	
S.aureusATCC2592	15	16	18	17	



Figure 4.35: Antibacterial Activity of AgNPs against *E. coli* ATCC35401, *P. aeruginosa, S. entertidis* ATCC BAA-711 *and Staph. aureus* ATCC2592.

CHAPTER 5: DISCUSSION

5.1 Selection of Bacterial Strain for Bioflocculant Production

A total of 15 *Bacillus salmalaya* 139SI strains were screened for bioflocculant production. From 15 strains appendix A-1, five strains presented the highest flocculating activities when tested against kaolin suspension (4g/L) as shown in Table 4.1. The strain *B. salmalaya* 139SI (BS 139SI-7), that was selected for this study produced a bioflocculant that showed the highest flocculating activity of 83.3% for kaolin clay suspension. Moreover, it is the first time that *B. salmalaya* 139SI strain used for bioflocculant production.

Based on the morphological and physiological characteristics, the selected strain are shown Figure 4.1a shows that the *B. salmalaya* strain 139SI colonies were large and white-opaque with a rough and irregular edge. The colonies were 2 mm–3 mm in diameter and showed strong β - haemolytic activity after 18h–24 h of incubation at 37°C on 5% v/v sheep blood agar, as shown in Figure 4.1b. Moreover. Figure 4.2, the *B. salmalaya* strain 139SI was observed under the microscope as gram-positive, short rods in chains and the formation of spore, which is characteristic of the *B. salmalaya* 139SI.

5.2 Optimisation of Bioflocculant Production by *B. salmalaya* 139SI

The inceptive pH of the fermentation media directly influenced the synthesis of the bioflocculant. The effect of initial pH of the culture broth on *B. salmalaya* 139SI growth and bioflocculant production are illustrated in Figure 4.3. At a glance it can be said that the growth of *B. salmalaya* 139SI and bioflocculant production was increased gradually by increasing of pH 3 to 7, while *B. salmalaya* 139SI growth and bioflocculant activity were decreased at pH higher than pH 7. The maximum flocculation activity in the fermentation broth culture was reported with 83.7% at pH 7. A similar result was obtained for bioflocculant production by *Halomonas* sp. and *Citrobacter* sp. TKF04 (Mabinaya *et*

al., 2011; Fujita *et al.*, 2000), and the increases in bioflocculant yield differ in different strains at their optimal pH (Xia *et al.*, 2008). The nutrient absorption capability of the cells and the presence of high electric charge could influence the enzymatic reaction (Nakata & Kurane, 1999; Xia *et al.*, 2008), thereby decreasing flocculating activity. At the neutral pH, the nutrient absorption capability of the cells was high due to the neutral electric charge, indicating that the highest flocculating activity was observed in the broth culture. Moreover, extreme pH not only inhibits microbial growth, but also bioflocculant (EPS) production, leading to changes in the morphological properties of cells and molecular weight of EPS compounds (Chug *et al.*, 2016). The differences between the effect of pH values on bioflucculation activity was found to be significant (p<0.000) as reported in Table 4.2. Moreover, the multiple comparison analysis shows the differences between each pH values on bioflucculation activity as presented in appendix F1.

The influence of inoculum size on bioflocculant production was determined according to Zhang *et al.* (2013). The differences between inoculum sizes showed a certain effect on the flocculating production and cell mass values, as shown in Figure 4.4. The obtained optimal inoculum size was 5% (v/v). The outcome results were analogous to those reported by Wang *et al.* (2007), where an optimal inoculum volume at 5% was detected for *Klebsiella mobilis*. The maximum flocculation activity was not in agreement with the maximum cell growth, but the flocculation activity for *B. salmalaya* strain 139SI was maintained within a certain range of inoculum volume. This condition was due to the influence of inoculum volume on microbial growth. A small inoculum volume, extended the lag phase growth, while outsized inoculum made the niche of the strain overlap extremely. Subsequently inhibiting the bioflocculant production due to inadequate supply of nutrient (Zhang *et al.*, 2014). While, the analysis of variance presented in Table 4.3, for the effect of the different inoculum size of *B. salmalaya* 139SI on flocculating activity was revealed a significant difference (p<0.000).

The influence of temperature on bioflocculant production by *B. salmalaya* 139SI was investigated. As shown in Figure 4.5, the highest bioflocculating activity 81.9% was obtained for the bioflocculant produced at 35.5° C. However, the flocculating activity of bioflocculant start to decrease at higher temperatures than 35.5° C, due to the metabolism of microorganisms is directly affected to a cultivation temperature degree (Xia *et al.*, 2008; Zhang *et al.*, 2007). Maximum enzymatic activation can only be obtained at an optimum temperature (Lixin *et al.*, 2016). The optimal temperature for *B. salmalaya* 139SI was found to be 35.5° C for the bioflocculant gradually declined. This decrease could be due to denaturation of extracellular enzyme at the higher temperatures.(Zhong *et al.*, 2018). Moreover, as showed in Table 4.4, the analysis of variance for the effect of different temperature on bioflocculant production, was revealed that there is a significant difference (p < 0.001).

The effect of shaking speed on bioflocculant production is shown in Figure 4.6. The flocculating activity start gradually to be increased under the effect of shaking speed between 100 to 160 rpm. While shaking speed higher than 160 rpm caused a decline in the flocculating activity. Agitation speed basically determines the concentration of dissolved oxygen, which can also affect microbial nutrient absorption and enzymatic reaction hence affecting both bacterial growth and bioflocculant production (Lixin *et al.*, 2016). High-speed mixing have a tendency to break the flocs, causing the coagulated cells to be redispersed and presented again into the medium (Ahmed *et al.*, 2011). Some scholars have recommended rapid mixing followed by slow mixing during the flocculation process. This phenomenon is caused by the restabilisation of the cells at high

mixing speed (Ahmed *et al.*, 2005). Furthermore, the analysis of variance shown in Table 4.5, for the effect of shaken culture at 72 h of incubation on bioflocculant production was found that there is a significant difference of the flocculating activity of shaken speed (p<0.000). In the course of the early growth phase, the biomass and bioflocculant production were lower, resulting in lower viscosity of culture broth and oxygen demand, when the strain *B. salmalaya* 139SI proceed in the logarithmic and stationary phases, the biomass and the bioflocculant production increased, and the corresponding viscosity of the culture broth and the oxygen demand also increased.

When the broth cultures of *B. salmalaya* 139SI were cultivated in non-shaking condition, it was found that the flocculating activities was revealed very much decreased as shown in Figure 4.7. The subsequent measurement showed that the maximum bioflocculant production occurred between 24-72 h. Compared to the shaking speed there is significant decreased of 20 % flocculating activity can be seen at 72 h of incubation, due to the agitation of the culture medium (Salehizadeh *et al.*, 2001). Furthermore, the concentration of dissolved oxygen can affect microbial nutrient absorption and enzymatic reaction, hence affecting both bacteria growth and bioflocculant production (Lixin *et al.*, 2016). Therefore, the yield of bioflocculant produced by *B. salmalaya* 139SI is significantly reduced under static condition. Therefore, an experiment was conducted to check the effect of bioflocculant production and measurement in terms of flocculation activity under shaking and non-shaking condition. Table 4.6 showed the analysis of variance in bioflocculanting performance of a static culture condition at 24 -168 h of incubation it was revealed that there are significant differences between the flocculating activity of static condition (p<0.000) at different incubation period. Moreover, the

multiple comparison analysis designates there is no significant differences between the 24 h and 96 h (p>0.05), appendix F4.

The carbon source tends to be the precursor for the metabolic pathway. The effect of carbon sources including sucrose, glucose, lactose, maltose, and starch as single sources on bioflocculant synthesis by *B. salmalaya* 139SI was evaluated as shown in Figure 4.8. In addition, the bioflocculant synthesis was also tested under the effect of a mixture of carbohydrate (CHO) which includes sucrose and glucose. The highest 90.1% and lowest 28.9% flocculating activities were obtained with a mixed carbohydrate source and starch with (1:1) mixing ratio, respectively. Moreover, as single carbon sources such as sucrose, glucose, lactose, and maltose were markedly appropriate for bioflocculant production with flocculating activity exceeding 60% after 72 h of cultivation period. Aljuboori et al. (2013) found that sucrose is the most desirable to Aspergillus flavus for bioflocculant production. The optimal composition of the growth medium for alkaliphilic bacterium Paenibacillus jamilae was found to be sucrose 25 g/L, urea 5 g/L, and KH₂PO₄ 5 g/L. (Zhong et al., 2018). Moreover, based on analysis of variance revealed in Table 4.7, that the overall result analysed there are significant differences (p < 0.000), between the bioflocculating activity of the different carbon source groups. Based on multiple comparison analysis presented in appendix F5, revealed that there is no significant difference between sucrose and glucose, lactose and maltose, respectively. To conclude, glucose and sucrose were the best carbon source to be utilized as they are relatively cheap. Glucose is favoured by all bacteria for the production of EPS e.g. Cryseobacterium daequene (Liu et al., 2010), while, Alkaligenus cupidus KT201 preferred sucrose for maximum production (Toeda & Kurane, 1991). The novel bioflocculant REA-11 synthesised by Corynebacteria glutamicum, sucrose was utilised as a carbon source containing corn steep liquor (He et al., 2004). It is well documented in the literature that various bioflocculant-producing microorganisms prefer organic carbon sources for

optimum bioflocculant production. On the other hand, the utilisation of inorganic carbon sources for bioflocculant production is quiet slight in literature (Makapela et al., 2016). The nitrogen source plays a crucial role in synthesis of bioflocculant (Cosa et al., 2011). The effect of nitrogen source on production was evaluated using different sources of organic origin, such as yeast extract, tryptone, peptone, urea and beef extract and ammonium sulphate was used as a source of inorganic nitrogen. In addition, the bioflocculant synthesis was also tested under the effect of mixture of nitrogen which include urea and yeast extract. Different microorganisms utilize nitrogen derived from both or either inorganic or organic sources for production (Okaiyeto et al., 2014). The influence of nitrogen sources, both organic and inorganic on the bioflocculant synthesis by *B. salmalaya* strain 139SI is illustrated in Figure 4.9. The highest flocculation activity reached at 72 h of cultivation was obtained with mixed nitrogen source 72.1%, while yeast extract gives 64.5% and urea 61.5%, whereas the lowest activity was detected with peptone 42.5%. A similar result was reported by You et al. (2008) that the B. subtilis, B. fusiformis and B. flexus preferred urea and yeast extract for the bioflocculant production. While the ammonium sulphate as inorganic nitrogen gives about 58.86% of flocculating activity. As reported by Ugbenyen et al. (2014), the flocculating activity was peaked with yeast extract followed by casein hydrolysate, tryptone, and beef extract. A similar finding indicated that organic nitrogen is more appropriate for bioflocculant production than inorganic nitrogen origin, because it is simply absorbed by the microbial cells (Cosa et al., 2012). Moreover, Ismail and Nampoothiri, (2010), indicated that yeast extract was a highly favourable nitrogen source that improved EPS production by Lactobacillus planetarium MTCC 9510. While, based on the analysis of variance shown in Table 4.8, indicated that there are significant differences of bioflocculant production of different nitrogen source (p<0.000). However, the multiple comparison analysis showed in

appendix F6 that there no significant differences between mixed nitrogen and urea or yeast extract, and between urea, yeast extract, tryptone and $(NH_4)_2 SO_4$ (p>0.05).

Bioflocculant production affected by the presence or absence of metal ions due to its responsible for activation of some of the biosynthetic pathways (Raza *et al.*, 2011). Table 4.9. Show that the flocculating activity was stimulated by several salt such as CaCl₂, KH₂P₄, K2HPO₄, KCl, MgSO₄ and Na₂HPO₄, but were adversely affected by CuSO₄ and Al₂ (SO₄)₃. While combination of KH₂P₄, and K₂HPO₄ as inorganic salts in the production medium for *B. salmalaya* 139SI significantly enhance bioflocculant production. Moreover, the presence of KH₂P₄ and K₂HPO₄ in the production medium are considered a good buffer, it alleviates the effect of pH change during the cultivation period. While, the bioflocculant yield by *A. flavus* was enhanced by the addition of K⁺, Na⁺, Ca⁺², Mg⁺² and Mn⁺² into the fermentation medium (Aljuboori *et al.*, 2015). Table 4.10, indicate that the overall results of treatment analysis showed that there are significant differences (p<0.00) between the bioflocculant production of different inorganic salts. However, based on multiple comparison analysis as shown in appendix F7, that there are no significant differences between the following CaCl₂, KH₂P₄, K₂HPO₄ and KCl (p>0.05).

5.3 Time Course for Bioflocculant Production by *B. salmalaya* Strain 139SI

The bioflocculant production by an organism is associated to its stages of the growth phase, and bioflocculant can be produced during growth phase through, biosynthesis consuming available nutrients or during decline phase due to autolysis of cells. As shown in Figure 4.10, the production of bioflocculant at the early growth stage of *B. salmalaya* 139SI was reflected by the 70% flocculating activity achieved at 24 h of incubation. The subsequent measurements showed that the maximum bioflocculant production occurred between 24-96 h of incubation and the flocculating activities achieved above 70%.

Whereas, the highest bioflocculant activity 92.6% achieved at 72 h of incubation and pH 7.0 a logarithmic growth phase. The stationary growth phase was attained after 72 h of cultivation. Flocculating activity ran parallel to cell growth, thereby exhibiting a concomitant increase in bioflocculant production with cell growth. The optical density of the culture increased progressively during the period 24 h-72 h with an increase in flocculating activity produced, although, after the 72 h the OD decreased with decreased of flocculation activities produced. This observation indicated that the production of bioflocculant was a result of biosynthesis during bacterial growth and not via cell autolysis (Czemierska et al., 2017). The decrease in flocculating activity detected after 72 h might be attributed to the existence of bioflocculant-degrading enzymes produced by the microorganisms (Li et al., 2009). A similar study was reported by Zheng et al. (2008), who found that the flocculating activity of the bioflocculant produced by Serratia fiacre and Bacillus sp. F19 reaches its maximum at the early stationary phase of 72 h. The initial pH of the production medium was adjusted to 7 and then monitored at regular intervals over the entire fermentation period. Therefore, factors affecting bioflocculant (EPS) production like growth phase, pH and temperature have been optimized for A. beijreinckii and B. subtilis. Both bacteria were found to produce maximum EPS at neutral pH (chung et al., 2016). Also, the flocculating activity of a novel bioflocculant from Paenibacillus jamilae increased significantly with time, achieving a peak rate of 87.5% at 72h and declining after 96h (Zhong et al., 2018). Moreover, the pH of the production medium governs the oxidation- reduction potential and the cells electrical charge thereby affecting enzymatic reaction and nutrient absorption (Zhang et al., 2013). Consequently, the pH of the fermentation medium decreased as cultivation period increased in time. The decrease in pH of the fermentation medium might be due to the production of organic acids as a result of glucose metabolism, because glucose was a constituent of the production medium or the presence of organic acids metabolically produced by bacteria

(Lu *et al.*, 2005). The produced amount of bioflocculant was 2.72 g/L. While, based on the analysis of variance for the time cource of bioflocculant production was presented in Table 4.11, revealed that there is a significant difference (p<0.00).

5.4 Characteristics of Bioflocculant QZ-7

The composition of bacterial bioflocculants plays a role in their flocculating activities (Wu & Ye, 2007). Reports showed that many types of bioflocculants comprise proteins, polysaccharides, glycoproteins, and glycolipids (Salehizadeh & Shojaosadati, 2003). Table 4.12, presents a clear indication that the bioflocculant composed of polysaccharides, proteins and uronic acid was found. On the basis of the chemical analysis of the produced bioflocculant, shown in Figure 4.11 and Figure 4.12, the total carbohydrate and protein compositions were found to be 79.08% and 15.4%, respectively. The bioflocculant mainly composed of polysaccharides and proteins. Further analysis revealed the presence of uronic acid in the bioflocculant. An adequate proportion of uronic acid molecules in the bioflocculant, carboxyl groups can be provided to the molecular chain. The carboxyl group present in the molecular sequence affords more active sites for constituent parts, so several molecules can bind to the elongated molecular chain (Aguilera *et al.*, 2008).

Fourier transform infrared (FTIR) spectroscopy was used to analyse the purity of bioflocculant, as illustrated in Figure 4.13, FTIR spectroscopy of purified QZ-7 biopolymer indicted a broad stretching intense peak at 3425.56 cm⁻¹ demonstrated the presence of hydroxyl and amino groups. The weak stretching band detected at 2929.82 cm⁻¹ revealed the presence of aliphatic C—H stretching. This result is similar to the results obtained by Ayat *et al.* (2017). The peaks at 1658.90 and 1432.97 cm⁻¹ were attributed to C=O and vibration COOH vibration, respectively, whereas, that at 1432.97 cm⁻¹ was due to the C=O antisymmetric extension in the carboxylate (Sun *et al.*, 2015a),

and thus the result showed the existence of carboxylate function groups in QZ-7. The carboxyl group may also serve as a functional moiety for the generation of modified or new polymers in different forms via different approaches, such as unique designed formulation by assembling such polymer to other synthetic polymers. Other bands observed at 1109.66 and 1187.49 cm⁻¹ were identified to be classic characteristics of all compounds derived from sugar, including sugar derivatives like mannuronic acid, guluronic acid, and uronic acid (Suh et al., 1997). Other absorption bands at 924.94 and 990.44 cm⁻¹ were related to the β -glycosidic bond linking the monomeric units present in sugars as recommended by the study Gupta et al., (1987). The corresponding small band absorption was observed at 446.32, 538.79, and 618.50 cm⁻¹, which indicated that the bioflocculant QZ-7 was a protein-bound polysaccharide (Yang et al., 2013). Similar finding of Makapela et al. (2016); Gommaa. (2012) that the FTIR spectrometry analysis of purified bioflocculant from Bacillus pumilus, Pseudomonas aeruginosa ATCC-10145 respectively, showed the display of amino, hydroxyl and carboxyl groups. Furthermore, the FT-IR spectra of the EPS peroduced by Leu. pseudomesenteroides YB-2 was highly similar to those of W. cibaria JAG8 (Tingirikari et al., 2014) and Leu. pseudomesenteroides R2 (Paulo et al., 2012), indicating a similar structure. The α -(1 \rightarrow 6) linkages were further confirmed by ¹H and ¹³C NMR analysis.

The high-performance gel permeation chromatography (HPGPC) spectrum of the purified bioflocculant QZ-7 exhibited a symmetrical and sharp peak in the retention time of 6.53 and 9.95, as shown in Figure.4.14. The molecular mass–retention time equation in accordance with the calibration curve is expressed as following:

Log (molecular weight) = -0.1368T + 8.3496

The average weight of the bioflocculant QZ-7 was calculated to be 5.13×10^5 Da, which was much higher than the weight of other bioflocculants (Yang *et al.*, 2015).

Bioflocculants with high molecular weight present stronger bridging, more adsorption points, and higher flocculating activities than those with low molecular weight (Salehizadeh & Shojaosadati, 2001). Besides, high molecular weight of EPSs were considered important for stimulating the formation of EPS-protein network structure and improving the consistency of fermented products (Du *et al.*, 2018).

The ¹H-NMR spectra of bioflocculant QZ-7 produced by *B. salmalaya* strain 139SI significantly reflected the results obtained from FTIR analyses. In the ¹H NMR spectrum (Figure 4.15), the anomeric region (3.29–5.47ppm) signals often functioned to differentiate the anomeric protons of sugar residues in bioflocculant QZ-7.The ¹H NMR spectra of QZ-7 confirmed the resonance of hydrogen protons as corroboration of the glucosyl residue as the repetitive unit of the biopolymer (Ye et al., 2019). The spectral analysis of bioflocculant QZ-7 presented in Figure.4.15, revealed the presence of a signal at 2.5 ppm, indicating the presence of -NH group. The solvent peak showed at 3.29 ppm. Furthermore, signals at 5.04- 5.47, 3.6-3.7 and 1.2-1.43 ppm confirmed the presence of hydroxyl proton, non-anomeric protons and CH₂ linkage in the sample, respectively. Again, a peak at 2.0 ppm, indicated the presence of carbonyl linkage of glycoprotein molecule. The H NMR chemical shift (δ) value at 0.85ppm–0.86 ppm an indicator of the presence methylene (CH₃) groups, also at 1.20 ppm–1.33 ppm indicated the methyl (CH₂) groups consistence in sugar moiety. Therefore, the chemical shift was detected in the NMR spectra of many bioflocculants although a similar finding was stated by Wenliang *et al.* (2015). The value of CH-proton of carbon of deoxy sugar exhibited δ at 1.747 ppm (Bayoumi et al., 2010). The strong water signal (HOD) showed at 2.5 ppm, and clouds the β -anomeric proton signals at about 5.04 ppm. The δ value at 5.05-5.47 ppm indicated the proton on anomeric carbon of sugar moieties. The existence of NH₂ group of the protein in bioflocculant was detected in the range of δ value of 4.65 ppm–5.03 ppm. Likewise, the chemical shift of the bioflocculant QZ-7 produced by B. salmalaya 139SI

was found to be as significance and assignment of the bioflocculant produced by *leuconostoc lactis* KC117496 (Saravanan & Shetty, 2016). The ¹³C NMR spectrum (Figure. 4.16) of bioflocculant QZ-7 included the anomeric carbon region (60–101 ppm) and ring carbon (60–79.9 ppm) region (Chen *et al.*, 2016). The anomeric carbon signal at 79.9 ppm confirmed that the sugar residues were linked by α -glycosides (Ye *et al.*, 2019). The peaks at 70.8, 72.2, 72.6, 73.8 and 78.9 ppm corresponded to C-2, C-3, C-4 and C-5, respectively (Seymour *et al.*, 1979). The downfield shift of the C-6 carbon signal for the glucose unit was detected at 61.2 ppm (Chen *et al.*, 2016). The result achieved from the NMR analysis showed the FTIR result and specified that the bioflocculant QZ-7 was a glycoprotein (Zaki *et al.*, 2013).

Characterisation and determination of EPS monomers and even the associated proteins were been possible through HPLC, LCMS/MS, and LC/MS-ESI techniques (Jachlewski *et al.*, 2015). As a result, obtained by the LC/MS based analysis, adducts of Na⁺, NH4⁺ and H⁺ ions were identified for the sugar monomers depended on fragmentation and elution time in present case. Figure 4.17, shows the presence of glycoprotein at 741 m/z-745 m/z at retention time 3.34. Meanwhile, Na⁺ adducts for glucose appeared as (Glc+Na) ⁺ (182.96 m/z) and rhamnose as (Rha⁺Rha⁺Na) ⁺ (m/z 354.3) at retention time 1.576 and 1.529 min, respectively. The m/z ratio of 212.0 m/z represents glucuronic acid may form adduct as (GlcA+NH4) ⁺ at 1.678 min. The solvent peak appeared at 5.65 min. These results indicated that the bioflocculant consisted of four basic sugar monomers.

Scanning electron microscopy (SEM) imaging showed the morphological surface structure of QZ-7 were illuminated prior to and after flocculation process with kaolin clay particles. As shown in Figure.4.18a, QZ-7 was grey with a condensed crystalline brick-shaped structure. The structure served as an attachment site to which suspended particles and cations could bind (Salehizadeh & Shojaosadati, 2001). Figure 4.18b illustrates how

the bioflocculant aggregated the kaolin particles, which resulted in the formation of larger flocs that were easily sedimented. Therefore, SEM images of QZ-7 and flocculating kaolin particles indicated that bridging could be liable for the flocculation capability of QZ-7. In accordance with the observation, previous research also reported related incidences with some biofloccuants (He *et al.*, 2010; Lian *et al.*, 2008).

The elemental analysis of bioflocculant QZ-7 was on the basis of SEM-EDX testing. Figure 4.19 reveals that the bioflocculant QZ-7 from *B. salmalaya* 139SI had content of carbon (55.74%) oxygen (42.74%), nitrogen (0.54%), phosphate (0.93%), and sulphur (0.06%). Similar results were obtained by Salehizadah *et al.* (2014) where phosphorus and sulphur were present in low quantities.

Thermogravimetric (TGA) determination of purified bioflocculant QZ-7 was tested to elucidate its property compartments at different heat degrees. The results are depicted in Figure 4.20 to assist in recognising its pyrolysis property after exposure to significantly elevated temperatures. Figure 4.20 shows approximately 12.14%, 23.76% in weight loss at 100°C, 200°C respectively, and a 45.28% weight loss at 200 °C –450°C. The pyrolytic process was almost completed at approximately 800 °C. TGA of bioflocculant QZ-7 exhibited thermo-labile and thermo-stable molecule contents, showing combination of sugar and protein substance, as indicated by examination. This stability was likely due to the polysaccharide composition of the main backbone of MBF-15 (Zhong *et al.*, 2018). The first weight decline may be due to the loss of available moisture or the presence of carboxyl and hydroxyl groups in the protein portion related to glycoprotein-like molecules of QZ-7. Analogous observation was organised according to the work of Wang *et al.* (2011) with respect to the bioflocculant gained by mixed consortium of bacteria

5.5 Factors Affecting of Bioflocculant QZ-7 Performance

Some external factors are closely linked to the performance of ionic flocculants, in certain dosage and pH value (Liu et al., 2017). In general, the inner factor of flocculants such as ionic groups exerts more significant impact on the flocculation performance (Yang *et al.*, 2013). As shown in Figure 4.21, bioflocculant OZ-7 was found to be a fairly steady at wider range of pH range 4-7, and over 70 % flocculating activity was observed at this pH value with significant different performance (p < 0.00), as shown in Table 4.13. The highest flocculating activity was achieved at pH 7 with a mean difference of about 92.26 % from the lowest activity measured at pH 2-3 and 11 (p<0.05), as depicted in Table 4.13. Consequently, to the multiple comparison analysis revealed that pH 5 and 6 has the same effect on flocculating activity of QZ-7 appendix F12. Therefore, bioflocculant QZ-7 was considered useful in neutral and acidic conditions, whereas a pH greater than 9 decreased the degrees of flocculation activity. QZ-7 displays diverse electric statuses at varying pH range, and this could be due to the destabilisation of the kaolin particles which inhibited agglomeration and bridging by the bioflocculant (Natarajan, 2015). Similar finding was reported by Liu et al. (2010), for bioflocculant produced by Chryseobacterium daequens W6 which favoured condition at low acidic and low alkali with the same pH range 4-5-8.

The influence of temperature on the flocculation activity of the purified QZ-7 was investigated. As shown in Figure 4.22. The study found the flocculating activity of QZ-7 was quite sustained and stable at different temperatures of kaolin suspension series from 20- 60° C, within this range, above 80% of flocculating efficiency was obtained. The thermal strength of this bioflocculant QZ-7 was attributed to the core backbone of QZ-7, which comprises polysaccharides (Lu *et al.*, 2005). The bioflocculant with polysaccharides-based structure are generally thermo-stable, but those with protein are sensitive to temperature (Salehizadeh & Shojaosadati, 2001; Gong *et al.*, 2008). A similar

study of MBF-15 indicated tremendous thermostability over a wide range of temperatures 1 from 10 to100°C. This stability was presumably due to the polysaccharide composition of the main backbone of MBF-15 (Luo *et al.*, 2014). However, higher temperature may cause degradation of this polysaccharides chains and diminish flocculating activity (Zhang *et al.*, 2007). Also, the treatment analysis depicted in Table 4.14, for the effect of different temperature degrees on flocculating activity revealed that there are significant differences (p<0.00). While, the multiple comparison showed in appendix F13, that are no significant differences between 20 °C and 45°C and between 55°C and 65°C (p>0.05).

Bioflocculant dosage concentration is one of the dynamic parameters deliberated, during investigating of optimum conditions for coagulant-flocculants production in the coagulation flocculation development (Hassan et al., 2009). Figure 4.23, shown the correlation concerning bioflocculant dosage and turbidity reduction in clay suspension. The performance of bioflocculant concentration applied through treatment can lead to a poor bridging phenomenon, thus resulting in low flocculating activities measured, while, excess concentration might induce re-stabilisation of the kaolin particles (Hassan et al. 2009; Gong et al., 2008). Consequently, bioflocculant optimal dosage should be set up to support reduction of cost and achieve better performance in treatment processes. Figure 4.23 showed the flocculating activity increased dramatically as the bioflocculant QZ-7 concentration increased from 0.25 to 2 mg/mL. The highest flocculating efficiency of 93.6% was obtained at 2 mg/mL of bioflocculant QZ-7. Also, QZ-7 show a good flocculating activity of 77.23% at 1 mg/L, 73.2% at 0.5 mg/mL, 84.2% at 4 mg/mL and over 64.93% at 6 mg/mL. However, at very low concentration of QZ-7 destabilisation and aggregation were not sufficiently to achieve high flocculating efficiency. Furthermore, the analysis of variance shown in Table 4.15, that the result obtained is in accordance with the reported findings in the flocculation performance of different concentration of bioflocculant has significant differences (p<0.005).

Cations improved the coagulation-flocculation process of neutralizing and destabilisation the negative charges residue of bioflocculant function groups through making linked that particle bonds to kaolin suspension (Zhang et al., 2013). For instance, Figure 4.24 shows that monovalent and divalent cations, including Al⁺³ stimulated the flocculating activity to a significant extent that matched that of trivalent Fe^{+3} cations. The maximum flocculating activity 92.7 % was detected with Ca⁺², because it is recognised that Ca⁺² could neutralize and stabilize negative charges of the functional groups of bioflocculant which resulted in forming bridges between suspended particles (Yim et al., 2007). Also, followed by Al⁺³ (82.9%), Mn^{+2} (76.5%), K^{+} (70.6%), Mg^{+2} (72.7.4%), Na^{+} (67.7%) and Li^+ (65.6%). These results are analogous to those stated by Manivasagan et al. (2015) described that in the presence of Ca^{+2} and Al^{+3} flocculating activity of bioflocculants, from Bacillus licheniformis and Streptomyces sp. was enhanced. Also, Okaiyeto et al. (2013). Wang et al. (2011), reported that multivalent of cations such as Ca⁺², Mn⁺² and Al⁺³ increased the flocculation activity of bioflocculant produced by Micrococcus sp. and Halomonas sp., xn11 and xn7 and mixed culture of Bacillus sphaeicus F6, with Rhizobium radiobacter F2. A similar finding reported by Rasulov et al. (2017), that Ca^{+2} contributed to the enhanced flocculating activity of the bioflocculant, produced by R. radiobacter SZ4S7S14. Moreover, the bioflocculant produced by aquatic bacteria, Oceanobacillus sp. which was the most stimulated by cations such as calcium chloride and aluminium chloride was also detected (Cosa et al., 2013a). Moreover, bioflocculant produced by Bacillus sp. and Virgibacillus sp. were interactive, increasing the flocculating activity in the influence of Ca^{+2} , Mg^{+2} and Mn^{+2} (Nwodo *et al.*, 2013). Calcium ions were more active and improved the assembly of larger flocs when matched with others. The statistical analysis presented in Table 4.16, indicated that there is a significant difference (p<0.00) between different cations. However, based on multiple

comparison analysis presented in appendix F9, showed that the potassium and magnesium have no significant difference between each other (p>0.05).

The effect of salt concentration on bioflocculation efficiency of bioflocculant QZ-7 is shown in Figure 4.25. Different concentrations of NaCl salt were tested to examine the effect of salinity on bioflocculant QZ-7 flocculation performance. As presented at 1% and 5% concentration of NaCl the bioflocculant QZ-7 maintained over 90 % of flocculating efficiency. While 10% and 15% (w/w), of NaCl in kaolin suspension has a 73.9%, 55.1% of the flocculating efficiency of bioflocculant QZ-7. In this experiment, the flocculation activity of QZ-7 was decreased as salt concentration increased. This might be due to the excessive concentration of Na⁺ ion interfere between the kaolin particles and QZ-7, or the physical properties of QZ-7 have been changed in high concentration of Na⁺ ion which affected the performance of bioflocculant QZ-7 activity. However, bioflocculant QZ-7 was found suitable for treating turbid water with salinity up to 15 % w/w. The statistical analysis of variance of the effect of salinity concentration on flocculating activity is depicted in Table 4.17, revealed that there is significant difference between different concentration of salt 1-30% (w/w) found to be (p < 0.000). But, in accordance to multiple comparison appendix F11, showed that there is no significant difference between concentration 1% and 5% as (p>0.05).

5.6 Application of Bioflocculant QZ-7

In this study the application of bioflocculant QZ-7 in water treatment and industrial wastewater were investigated by using surface water, sugar mill, rubber wastewater and hospital wastewater. Bioflocculant QZ-7 produced by *Bacillus salmalaya* 139SI shown different levels of COD, TSS, TDS, Turbidity and BOD removals, heavy metal adsorption. Differences in the attraction of metals for bioflocculants are due to attractive interaction, charge density and polymers conformation natures with adsorbed ions

(Morillo *et al.*, 2006). The metal biosorption mechanism and kinetics influenced by experimental conditions mainly, pH medium, bioflocculant concentrations and initial heavy metal concentrations (Converti *et al.*, 2006).

5.6.1 Surface Water Treatment by Bioflocculant QZ-7

The main backbone of the bioflocculant QZ-7 was composed of several sugars and proteins as the side chains which provided numerous flocculant binding sites, while the long backbone ensured the formation of large flocs (Shi-Jie et al., 2011). The raw water sample collected from Langat River, in Kajang, Selangor, Malaysia. This river was the source of drinking water for more than one million people. The physical and chemical characteristics of Langat River are shown at Table 4.18. As indicated in Figure 4.26, the turbidity removal of purified QZ-7 and crude QZ-7 was increased gradually as the concentration increased. The turbidity removal of purified QZ-7 was found higher than crude bioflocculant, especially at low flocculant concentration. Purified QZ-7 concentration of 5 mg/L, reduced turbidity of river water from 178.5 NTU to 5.2 NTU and the turbidity was 96.8%. Gong et al., (2008), found that the bioflocculant produced by Serratia ficaria was used to remove some of the impurities from river water was able to reduce the initial turbidity by 84.2%. The crude bioflocculant at 5 mg/L dosage produced water with final turbidity of 35.6 NTU with removal efficiency of 83.3% because crude bioflocculant may contain some impurities which affect the flocculation performance and decreased the turbidity removal. This indicates that lower concentration of purified bioflocculant could be applied in water treatment. In present case occurrence of glucuronic acid in the bioflocculant QZ-7 from B. salmalaya 139SI this in confirmed through mass spectrometric analysis. Carboxyl groups present on the molecular chain of bioflocculants provide more sites for particle attachment and makes bridging between bioflocculant and discrete particles effective (Farooq et al., 2010). Based on the statistical analysis of the effect of pure and crude bioflocculant QZ-7 in Table 4.19 and Table 4.20,

it shows that there was a significant difference (p<0.000) on the different concentration of the two types of QZ-7.

In general, the TSS removal increased as the bioflocculant concentration increased as presented in Figure 4.27. The TSS removal efficiency by purified bioflocculant QZ-7 gradually increased at concentration range of 0.5 mg/L – 7 mg/L. In addition, 5 mg/L of purified bioflocculant produced water with final TSS of 4.53 mg/L (97.7%). However, beyond 3 mg/L of purified QZ-7 (optimum concentration), TSS removal was decreased. Because of excess purified QZ-7 concentration. Similar study by Lian *et al.* (2008), reported that the bioflocculant produced by *B. mucilaginosus* was able to reduce the TSS of wastewater by 93.3%. Also, Salehizadeh *et al.* (2001) mentioned that *Alc. Latus* bioflocculant was used to treat coke wastewater and final TSS of 80 mg/L could be obtained from the 370 mg/L (78% removal) in the presence of Ca⁺². The removal of crude bioflocculant was gradually increased as the bioflocculant concentration increased and the lowest final concentration of TSS 9.7 mg/L was obtained. The result obtained by statistical analysis in Table 4.21 and Table 4.22, found that there was a significant difference (p<0.000) between pure and crude bioflocculants on TSS removal.

Results in Figure 4.28 showed that the COD reduction was increased dramatically as the purified bioflocculant QZ-7 concentration increased from 0.5 mg/L to 5 mg/L. The highest reduction of COD was obtained at 91.2% of optimum bioflocculant concentration at 5 mg/L, while the COD reduction dropped as the purified bioflocculant QZ-7 concentration increased from 5 mg/L to 7 mg/L, which might be the excess concentration of the bioflocculant that raised the residual of COD. Gong *et al.* 2008; Lian *et al.* (2008), showed that the bioflocculant of *Serratia ficaria* reduced the COD of river water by 87.1% and the bioflocculant of *B. mucilaginosus* decreased the COD by 74.6%. The statistical analysis of the effect of pure and crude bioflocculant QZ-7 on COD removal is

presented in Table 4.23 and Table 4.24, revealed that there was a significant difference (p<0.000), between the two types of QZ-7.

5.6.2 Removal of Heavy Metals Aqueous Solution using Bioflocculant QZ-7

The most essential parameter as pH has been recognized that was very active on heavy metal sorption. It is directly linked through hydrogen ions interaction competency with metal ions on the biosorbent surface active sites (Loderio et al., 2006). The spectroscopic analysis (FTIR), displayed that the pure bioflocculant QZ-7 had an assortment of functional groups, include carboxyl, hydroxyl and amino and these groups are virtually involved in binding mechanisms potentials. Furthermore, these functional groups contribute in metal ion binding are depend on the pH values of the aqueous solution (Dobrowolski et al., 2017). Adsorption of heavy metals was high at neutral and alkaline pH values, this is described that at lower pH values, and a higher proton concentration competing on the same anionic sites of the polymer as the divalent cations (Figure 4.29). The proton mass leads to their preferred binding and so divalent cation binding is low (Sahoo et al., 1992). For instance, the increased of the pH towards optimal value, which is contrasted from metal ion to another, also the saturated superficial adsorbed by negative charges, lead to, to increase the efficiency of positive charges to bound and adsorb metal ions (Bayramoglu et al., 2003). While at a higher pH than its optimal importance, hydroxide metals could be made and the adsorption sites on the surface did not bind the adsorbent (Kacar et al., 2000). The stability of pH obtained in the present study could be advantageous in removing other heavy metals and pollutants too from various waste water streams by bioflocculants in required doses.(Pathak et al., 2017). Regarding to the analysis of variance of the effect of different pH value on heavy metal removal efficiency are presented in Table 4.25 to 4.30. It is revealed that the different pH value has a significant difference (p<0.00) on heavy metal removal. Whereas, the multiple

comparison analysis obtained in appendix F20 to F25, indicated that there is no significant difference between pH 3 and 11, pH 5 and 6, and pH 6 and 7 or 7 and 9.

Biofloocculant QZ-7 displayed a good removal efficiency of heavy metals at low bioflocculant concentration as designated by other studies (Das & Santra, 2007). Figure 4.30 showed that an exceed 50% of metals removal were observed at 20 mg/L - 100 mg/Lof bioflocculant QZ-7 concentration. Whereas, the highest metal removal was achieved at 60 mg/L of bioflocculant concentration, as 84% of As, 80.7% of Cu⁺², 78.5% of Zn⁺² and 77.4 Cd⁺² and Pb⁺² 72.43% at 100 mg/L, besides, there is a significance difference (p<0.000) of different QZ-7 concentration as shown in Tables 4.31 to 4.35. Similarly, finding was reported by Gomma. (2012), the highest metal removal efficiency for the bioflocculant produced by P. aeruginosa was obtained at 60 mg/L of bioflocculant concentration i.e Zn⁺² 80.5%, and As 72.96%, and Cd⁺² 79.93% at 40 mg/L. The main backbone of the bioflocculant QZ-7 was poised of several sugars and proteins as the side chains which provided several flocculant binding sites, although the long backbone confirmed the formation of large flocs (Shi-Jie et al., 2011). Supporting the contribution of the functional groups of associated with bioflocculant, it was also manifest from studies on Zn^{+2} sorption on different bacterial strains which indicated presence of acidic, neutral and basic group on the cell surface essential for biosorption (Guinea et al., 2006). The interactions between the exopolysaccharide based bioflocculant, and Ni⁺², Zn⁺² and other metal cations in aqueous solution leads to removal of the metal ions from water. The increased heavy metals removal efficiencies at low concentrations of bioflocculant develop more attractive in the industrial effluent wastewaters treatment. Besides, in accordance with the multiple analysis presented in appendix F26 to F30, revealed that there is no significant difference (p>0.05), between 10 mg/L and 20 mg/L, 20 mg/L and 100 mg/L, 40 mg/L and 80 mg/L and 100 mg/L.
Results of the effect of metals concentration on the removal efficiency presented in Figure 4.31 shows that the heavy metals adsorption exceeded 60%, with the increased of it's an initial concentration of 10 mg/L to 100 mg/L, also there is a significant difference (p<0.00) of the different concentration of heavy metals on bioflocculant QZ-7 efficiency adsorption was reaveled in Tables 4.36- 4.40. Whereas, the multiple comparison analysis in the appendix F30 - F35 showed that there is no significant difference (p>0.05), between 20 mg/L and 40 mg/L, 20 mg/L and 60 mg/L, 40 mg/L and 60 mg/L, 40 mg/L and 80 mg/L, also 80 mg/L and 100 mg/L. For example, indicated the maximum removal efficiency achieved of As 94.3% at 60 mg/L and the highest removal of Cu⁺² (85.2%). Cd⁺² (84.5%), Zn⁺²(84.5) and Pb⁺² (82.76%), at 100 mg/L. Moreover, the presence of carboxylate, hydroxyl, amino and phosphate groups in the bioflocculant displays involvement in metals binding capacity (Chug et al., 2016). In addition, the presence of Sulphur in bioflocculant QZ-7, enhances its binding capacity and makes it a perfect substrate for utilisation in heavy metals sequestration (Kumari et al., 2017). In addition, the improvement of metal adsorption might be due to an increase in electrostatic interactions, steadily lower affinity linking sites for metal ions (Puranik & Pakniker, 1999).

5.6.3 Removal of Heavy Metal from Industrial Wastewater

The treatment of industrial effluents with bioflocculant QZ-7 at different concentration of 20, 40 and 60 mg/L showed effective flocculation with a concomitant reduction in heavy metals. Likewise, the effect of the dosage of adsorbent on the adsorption of heavy metals by bioflocculant QZ-7 is depicted in Figure 4.32. For example, As removal increased from 72.6% to 89.2% with the increased of the amount of adsorbent concentration, while the Zn⁺² removal also increased from 63.5% to 76.3%. Whereas, the Cu⁺² removal slightly increased from 52.3% to 56.5%. There is proportional of adsorbent concentration and removal efficiency of heavy metal due to bioflocculant QZ-7 has many functional groups and sulphur ions which provides high affinity towards many heavy metals (Kumari *et al.*, 2017). Similar results were obtained by Taha *et al.* (2011) that higher adsorption capacity of QZ-7 due to active functional groups such as hydroxyl and carboxylic group. In case of the removal of As, Zn^{+2} , Cd^{+2} , Cu^{+2} and Pb⁺² using QZ-7 the ion exchange might be the dominant mechanism (Feng & Guo,. 2012). Furthermore, the presence of phosphate and carboxyl group on EPS produced by *B. subtilis* facilitates greater binding of Cd⁺² in comparison to EPS derived from *Pseudomonas putida* (Wei *et al.*, 2011). Following the guidelines of MTCC, *B. subtilis* (MTCC 8363) is also capable to remove heavy metals.

5.6.4 Removal of BOD and COD from Industrial Wastewater

The conjugation of COD analysis test with BOD test is a useful toxic sign of circumstances and the incidence of biological resistance (Poddar & Sahu, 2017). The crude bioflocculant produced by B. salmalaya strain 139SI was used to treat an industrial wastewater with initial concentration showed in Appendix (D.4). After the treatment, results shown in Table 4.42, there was a significant removal rate of COD about 93%. Furthermore, crude QZ-7 showed exceptional turbidity removal of 90.7% for sugar mill effluent and 62.2% for rubber effluent. Besides, crude QZ-7 was capable to reduce TSS in sugar mill effluent and rubber effluent to 89.69%, 61.5%, respectively. While, QZ-7 revealed TDS removal by 88.2% of sugar mill effluent, 61.5% of rubber effluent and 52.1% of ethanol mill effluent. There were many studies reported on the reduction of COD and TSS by using bioflocculants. For example, bioflocculant from Azotobacter indicus was able to reduce BOD, COD and TSS in different wastewater samples in the range of 38%–80%, 37%–79% and 41%–68%, respectively, (Patil et al., 2011). Moreover, bioflocculant from Klebsiella sp. showed tremendous turbidity removal of 99% and COD of 68.4% at low concentration of 40 mg/L (Feng et al., 2009). The B. salmalaya strain 139SI used to treat an industrial wastewater of initial BOD concentration

of 4018 mg/L. After the treatment, the final BOD concentration was found to be 302 mg/L. Therefore, the percentage of BOD removal estimated to be 92.4%. Hence, *B. salmalaya* strain 139SI can be used for BOD removal from wastewater.

The removal of pharmaceuticals during wastewater treatment was estimated from concentration data in influent wastewater and effluent wastewater. Considering that pharmaceuticals have rather different physicochemical characteristics, their removal during treatment is expected to be diverse (Gracia-Lor et al., 2012) In the literature, the removal efficiency is generally computed as the percentage of reduction between the dissolved aqueous phase concentration of the contaminant in the influent and the dissolved aqueous phase concentration of the contaminant in the effluent. As far as hospital effluent wastewater was concerned, bioflocculant QZ-7 was effectively used to treat the hospital effluent sample, with removal of 88.5% of COD, 86.7% of BOD and 92.5% of TSS. Meanwhile, the primary removal of selected pharmaceuticals as shown in Table 4.42, from hospital wastewater was determined at the laboratory. The result obtained showed that the bioflocculant QZ-7 had a significant removal efficiency of simvastin 92.45 % and salbutmon 88.69 %, while acetaminophen and caffeine was found to be 69.94 %, 66.52 %, respectively. However, there was no significant removal of nifedipine, but for the perindopril, mefenamic acid diclofence was detected at 25.49 %, 13.64 %, 4.04 %, respectively.

5.7 Bioflocculant QZ-7 Mediated Synthesis of AgNPs, Characterisation, and Antibacterial Activity.

The biopolymers-based synthesis of AgNPs has been carried out for a decade and a variety of polymers viz., algal, bacterial, fungal and plant-based polymers were exploited greatly. The bacterial bioflocculant being conjugates of carbohydrates and proteins showed the best to bring out the reduction of AgNO₃ to particulate Ag-capped with

polymer, providing stability by preventing particles to avoid aggregation. The synthesis of metal nanoparticles has been extensively studied, among which AgNPs were intensively focused due to its application in the field of biomedical optical and solar. The antimicrobial activity, especially antibacterial effect was detected as the potent quality of silver for centuries.

5.7.1 Synthesis of AgNPs

The nutrient broth supplemented with glucose showed synthesis of AgNPs which was primarily distinguished by visual colour change (Figure 4.33) in a bottle on the 6th day of cultivation of *B. salmalaya* 139SI. While, the control bottle containing nutrient broth supplemented with and without AgNO₃ did not display change colour. The constant colour was changed from yellow to dark brown in testing bottle revealed the presence of Ag particles. The reduction of silver nitrate was carried out through bioflocculant QZ-7 and the mechanism is called excitation of surface plasmon response (Zaki *et al.*, 2014).

In the cell-free supernatant having bioflocculant QZ-7 the colour change was detected after 3rd day of the incubation. The bioflocculant QZ-7 reduced AgNO₃ to produce AgNPs (Figure 4.34). A sharp peak at 402 nm achieved was designated to AgNPs. Report of Zaki *et al.* (2014) showed that particular bioflocculant producing media provide a better effect on synthesis as the amount of extracellular substances increase in the particular medium. Therefore, the concentration of bioflocculant in free supernatant might be higher as compared to nutrient broth leading to the faster synthesis of AgNPs.

Ten mL of purified bioflocculant QZ-7 (1%) was added to 100 mL of 3 mM AgNO₃ and kept overnight on a shaker incubator at 37°C, at 120 rpm under dark condition. The purified bioflocculant QZ-7 showed synthesis after 1st day as compared to cell-free supernatant which was acquired on 3rd day (Figure 4.35). Bioflocculant QZ-7 comprised of carbohydrate and protein, involve carboxyl and hydroxyl group supported in the

reduction and stabilisation of Ag^+ (Sharma *et al.*, 2009). Ag^+ ions oxidise the hydroxyl groups to carbonyl groups, reducing itself to elemental silver. The carboxylate and amino groups generated a capping that keeps the particles separated. The aggregation is a major problem of chemically synthesised AgNPs and requires a stabilising chemical. The bioflocculant not only synthesises nanoparticles but also provides stability to the particles (Sathiyanarayan *et al.*, 2013; Zaki *et al.*, 2014). Consequently QZ-7 performed as reducing and stabilising agent which is an advantage of biological synthesis of AgNPs over chemical synthesis. Sathiyanarayan *et al.* (2013) found that the bioflocculant produced by marine *B. subtilis* could produce AgNPs with 5% bioflocculant concentration with 48 h incubation time. Furthermore, bioflocculant has an advantage over other microbial techniques, whereby the separation of nanoparticles from the bacterial cell is difficult. The separation of bacterial cells from aqueous solution by using series processes; centrifugation accumulates Ag^+ particles together with bacterial cells, whereas some of the bacteria produce intracellular nanoparticles that need disruption of the cell (Singh *et al.*, 2015).

5.7.2 Characterisation of AgNPs.

Spectrophotometric analysis is the simplest, unique as well as the most sensitive technique to determine the presence of AgNPs and their size distribution as shown in Figure 4.36. The spectrum was determined every 24 h until the peak was detected at 412 nm and absorbance was 2.7 called excited surface plasmon response. This result indicated the presence of 10-40 nm sized AgNPs which was well described by Kumar *et al.* (2012). According to Shrivastava *et al.* (2007) defined that peak at 300 nm is labelled to AgNO₃ solution while peak around 400 nm is of AgNPs and range of peak from 400-450 nm exhibit particle size.

The reducing/capping protein responsible for the formation of AgNPs were recognised by FTIR spectrum. Therefore, the FTIR spectrum of silver nanoparticles between the wave numbers 400 cm⁻¹ –4000 cm⁻¹ proposed that the active bioflocculant QZ-7 was found to be responsible for the reduction of Ag+ ions into metallic silver nanoparticles, which were produced by bond stretching. The specific bond contribution in this phenomenon included O-H at 3421, C-H at 2921.48, =C-H 2851.09, C=C 2321.94, C=C at 2163.83, C-C at 1578.26, C-N at 999.07 and C-H at 478.31cm⁻¹ and 463.3cm⁻¹ as shown in (Figure 4.37). Moreover, the results of the present study clearly suggested that protein or amino acids were responsible for the stabilisation of the silver nanoparticles synthesised by using the purified bioflocculant QZ-7. The results achieved from the present study were in agreement with the findings described by Viswanathan *et al.* (2016).

The FE-SEM and EDAX were used to study the morphology, size and elemental composition of biosynthesized AgNPs. The elements present in the AgNPs were confirmed by EDAX spectrum spot profile made from the densely populated nanoparticles region of the slide surface. The synthesized nanoparticles at micro (10^{-6}) and nano (10^{-9}) can be identified by FE-SEM. The FE-SEM clearly shows that the particles is spherical and the size is found to be about 25 nm -87 nm for biosynthesised AgNPs as shown in Figure 4.38a. The elemental composition of AgNPs is examined by EDAX spectroscopy and is a chemical analysis method combined with FE-SEM Figure 4.38b. (Lawrence & Prakash, 2019). In EDAX, the optical absorption peak at 2kev is observed and strong signal for Ag was detected in which designates that the AgNPs were successfully made by the bioflocculant QZ-7. This shows that metallic silver nanocrystallites which are due to surface plasmon resonance. Sathiyanarayanan *et al.* (2013), reported 60 nm average sized particles synthesized by bioflocculant produced by *B. subtilis.* Recent literature also reports a similar synthesis procedure for AgNPs concerning the treatment of silver nitrate solution with free- supernatant from culture of

K. pneumoniae, in which the particles range in size from 28.2 to 122 nm and exhibited an average size of 52.5 nm (Viswanathan *et al.* 2016). In addition, the manifest of AgNPs synthesized by all bio-groups of *Morganella* spp. were found quasi-spherical in shape, and ranged between 10-50 nm in diameter (Parikh *et al.* 2011). Wei *et al* (2012) stated 4.8-23.7 nm sized AgNPs while Zaki *et al.* (2014), found 6-72 nm sized AgNPs synthesized by bioflocculant produced from *B. mojavensis*.

The XRD configuration shown metallic nature of AgNPs and in this technique, electromagnetic radiation (X-ray) with typical photon energy penetrates into the core of the material to distinguish the nature of bulk structure (Zaki et al., 2014). As illustrated in Figure 4.39 the XRD spectrum of AgNPs showed 4 intense peaks in rang of 20-80° at 20 value; particularly at 38.2667°, 44.4964°, 64.5751° and 77.5673° with conforming degree of 111, 200, 220 and 311 respectively, and this result was found to be in agreement with database of the joint committee on Powder Diffraction Standards (JCPDS File no.04-0783). Which are particular to crystalline silver indexed to face centered cubic structure of silver. The AgNPs mean crystallite size was measured by the Debye-Scherrer formula $D = K\lambda/\beta \cos\theta$ (K – Scherrer constant, related to crystalline shape, λ –Radiation wavelength, β – Full width at half maximum of the diffraction peak, and θ – Bragg's angle). The average crystallite size of AgNPs was 40 nm (Monshi et al., 2012) Similar finding was reported by Manivasagan et al. (2015) that the bioflocculant synthesized by Streptomyces sp. is capable of producing AgNPs with similar XRD pattern. Likewise, diffraction patterns were gained with silver nanoparticles synthesized by cell free supernatant of B. cereus (Ganesh & Gunasekaran. 2009), B. Bacillus casei (Kalishwaral et al., 2010).

5.7.3 Antibacterial Activity of AgNPs Synthesised by Purified Bioflocculant QZ-7

Silver is known for its antimicrobial property since ancient times; hence, used widely in the biomedical field. Results are presented in Table 4.43 and Figure 4.40. E. coli ATCC35401. Pseudomonase aeruginosa, Salmonella entertidis ATCCBBA-711 and staphylococcus aureus ATCC 2592 showed almost consistent in the zone of inhibition. Meanwhile, E. coli ATCC35401 and Salmonella enteritidis ATCCBAA-711 showed a slight increase with an increase in the concentration of AgNPs. These values are compared with amoxicillin (standard). These results were supported by antimicrobial activity in liquid media. The bactericidal mechanism of AgNPs was not specific as antibiotics, but rather it has a multiplicity of mechanisms. Meanwhile, smaller particles that possess larger surface area accessible for interaction will contribute more bactericidal effects than larger particles. The possible mechanisms are cell disruption, protein denaturation and subsequent loss of enzyme activity, denaturation of nucleic acid and implication of oxidative stress by reactive oxygen species. Furthermore, the study showed that by using the scanning tunnelling electron microscopy (STEM), and the X-ray energy dispersive spectrometer (EDS) revealed AgNPs can penetrate inside the bacterial cell and cause damage by interacting with phosphorus and sulphur-containing compounds such as DNA (Panacek et al., 2006).

CHAPTER 6: CONCLUSION AND RECOMMENDATIONS

6.1 Summary

In this thesis, the main experiments were completed and summarised as in Chapter 4, a novel bioflocculant named QZ-7 produced by *B. salmalaya* strain 139SI was investigated in this study. The flocculating activity of bioflocculant QZ-7 presented in the selected strain was found to be 83.3%. The effects of carbon source, nitrogen source, metal ions, initial pH, inoculum size, temperature, shaking speeds, static condition and time course on bioflocculant production were investigated. The maximum production was obtained when mixture of carbon source was sucrose and glucose and urea plus yeast extract as nitrogen source, the optimum initial pH 7 inoculum size 5% (v/v), temperature 35.5°C and shaking speed 160 rpm. The maximum flocculating activity of the optimised strain *B. salmalaya* 139SI was found to be 92.6%.

Chemical analysis revealed that the pure bioflocculant QZ-7 consisted of 79.08% carbohydrates and 15.68% proteins. About 2.72 g of purified bioflocculant with average molecular weight of 5.13×10^5 Da was recovered from 1.0 L of fermentation broth. Fourier-transform infrared spectroscopy (FTIR) analysis showed the presence of carboxyl (COO-), hydroxyl (-OH), and amino (-NH₃) groups; which were typically of polysaccharides; and proteins. The NMR spectroscopy analysis confirmed the result of FTIR, through the presence of functional groups of the QZ-7. Scanning electron microscopy (SEM) analysis showed that QZ-7 exhibited a clear crystalline brick-shaped structure. LC-MS analysis confirmed that QZ-7 was a glycoprotein compound detected at 741m/z-745m/z. Moreover, the presence of glucose at 182.96 m/z, rhamnose at 354.3 m/z, and glucuronic acid at 212 m/z. The elemental analysis of purified QZ-7 revealed that the weight fraction of elements C, H, N, O and S were 55.75%, 0.54%, 42.74%, 0.93% and 0.06%, respectively. The TGA of bioflocculant was found to be heat-stable and its activity was only decreased by about 12.14% and 23.76% in weight at 100°Cand

200°C , respectively, with 45.28% loss of weight at 300°C. Some physical properties of QZ-7 bioflocculant such as pH stability and thermo-stability were investigated as well. Results showed bioflocculant QZ-7 exhibited wide pH stability ranging from 4 - 7, with a flocculation activity above 70% at pH 7. Likewise, to study the thermo-stability of QZ-7, the QZ-7 was kept for 1h at different temperatures of 20 °C – 100°C. Result revealed that QZ-7 was thermally stable and reserved more than 80% of its flocculating efficiency after being heated at 80 °C for 30 min.

In this study, the effect of different cation on flocculation efficiency was established by KCl, NaCl, LiCl (monovalent), MnCl₂, MgCl₂ (divalent), and AlCl₃, FeCl₃ (trivalent) and flocculating efficiency was measured. Over--all, most of the cations enhanced the flocculating efficiency of QZ-7, the highest flocculating activity (92.9%) was achieved for Ca⁺² at pH 7, followed by Al⁺³ (83.3%), Mn⁺² 75.6 K⁺² 71.6, Mg⁺² 71.4, 70%, Na⁺ 67.3 and Li⁺ 64%.

The treatment of river water was by 5 mg/L of bioflocculant QZ-7. Results showed that the final turbidity, total suspended solids and COD were found to be reduced by 96.8%, 97.7% and 91.2%, respectively. These results were the lowest in comparison with the water treated by crude QZ-7 at 5 mg/L.

Bioflocculant QZ-7 could successfully flocculate artificial and real wastewater, with a concomitant reduction in COD, TDS, turbidity, colour and heavy metals from wastewater of rubber, sugar mill, and ethanol industries. In addition, for the removal of heavy metals from wastewater, the results revealed that the bioflocculant QZ-7 was capable to remove heavy metals. For example, the maximum adsorption of As (89.3 %), and Zn^{+2} (81.3%) were detected at pH 7. Meanwhile, the removal of Pb⁺² (77.9%), Cu⁺² (76.1%) and Cd⁺² (68.7%) was achieved at pH 9. Whereas, the optimal dosage of bioflocculant of 60 mg/Lfound that the removal efficiency of As, Cu⁺², Zn⁺² and Cd⁺² were detected at 84.4%,

80.7%, 78.5% and 77.4%, respectively. Furthermore, results for the removal of heavy metals from industrial wastewater revealed that the bioflocculant QZ-7 was capable of removing the heavy metals. For example, the maximum adsorption of As (89.8 %), and Zn^{+2} (77.4 %), and Cu^{+2} (58.4%).

The pharmaceuticals from hospital effluent wastewater were also treated with QZ-7 with a reduction in COD and suspended particles. Meanwhile, the removal efficiency of drug compounds, such as Simvastatin 92.45%, Salbutamol 88.69%, Acetaminophen 69.94, Caffeine 66.52% Perindopril 25.49%, and Mefenamic acid 13.65%.

Apart from flocculation, *B. salmalaya* 139SI, supernatant and pure bioflocculant QZ could synthesise AgNPs and were coated with biopolymer which created a repulsive force between the particles to keep them apart and escape aggregation. The antibacterial activity of AgNPs was obtained against different strain of bacteria as application of AgNPs.

The overall results analysed by the analysis of variance revealed that there are significant differences (p<0.00).

6.2 Recommendation for Future Research

This study creates a basal platform on fermentation technologies for the production of bio-products of environmental significance, particularly in the production of bioflocculant for the treatment of wastewater. Therefore, it is recommended for future research to further exploit the production through submerged fermentation (SmF) by scaling– up the process on a pilot-scale and industrial level to increase yield for possible mass production and potential commercialisation while maintaining the quality and performance. Moreover, the sugar processing industries wastewater must be studied as cheap substrate as the carbohydrates source for bioflocculant production by *B. salmalaya* 139SI.

REFERENCES

- Abdel-Aziz, S., Hoda, A., Foukia, E., Nayera, A., & Abdelwahed, M. (2011). Extracellular metabolites produced by a novel strain, *Bacillus alvei* NRC-14: 3. Synthesis of a bioflocculant that has chitosan-like structure. *Life Science Journal*, 8(4), 883-890.
- Abd-El-Salam, A., Abd-El-Haleem, D., Youssef, A., Zaki, S., Abu-Elreesh, G., & El-Assar, A. (2017). Isolation, characterization, optimization, immobilization and batch fermentation of bioflocculant produced by *Bacillus aryabhattai* strain PSK. *Journal of Genetic Engineering and Biotechnology*, 15, 335-344.
- Abdul-Razack, S., Velayutham, V., & Thangavelu, T. (2014). Medium optimization and in vitro antioxidant activity of exopolysaccharide produced by *Bacillus subtilis*. *Korean Journal of Chemical Engineering*, 31, 296-303.
- Aboulhassan, M., Souabi, S., Aljuboori, A., Yoshimitsu, U., Noridah, B., & Suzana, Y. (2014). Production of a bioflocculant from *Aspergillus niger* using palm oil mill effluent as carbon source. *Bioresource Technology*, 171, 66-70.
- Aboulhassan, M., Souabi, S., Yaacoubi, A., & Baudu, M. (2006). Improvement of paint effluents coagulation using natural and synthetic coagulant aids. *Journal Hazardours Materials B*, 138, 40-45.
- Adebayo-Tayo, B., & Adebami, G. (2014). Production and characterization of bioflocculant produced by *Bacillus clausii* NB2. *Innovative Romanian Food Biotechnology*, 14, 13-25.
- Aguilera, M., Quesada, T., Aguila, G., Morillo, A., Rivadeneyra, A., Cormenzana, R., ... Sanchez, M. (2008). Charaterization of *paenibacillus jamilae* strains that produce exopolysaccharides during growth on and detoxification of olive mill wastewater. *Bioresource Technology*, 99, 5640–5644.
- Ahmad, A.L., Sumathi, S., & Hameed, B.H. (2005). Residue oil and suspended solid removal using natural adsorbents chitosan, bentonite and activated carbon: a comparative study. *Chemical Engineering Journal*, 108, 179–185.
- Ahmad, A.L., Yasin, N.H., Derek, J.C., & Lim, J.K. (2011). Optimization of *microalgae* coagulation process using chitosan. *Chemical Engineering Journal*, 173, 879–882.
- Al-Asheh, S., Banat, F., & Al-Hamed, N. (2003). Adsorption of pollutants from aqueous solutions using activated and non-activated oak shells: Parametric and fractional factorial design study. Part I. Removal of Copper. Adsorption Science and Technology, 21(2), 177-188.
- Aljuboori, A., Idris, A., Al-joubory, H., Yoshimitsu, U., & Abubakar, B. (2015). Flocculation behavior and mechanism of bioflocculant produced by Aspergillus flavus. Journal of Environmental Management, 150, 466-471.

- Aljuboori, A., Idris, A., Al-joubory, H., & Abubakar, B. (2015). Flocculation behavior and mechanism of bioflocculant produced by Aspergillus flavus. Journal of Environmental Management, 135, 466-471.
- Aljuboori, A., Yoshimitsu, U., Noridah, B., & Suzana Y. (2014). Production of a bioflocculant from *Aspergillus niger* using palm oil mill effluent as carbon source. *Bioresource Technology*, 171, 66-70.
- Aljuboori, A., Rajab A., Norhafizah, A., & Rosfarizan, M. (2013). Production and characterization of a bioflocculant produced by *Aspergillus flavus*. *Bioresource Technology*, 127, 489-493.
- Altumn, M., Sahinkaya E., Durukan, I., Bektas, S., & komnitsas. (2014). Arsenic removal in a sulfidogenic fixed-bed column bioreactor. *Journal of Hazard Material*, 269, 31-37.
- American Public Health Association (APHA). Standard Methods for the Examination of Water and Wastewater APHA, 2005, (21th ed). Washington, DC.
- Arezoo, A., Azari, A., Rezakazemi, M., & Ansarpou, M. (2017). Removal of Heavy Metals from Industrial Wastewaters: A Review. *Chemical and Biochemical Engineering Reviews*, 4(1), 37-59.
- Arezoo, C. (2002). The potential of aluminium in Alzheimer's disease. *Nephrology Dialysis Transplantation*, 17(2), 17-20.
- Ates, O., Arga, K., & Oner, E. (2013). The stimulatory effect of mannitol on levan biosynthesis: Lessons from metabolic systems analysis of *Halomonas smyrnensis* AAD6T. *Biotechnology Progress*, 29, 1386-1397.
- Ayat, E., Desouky, A., Youssef, A., Zaki, S., Abu-Elreesh, G., & El-Assar, S., (2017). Isolation, characterization, optimization, immobilization and batch fermentation of bioflocculant produced by *Bacillus aryabhattai* strain PSK1. *Journal of Genetic Engineering and Biotechnology*, 15(2), 335-344.
- Aydin, H., Bulut, Y., & Yerlikaya, C. (2008). Removal of copper (II) from aqueous solution by adsorption onto low-cost adsorbents. *Journal of Environmental Management*, 87(1), 37-45.
- Azimi, A., Azari, A., Rezakazemi, M., & Ansarpour, M. (2017). Removal of heavy metals from industrial wastewaters: A review. *Chemical and Biochemical Engineering Reviews*, 4(1), 37-59.
- Banks, W., Niehoff, M., Drago, D., & Zatta, P. (2006). Aluminum complexing enhances amyloid protein penetration of blood-brain barrier. *Brain Research*, 1116, 215-221.
- Batta, N., Subudhi, S., Lal, B., & Devi, A. (2013). Isolation of a lead tolerant novel bacterial species, *Achromobacter* sp. TL-3: Assessment of bioflocculant activity. *Indian Journal of Experimental Biology*, *51*, 1004-1011.

- Bayramoglu, G., Baktas, S., & Arica, M. (2003). Biosorption of heavy metal ions on immobilized white-rot fungus *Trametes versicolor*. Journal of Hazardous Materials, 101, 285–300.
- Bayoumi, R.A., Haroun, B.M., Ghazal, E.A. & Maher, Y.A. (2010). Structural analysis and characteristics of biosurfactant produced by some crude oil utilizing bacterial strains. *Australian Journal of Basic and Applied Sciences*, *4*, 3484-3498.
- Bazrafshan, E., Mohammadi, L., Ansari-Moghaddam, A., & Mahvi, A.H. (2015). Heavy metals removal from aqueous environments by electrocoagulation process– a systematic review. *Journal of Environmental Health Science and Engineering*, 13, Article # 74.
- Beltran-Heredia, J., & Sanchez-Martin, J. (2009). Removal of sodium lauryl sulphate by coagulant/flocculation with *Moringa oleifera* seed extract. *Journal of Hazardous Materials*, 164, 713-719.
- Bhatnagara, A., & Sillanpaa, M. (2010). Utilization of agro-industrial and municipal waste materials as potential adsorbents for water treatment: A review. *Chemical Engineering Journal*, 157, 277-296.
- Bhunia, B., Mukhopadhy, D., Goswami, S., Mandal, T., & Dey, A. (2012). Improved production, characterization and flocculation of poly (γ)-glutamic acid produced from *Bacillus subtilis*. *Journal of Biochemical Technology*, *3*, 389-394.
- Bolto, B. (1995). Soluble polymer in water purification. *Progress in Polymers Science Journal*, 20, 981-1041.
- Bordet, J. (1899). Annales Institute Pasteur, Paris, (13), 225-250.
- Bound, J., & Voulvoulis, N. (2005). Household disposal of pharmaceuticals as a pathway for aquatic contamination in the United Kingdom. *Environmental Health Perspectives*, *113*(12), 1705-1711.
- Bradford, M. M. (1976) A rapid and sensitive method for the quantitation of microgram quantities of protein utilizing the principle of protein-dye binding. *Analytical Biochemistry*, 72, 248–254.
- Butterfield, C. (1935). Studies of sewage purification. II. A *zoogloea* forming bacterium isolated from activated sludge. U.S. Public Health Reports, 50, 671-684.
- Cacicedo, M., Castro, M., Servetas, I., Bosnea, L., Boura, K., & Tsafrakidou, P. (2016). Progress in bacterial cellulose matrices for biotechnological applications. *Bioresource Technology*, 213, 172-180.
- Cai, L., Xu, Z., Qi, J., Feng, Z., & Xiang, T. (2015). Assessment of exposure to heavy metals and health risks among residents near Tonglushan mine in Hubei, *China. Chemosphere*, 127, 127-135.

- Castiglioni, S., Bagnati, R., Fanelli, R., Pomati, F., Calamari, D., & Zuccatto, E. (2006). Removal of pharmaceuticals in sewage treatment plants in Italy. *Environmental Science and Technology*. 40, 357-363.
- Chaplin, M., & Kennedy, J. (1994). *Carbohydrate analysis* (2nd ed.). Oxford, UK: Oxford University Press.
- Chen, L., Zhou, S., Shi, Y., Wang, C., Li, B., Li Y.,... Wu, S. (2018). Heavy metals in food crops, soil, and water in the Lihe River Watershed of the Taihu Region and their potential health risks when ingested. *The Science of the Total Environment*, 615, 141-149.
- Chen, Y., Fu, Y., & Li, X. (2003). Characteristics and treatment of brewery wastewater. *Journal Pollution Control Technology*, 16, 146-149.
- Chen, Z., Li Z., Liu, P., Liu, Y., Wang, Y., Li, Q., ... He, N. (2017). Characterization of a novel bioflocculant from a marine bacterium and its application in dye wastewater treatment. *BMC Biotechnology*, *17*, Article # 84.
- Chouchane, H., Mahjoubi, M., Ettoumi, B., Neifar, M., & Cherif, A. (2017). A novel thermally stable heteropolysaccharide based bioflocculant from hydrocarbonoclastic strain *Kocuria rosea* BU22S and its application in dye removal. *Environmental Technology*, 39(7), 859-872.
- Chug, R., Gour, V., Mathur, S., & Kothari, S. (2016). Optimization of extracellular polymeric substances production using *Azotobacter beijreinckii* and *Bacillus* subtilis and its application in chromium (VI) removal. *Bioresource Technology*, 214, 604-608.
- Cobbing, J. (2008). Institutional linkages and acid mine drainage: The case of the Western Basin in South Africa. In Patrick M.J, Rascher J., and Turton, A.R., (Eds) Reflections on Water in South Africa. *International Journal Water Research Development*, 24, 451-462.
- Converti, A., Lodi, A., Solisio, C., Soletto, D., Del Borghi, M., & Carvalho, J. (2006). *Spirulina platensis* biomass as adsorbent for copper removal. *Ciencia Technology Alimentaria*, 5, 85–88.
- Cosa, S., Ugbenyen, M. A., Mabinya, L. V., Rumbold, K., Olaniran, A. O., Aghdasi, F.,... Okoh, A. (2013a). A marine bacterium, *Oceanobacillus* sp. Pinky, isolated from Algoa Bay sediment produces a thermostable glycoprotein flocculant. *African Journal Biotechnology*, 12, 4135-4146.
- Cosa, S., Ugbenyen, M., Mabinya, L., & Okoh, A. (2013b). Characterization of a thermostable polysaccharide bioflocculant produced by *Virgibacillus* species isolated from Algoa bay. *African Journal Microbiology Research*, 7, 2925-2938.
- Cosa, S., & Okoh, A. (2014). Bioflocculant production by a consortium of two bacterial species and its potential application in industrial wastewater and river water treatment. *Polish Journal Environmental Studies*, 23, 689-696.

- Cosa, S., Mabinya, L., Olaniran, A., & Okoh, A. (2012). Production and characterization of bioflocculant produced by *Halobacillus* sp. Mvuyo isolated from bottom sediment of Algoa Bay. *Environmental Technology*, 33, 967-973.
- Cosa, S., Mabinya, L., Olaniran, A., Okoh, O., & Okoh, A. (2011). Bioflocculant production by *Virgibacillus* sp. Rob isolated from the bottom sediment of Algoa Bay in the Eastern Cape, South African. *Molecules*, *16*, 2431-2442.
- Cosa, S., Ugbenyen, M., Mabinya, L., & Okoh, A. (2013a). Characterization of a thermostable polysaccharide bioflocculant produced by *Virgibacillus* species isolated from Algoa Bay. *African Journal Microbiology Research*, 7(23), 2925-2938.
- Cristina, Q., Vitor, B., Silva, D., Bruna, S., Hugo, F., & Teresa, T. (2011). Optimization of production of extracellular polymeric substances by *Arthrobacter viscosus* and their interaction with a 13X zeolite for the biosorption of Cr (VI). *Environmental Technology*, 32, 1541-1549.
- Czemierska, M., Szczes, AL., Hołysz, L., Wiater, A., & Wilkołazka, A. (2017). Characterization of exopolymer R-202 isolated from *Rhodococcus rhodochrous* and its flocculating properties. *Europen Polymers Journal*, 88, 21-33.
- Das, S., & Santra, S.C. (2007). Microbial interactions with heavy metals and their applications in bioremediation of wastewater. *New Frontiers of Environmental Biotechnology and Application*, *3*, 1–10.
- Defang, Z., Juanjuan, W., & John, F. (2008). Application of a chitosan flocculant to water treatment. *Carbohydrate Polymers*, *71*, 135-139.
- Delbarre-Ladrat, C., Corinne, S., Lou, L., Agata, Z., & Colliec-Jouault, S. (2014). Exopolysaccharides produced by marine bacteria and their applications as glycosaminoglycan-like molecules. *Fontiers in Chemistry*, 2, Article # 85.
- Deng, S., Yu, G., & Ting, Y. (2005). Production of a bioflocculant by Aspergillus parasiticus and its application in dye removal. Colloids and Surfaces Biointerfaces, 44, 179-186.
- Deng, S., Bai, R., Hu, X., & Luo, Q. (2003). Characteristics of a bioflocculant produced by *Bacillus mucilaginosus* and its use in starch wastewater treatment. *Applied Microbiology and Biotechnology*, 60(5), 588-593.
- Dermlim, W., Prasertan, P., & Doelle, H., (1999). Screening and characterization of bioflocculant produced by isolated *Klebsiella sp. Applied Microbiology and Biotechnology*, 52, 698–703.
- Desouky, A., Haleem, A., Roda, F., Thourya, M., Sidra, M., & Fatima, H. (2008). Isolation and characterization of extracellular bioflocculants produced by bacteria isolated from Quatari Ecosystems. *Polish Journal of Microbiology*, 57(3), 231-239.

- Dobrowolski, R., Szczes, AL., Czemierska, M., & Wikolazka, A. (2017). Studies of cadmium (II), lead (II), nickel (II), cobalt (II) and chromium (VI) sorption on extracellular polymeric substances produced by *Rhodococcusopacus* and *Rhodococcus rhodochrous. Bioresource Technology*, 225, 113-120.
- Du, R., Qiao, X., Zhao, F., Song, Q., Zhou, Q., Wang, Y.,... Zhou, Z. (2018). Purification, characterization and antioxidant activity of dextran produced by *Leuconostoc pseudomesenteroides* from homemade wine. *Carbohudrate Polymers*, 198, 529-536.
- Elkady, M., Soha, F., Sahar, Z., Gadallah, A., & Desouky, A. (2011). *Bacillus mojavensis* strain 32A, a biofloculant-produing bacterium isolated from an Egyptian salt production pond. *Bioresource Technology*, *102*(17), 8143-8151.
- Fang, Y., Ahmed, S., Liu, S., Wang, S., Lu, M., & Jiao, Y. (2013). Optimization of antioxidant exopolysaccharides production by *Bacillus licheniformis* in solid state fermentation. *Carbohydrate Polymers*, 98, 1377-1382.
- Farooq, U., Kozinski, J. A., Khan, M. A., & Athar, M. (2010). Biosorption of heavy metal ions using wheat based biosorbents-A review of the recent literature. *Bioresource Technology*, 101, 5043–505.
- Feng, N., & Guo, X. (2012). Characterization of adsorptive capacity and mechanisms on adsorption of copper, lead and zinc by modified orange peel. *Transaction of Nonferrous Metals Society of China*, 22, 1224–1231.
- Feng, P., Wang, S., & Zhou, L. (2009). Isolation, identification of a bioflocculantproducing bacterium and its application in drinking water purification. *Acta Scientiae Circumstanti*, 29, 1666-1671.
- Feng, D., & Xu, S. (2008). Characterization of bioflocculant MBF3-3 produced by an isolated *Bacillus* sp. World Journal of Microbiology and Biotechnology, 24(9), 1627-1632.
- Fujita, M., Ike, M., Tachibana, S., Kitada, G., Kim, S., & Inoue, Z. (2000). Characterization of a bioflocculant produced by *Citrobacter* sp. TKF04 from acetic and propionic acids. *Journal of Bioscience Bioengineering*, 89(1), 40-46.
- Gao, J., Bao, H., Xin, M., Liu, Y., Li, Q., & Zhang, Y. (2006). Characterization of a bioflocculant from a newly isolated *Vagococcus* sp. W31. *Journal of Zhejiang University, Science. B*, 7(3), 186-192.
- Gao, B., Wang, Y., Yue, Q., Wei, J., & Li, Q. (2007). Color removal from simulated dye water and actual textile wastewater using a composite coagulant prepared by polyferricchloride and poly-dimethyldiallylammonium chloride. *Separation and Purification Technology*, 54, 157-163.
- Gao, B., Wang, Y., Yue, Q., Wei, J., & Li, Q. (2008). The size and coagulation behavior of a novel composite inorganic-organic coagulant. *Separation and Purification Technology*, 62, 546-552.

- Gao, Q., Zhu, X., Mu, J., Zhang, Y., & Dong, X. (2009). Using *Ruditapes philippinarum* conglutination mud to produce bioflocculant and its applications in wastewater treatment. *Bioresource Technology*, 100, 4996-5001.
- Ganesh, B.M.M., & Gunasekaran, P. (2009). Production and structural characterization of crystalline silver nanoparticles from *Bacillus cereus* isolate. *Colloids and surfaces B: Biointerfaces*, 74(1), 191-195.
- Geddie, L., & Sutherland, W. (1993). Uptake metals by bacterial polysaccharides. *Journal Applied Bacteriology*, 74(40), 467-472.
- Girma, K., Lorenz, V., Blaurock, S., & Edelmann, F. (2005). Coordination chemistry of acrylamide. *Coordination Chemistry Reviews*, 249, 1283-1293.
- Giriet, S., Harshiny, M., Sen, S., Sukumaran, V. & Park, S. (2015). Production and characterization of a thermostable bioflocculant from *Bacillus subtilis* F9, isolated from wastewater sludge. *Ecotoxicology and Environmental Safety*, 121, 45-50.
- Gomma, E. Z., (2012). Production and characteristics of a heavy metals removing bioflocculant produced by *Pseudomonas aeruginosa*. *Polish Journal of Microbiology*, 61(4), 281–289.
- Gonzalez, J., & Hu, W. (1991). Effect of glutamate on the degradation of pentachlorophenol by *Flavobacterium* sp. *Applied Microbiology and Biotechnology*, 35, 100-104.
- Gong, X., Luan, Z., Pei, Y., & Wang, S. (2011). Culture conditions for flocculant production by Paenibacillus polymyxa BY-28. Journal of Environmental Science and Health Part A-Toxic/Hazardous Substances and Environmental Engineering, 38(4), 657-663.
- Gong, W., Wang, S., Sun, X., Liu, X., Yue, Q., & Gao, B. (2008). Bioflocculant production by culture of *Serratia ficaria* and its application in wastewater treatment. *Bioresource Technology*, 99(11), 4668-4674.
- Gracia-Lor, E., Sancho, J.V., Serrano, R., & Hernández, F. (2012). Occurrence and removal of pharmaceuticals in wastewater treatment plants at the Spanish Mediterranean area of Valencia. *Chemosphere*, 87(5), 453-462.
- Grandjean, P., & Landrigan, P. J. (2014). Neurobehavioural effects of developmental toxicity. *The Lancet Neurology*, 13(3), 330-338.
- Guangyu, Y., & Thiruvenkatachari, V. (2003). Heavy metals removal from aqueous solution by *Mucor rouxii*. *Water Research*, *37*(18), 4486-4496.
- Gudikandula, K., & Maringanti, S. (2016). Synthesis of silver nanoparticles by chemical and biological methods and their antimicrobial properties. *Journal of Experimental Nanoscience*, *11*(9), 714-721.
- Guinea, V., Spadini, L., Sarret, G., Muris, M., Delolme, C., & Gaudet, J. P. (2006). Zinc sorption to three gram-negative bacteria, combined titration modeling, and exafs study. *Environmental Science and Technology*, 40, 1806–1813.

- Guo, J., & Yu, J. (2014). Sorption characteristics and mechanisms of Pb (II) from aqueous solution by using bioflocculant MBFR10543. *Applied Microbiology and Biotechnology*, 98(14), 6431-6441.
- Guo, J., Yu, J., Xin, X., Zou, C., Cheng, Q., Yang, H.,... Li, N. (2015b). Characterization and flocculation mechanism of a bioflocculant from hydrolyzate of rice stover. *Bioresource Technology*, 177, 393-397.
- Gupta, S., Madan, R.N., & Bansal, M.C. (1987). Chemical composition of pinus carbaea hemicellulose. *Tappi Journal*, 70, 113-114.
- Hashim, R., Nadhari, W., Sulaiman, O., Kawamura, F., & Hiziroglu, S. (2011). Characterization of raw materials and manufactured binderless particle board from oil palm biomass. *Materials and Design*, 32(1), 246-254.
- Hassan M.A., Li T.P & Noor Z.Z. (2009). Coagulation and flocculation treatment of wastewater in textile industry using chitosan. *Journal of Chemical and Natural Resources Engineering*, 4(1), 43-53.
- He, J., Zou, J., Shao, Z., & Zhang, J. (2010). Characteristics and flocculating mechanism of a novel bioflocculant HBF-3 produced by deep-sea bacterium mutant *Halomona* sp. V3a. World Journal of Microbiology and Biotechnology, 26, 1135-1141.
- He, N., Li, Y., & Chen, J. (2004). Production of a novel polygalacturonic acid bioflocculant REA-11 by Corynebacterium glutamicum. Bioresource Technology, 94(1), 99-105.
- He, P., Lujing, G., Zheng, W., Duobin, M., Jizhong, W., & Chunping, X. (2012). Fermentation optimization, characterization and bioactivity of exopolysaccharides from *Funalia trogii*. Carbohydrate Polymers, 89, 17-23.
- Heberer, T. (2002). Occurrence, fate, and removal of pharmaceutical residues in the aquatic environment: A review of recent research data. *Toxicology Letters*, 131, 5-17.
- Huang, X., Xiaowen, B., Yanxia, Z., Baoyu, G., Yan, W., Shenglei, S.,... Qian, L. (2014). Effects of compound bioflocculant on coagulation performance and floc properties for dye removal. *Bioresource Technology*, 165, 116-121.
- Huang, M., Shen, R., & Xia, J. (2001). Microbial flocculates: Production and application on wastewater treatment. *Journal of Shanghai University*, (Natural Science), 7(3), 244-248.
- Ishii, H., Koyama, M., & Mitani, T. (2008). Removal of organic acids by highly swollen chitosan beads. Soluble polymer in water purification. *Journal of Environmental Science and Health*, 30(5), 945-950.
- Ismail, S., & Dadrasnia, A. (2015). Biotechnological potential of *Bacillus salmalaya* 139SI: A novel strain for remediating water polluted with crude oil waste. *PLoS ONE*, 10(4), Article # e0120931.

- Ismail, B., & Nampoothiri, K. (2010). Production, purification and structural characterization of an exopolysaccharide produced by a probiotic *Lactobacillus plantarum* MTCC 9510. *Archives of Microbiology*, *192*(12), 1049-1057.
- Jachlewski, S., Jachlewski, W. D., Linne, U., Brasen, C., Wingender, J., & Siebers, B. (2015). Isolation of extracellular polymeric substances from biofilms of the thermoacidophilic archaeon Sulfolobus acidocaldarius. Frontiers Bioengineering and Biotechnology, 3, Article # 123.
- Jang, W., Hwang, J., Kim, H., Ryu, J., & Le, S. (2017). Bacterial cellulose as an example product for sustainable production and consumption. *Microbial Biotechnology*, 10 (5), 1181-1185.
- Jarvis, P., Sharp, E., Pidou, M., Molinder, R., Parsons, S.A., & Jefferson, B. (2012). Comparison of coagulation performance and floc properties using a novel zirconium coagulant against traditional ferric and alum coagulants. Water Research, 46(13), 4179-4187.
- Jelic, A., Gros, M., Ginebreda, A., Cespedes-Sánchez, R., Ventura, F., Petrovic, M.,... Barcelo, D. (2011). Occurrence, partition and removal of pharmaceuticals in sewage water and sludge during wastewater treatment. Water Research, 45(3), 1165-1176.
- Jiao, R., Fabris, R., Chow, C., Drikas, M., Leeuwen, J., &Wang D. (2016). Roles of coagulant species and mechanisms on floc characteristics and filterability. Chemosphere, 150, 211-218.
- Jochen, S., Volker, S., & Bernd, R. (2015). Bacterial exopolysaccharides biosynthesis pathways and engineering strategies. *Frontiers in Microbiology*, *6*, 1-24.
- Joshi, N.K. (2016). *Production, Characterization and application of polymeric bacterial bioflocculant.* (PhD, Thesis, CSIR-Central Salt and marine Chemical research Institute, Bhavnagr University, India) Retrieved on 12 April 2018, https://www.csmcri.org.
- Kacar, Y., Arpa, C., Tan, S., Denizli, A., Genc, O., & Arica, M.Y. (2000). Biosorption of Hg (II) and Cd (II) from aqueous solution: Comparison of biosorptive capacity of alginate and immobilized live and heat inactivated *Phanerochaete chrysosporium*. *Process Biochemsitry*, 37, 601–610.
- Kafeenah, H. I., Osman, R., & Bakar, N. K. A. (2018). Disk solid-phase extraction of multi-class pharmaceutical residues in tap water and hospital wastewater, prior to ultra-performance liquid chromatographic-tandem mass spectrometry (UPLC-MS/MS) analyses. *Royal Society of Chemistry Advances*, 8(70), 40358-40368.
- Kalishwaral, K., Pandian, S., Deepak, V., & Samy K. (2010). Biosynthesis of silver and gold nanoparticles using *Brevibacterium casei*. Colloids and Surfaces B: Biointerfaces, 77(2), 257-262.
- Kanmani, P., Kumar, R., Yuvaraj, N., Paari, K., Pattukumar, V., & Venkatesan, A. (2011). Production and purification of a novel exopolysaccharide from lactic acid

bacterium *Streptococcus phocae* PI80 and its functional characteristics activity in vitro. *Bioresource Technology*, *102*(7), 4827-4833.

- Karthiga, K., & Natarajan K. (2015). Production and characterization of bioflocculants for mineral processing applications. *International Journal of Mineral Processing*, 137, 15-25.
- Kay, P., Blackwell, P., & Boxall, A. (2005). Transport of veterinary antibiotics in overland flow following the application of slurry to arable land. *Chemosphere*, 59(7), 951-959.
- Khiari, R., Dridi-Dhaouadi, S., Aguir, C., & Mhenni, M. (2010). Experimental evaluation of eco-friendly flocculants prepared from date palm rachis. *Journal of Environmental Science*, 22(10), 1539-1543.
- Keeley, J., Smith, A., Judd, S., & Jarvis, P. (2016). Acidified and ultrafiltered recovered coagulants from water treatment works sludge for removal of phosphorus from wastewater. *Water Research*, 1(88), 380-388.
- Kumari, S., Mangwani, N., & Das, S. (2017). Interaction of Pb (II) and biofilm associated extracellular polymeric substances of a marine bacterium *Pseudomonas* pseudoalcaligenes NP103. Spectrochimica Acta Part A: Molecular and Biomolecular Spectroscopy, 173, 655-665.
- Kumar, P., Senthamil, S., Lakshmi, P. A., Prem, K., Ganeshkumar, R. S., & Govindaraju, M. (2012). Synthesis of silver nanoparticles from *Sargassum tenerrimum* and screening phytochemicals for its antibacterial activity. *Nano Biomedicine Engineering*, 4(1),12-16.
- Kumar, C., Joo, H., Kavali, R., Choi, J., & Chang, C. (2004). Characterization of an extracellular biopolymer flocculant from a haloakalophilic *Bacillus* isolate. *World Journal of Microbiology and Biotechnology*, 20, 837-843.
- Kumari, K., Vijay, K., Avinash, M., & Bhavanath, J. (2014). Characterization and antibiofilm activity of extracellular polymeric substances from *Oceanobacillus iheyensis*. *Carbohydrate Polymers*, 101, 29-35.
- Kurane, R., Hatamochi, K., Kakuno, T., Kiyohara, M., Hirano, M & Taniguchi, Y. (1994). Production of a bioflocculant by *Rhodococcus erythropolis* S-l grown on alcohols. *Journal Bioscience Biotechnology and Biochemistry*, 58(2), 428-429.
- Kurane, R., Hatatkeyama, S., & Tsugeno, H. (1991). Correlation between flocculation production and morphological changes in *Rhodococcus erythropolis* S-1. *Journal of Fermentation and Bioengineering*, 72, 498-500.
- Kurane, R., & Matsuyama, H. (1994). Production of a bioflocculant by mixed culture. *Journal Bioscience Biotechnology and Biochemistry*, 58(9), 1589-1594.
- Kurane, R. & Nohata, Y. (1994). A new water-absorbing polysaccharide from *Alcaligenes latus. Bioscience, Biotechnology and Biochemistry*, 58, 235-238.

- Kurniawan, T. (2002). A research study on Cr (VI) removal from electroplating wastewater using chemically modified low-cost adsorbents and commercial activated carbon (Master thesis, Thammasat University). Sirindhorn International Institute of Technology (SIIT), Pathumthani. Thailand.
- Lachhwani, P. (2005). *Studies on polymeric bioflocculant producing microorganisms*. (Master thesis, Thapar Institute of Engineering and Technology). Patiala. India.
- Langley, R.L., Kao, Y., Mort, S.A., Bateman, A., Simpson, B.D & Reich, B.J. (2015). Adverse neurodevelopmental effects and hearing loss in children associated with manganese in well water, North Carolina, USA. *Journal of Environmental and Occupational Science*, 4(2), 62-69.
- Larsson, D.G.J., Pedro, C., & Paxeus, N. (2007). Effluent from drug manufactures contains extremely high levels of pharmaceuticals. *Journal of Hazardous Materials*, 30(148), 751-755.
- Lawrence A., & Prakash T.J. (2019). Physical and chemical characterization of green synthesized silver nanoparticles using stem of *Hibiscus vitifolius* L. and its antimicrobial and antioxidant potential. *Asian Journal of Pharmaceutical Clinical Research*, 12(6), 46-53.
- Lee, S., John, R., Mei, F., & Chong, C. (2014). A review on application of flocculants in wastewater treatment. *Process Safety Environmental Protection*, 92(6), 489-508.
- Lee, K., Morad, N., Teng, T., & Poh, B. (2012). Development, characterization and the application of hybrid materials in coagulation/flocculation of wastewater: a review. *Chemical Engineering Journal*, 203, 370-386.
- Lee, L.S., Fong, C.M., John, P. R., & Eleanor, B. (2015). Optimization of extraction and sludge dewatering efficiencies of bioflocculants extracted from *Abelmoschus* esculentus (okra). Journal of Environmental Management, 157, 320-325.
- Lei, L., Liang, D., Yu, D., Chen, Y., Song, W., & Li, J. (2015). Human health risk assessment of heavy metals in the irrigated area of Jinghui, Shaanxi, China, in terms of wheat flour consumption. *Environmental Monitoring and Assessment*, 187(10), Article # 647.
- Le-Minh, N., Khan, S., Drewes, J., & Stuetz, R. (2010). Fate of antibiotics during municipal water recycling treatment processes. *Water Research*, 44(15), 4295-4323.
- Li, L., Ma, F., & Zuo, H. (2016). Production of a novel bioflocculant and its flocculation performance in aluminium removal. *Bioengineered*, 7(2), 98-105.
- Li, Y., Xu, Y., Zheng, T., & Wang, H. (2017). Flocculation mechanism of the *actinomycete Streptomyces* sp. hsn06 on *Chlorella vulgaris*. *Bioresource Technology*, 239, 137-143.
- Li, O., Lu, C., Liu, A., Zhu, L., Wang, P., Qian, C.,... Wu, X. (2013). Optimization and characterization of polysaccharide-based bioflocculant produced by *Paenibacillus*

elgii B69 and its application in wastewater treatment. *Bioresource Technology*, 134, 87-93.

- Li, Q., Liu, H.L., Qi, Q., Wang, F., & Zhang, Y. (2010). Isolation and characterization of temperature and alkaline stable bioflocculant from Agrobacterium sp. M503. New Biotechnology, 27(6), 789 -794.
- Li, W., Zhou, W., Zhang, Y., Wang, J., & Zhu, X. (2008). Flocculation behaviour and mechanism of an exopolysaccharide from the deep sea psychrophilic bacterium *Pseudoalteromonas* sp. SM9913. *Bioresource Technology*, 99(15), 6893-6899.
- Li, X., Yang, Q., Huang, K., Zeng, G., Liao, D., & Liu, J. (2007). Screening and characterization of a bioflocculant produced by *Aeromonas* sp. *Biomedical and Environmental Science*, 20, 274-278.
- Li, X.Y., & Yang, S.F. (2007). Influence of loosely bound extracellular polymeric substances (EPS) on the flocculation, sedimentation and dewaterability of activated sludge. *Water Research*, 41(5), 1022-1030.
- Li, Y., Li, Q., Hao, D., Hu, Z., Song, D., & Yang, M. (2014). Characterization and flocculation mechanism of an alkali-activated polysaccharide flocculant from *Arthrobacter* sp. B4. *Bioresource Technology*, *170*, 574-577.
- Li, Z., Zhong, S., Lei, H., Chen, R., Yu, Q., & Li, H. (2009). Production of a novel bioflocculant by *Bacillus licheniformis* X14 and its application to low temperature drinking water treatment. *Bioresource Technology*, *100*(14), 3650-3656.
- Lian, B., Ye, C., Jin, Z., Teng, H.H., Zhu, L., & Yua, S. (2008). Microbial flocculation by *Bacillus mucilaginosus*: Applications and mechanisms. *Bioresource Technology*, 99(11), 4825-4831.
- Lin, J. & Harichund, C. (2011). Isolation and characterization of heavy metal removing bacterial bioflocculants. African Journal of Microbiology Research, 5(6), 599-607.
- Lin, J. & Harichund, C. (2012). Production and characterization of heavy-metal removing bacterial bioflocculants. *African Journal of Biotechnology*, *11*(40), 9619-9629.
- Lin, Z. & Zhang, H. (2004). Anti-tumor and immune-regulatory activities of *Ganoderma lucidum* and its possible mechanisms. *Acta Pharmacologica Sinica*, 25(11), 1387-1395.
- Liu, C., Lu, J., Lu, L., Liu, Y., Wang, F., & Xiao, M. (2010). Isolation, structural characterization and immunological activity of an exopolysaccharide produced by *Bacillus licheniformis 8-37-0-1*. *Bioresource Technology*, *101*(14), 5528-5533.
- Liu, C., Wang, K., Ji-Hong, J., Wei-Jie, L., & Jing-Yuan, W. (2015). A novel bioflocculant produced by a salt-tolerant, alkaliphilic and biofilm-forming strain *Bacillus agaradhaerens* C9 and its application in harvesting *Chlorella minutissima* UTEX 2341. *Biochemical Engineering Journal*, 93, 166-172.

- Liu, L. & Chen, W. (2010). Characteristics and culture conditions of a bioflocculant produced by *Penicillium* sp. *Biomedical and Environmental Science*, 23(3), 213-218.
- Liu, W., Kai, W., Li, B., Hong, L., & Jin, S. (2010). Production and characterization of an intracellular bioflocculant by *Chryseobacteria daeguense* W6 cultured in low nutrition medium. *Bioresource Technology*, 101, 1044-1048.
- Liu, Z., Zhi-quan, H., Tao, W., Yan-ying, C., Jianbin, Z., Jing-ran, Y.,... Yong, L. (2013). Production of novel microbial flocculants by *Klebsiella* sp. TG-1 using waste residue from the food industry and its use in defecating the trona suspension. *Bioresource Technology*, 139, 265-271.
- Liu, H., Chen, G., & Wang, G. (2015b). Characteristics for production of hydrogen and bioflocculant by *Bacillus* sp. XF-56 from marine intertidal sludge. *International Journal of Hydrogen Energy*, 40(3), 1414-1419.
- Liu, W., Hao, Y., Jiang, J., Zhu A., Zhu, J., & Zhen, D. (2016). Production of a bioflocculant from *Pseudomonas veronii* L918 using the hydrolyzate of peanut hull and its application in the treatment of ash-flushing wastewater generated from coal fired power plant. *Bioresource Technology*, 218, 318-325.
- Liu, J., Liang C., & Dou, S. (2017). Bioaccumulation of heavy metals and health risk assessment in three benthic bivalves along the coast of Laizhou bay, China. *Marine Pollution Bulletin*, 117(1-2), 98-110.
- Lixin, L., Maa, F., & Zuob, H. (2016). Production of a novel bioflocculant and its flocculation performance in aluminum removal. *Bioengineered*, 7(2), 98-105.
- Lodeiro, P., Barriada, J.L., Herrero, R., & Sastre de Vicente M.E (2006). The marine macroalga *Cystoseira baccata* as biosorbent or cadmium (II) and lead (II) removal: kinetic and equilibrium studies, *Environmental Pollution*, 142, 264–273.
- Lofrano, G., Meric, S., Zengin, G.E., & Orhon, D. (2013). Chemical and biological treatment technologies for leather tannery chemicals and wastewaters: a review. *Science of the Total Environment*, 461-462, 265-281.
- Low, L., Teng, T., Anees, A., Morad, N., & Wong, Y. (2011). A novel pretreatment method of lignocellulosic material as adsorbent and kinetic study of dye waste adsorption. *Journal of Water Air Soil Pollution*, 218(1-4), 293-306.
- Lopez, E., Ramos, I., & Sanroman, M. (2003). Extracellular polysaccharides production by *Arthrobacter viscosus*. *Journal of Food Engineering*, *60*(4), 463-467.
- Lu, W., Zhang, T., Zhang, D., Li, C., Wen, J., & Du, L. (2005). A novel bioflocculant produced by *Enterobacter aerogenes* and its use in defaecating the trona suspension. *Biochemical Engineering Journal*, 27, 1-7.
- Luo, Z., Li, C., Changhong, C., Wei, Z., Ming, L., Ye, H.,... Jiangang, Z. (2014). Production and characteristics of a bioflocculant by *Klebsiella pneumonia* YZ-6 isolated from human saliva. *Applied of Biochemistry and Biotechnology*, 172(3), 1282-1292.

- Ma, F., Zheng, L., & Chi, Y. (2008). Applications of biological bioflocculants (BFs) for coagulation treatmentin water purification: Turbidity elimination. *Chemical Biochemical Engineering Quarterly*, 22(3), 321-326.
- Mabinya, L., Cosa, S., Mkwetshana, N., & Okoh, A. (2011). *Halomonas* sp. OKOH- A marine bacterium isolated from the bottom sediment of Algoa Bay-produces a polysaccharide bioflocculant: Partial characterization and biochemical analysis of its properties. *Molecules*, 16(6), 4358-4370.
- Mabinya, L., Cosa, S., Nwodo, U., & Okoh, A. (2012). Studies on bioflocculant production by *Arthrobacter* sp. Raats, a freshwater bacteria isolated from Tyume River, South Africa. *International Journal of Molecular Science*, 13(1), 1054-1065.
- Makapela, B., Okaiyeto, K., Ntozonke, N., Nwodo, U.U., Green, E., Mabinya, L.V.,... Okoh, A.I. (2016). Assessment of *Bacillus pumilus* isolated from fresh water milieu for bioflocculant production. *Applied Science*, 6(8), Article # 211.
- Maharani, V., Sundaramanickam, A., & Balasubramanian, T. (2016). In vitro anticancer activity of silver nanoparticle synthesized by *Escherichia coli* VM1 isolated from marine sediments of Ennore southeast coast of India. *Enzyme and Microbial Technology*, 95, 146-154.
- Mao, Y., Tian, C., Zhu, J., Zhang, T., & Tong, L. (2010). Production of a novel biopolymer by the culture of *Bacillus cereus* B-11 using molasses wastewater and its use for dye removal. *Advanced Materials Research*, 230, 1119-1122.
- Martín-Cereceda, M., Jorand, F., Guinea, A., & Block, J.C. (2001). Characterization of extracellular polymeric substances in rotating biological contactors and activated sludge flocs. *Environmental Technology*, 22(8), 951-959.
- Marsili, E., & Das, S. (2016). Biosynthetic nanoparticles for biotechnological and biomedical applications. *Enzyme and Microbial Technology*, 95, 1-3.
- Manivasagan, P., Kang, K.H., Kim, D.G., & Kim, S.K. (2015). Production of polysaccharide-based bioflocculant for the synthesis of silver nanoparticles by *Streptomyces* sp. *International Journal of Biological Macromolecules*, 77, 159– 167.
- Mata, Y.N., Blazquez, M.L., Ballester, A., Gonzalez, F., & Munoz, J.A. (2009). Sugarbeet pulp pectingels as biosorbent for heavy metals: Preparation and determination of biosorption and desorption characterization. *Chemical Engineering Journal*, 150, 289-301.
- Ma, X., Zuo, H., Tian, M., Zhang, L., Meng, J., Zhou, X.,... Liu, Y. (2016). Assessment of heavy metals contamination in sediments from three adjacent regions of the Yellow River using metal chemical fractions and multivariate analysis techniques. *Chemosphere*, 144, 264-272.
- Mishra, A. & Jha, B. (2013). Microbial Exopolysaccharides, In Rosenberg E, DeLong E.F, Thompson F, Lory S, Stackebrandt E, (Eds.), *The prokaryote: Applied*

bacteriology and biotechnology, (179-192). Berlin Heidelberg, Germany: Springer.

- Mokaddem, H., Sadaoui, Z., Boukhelata, N., Azouaou, N., & Kaci, Y. (2009). Removal of Cadmium from aqueous solution by polysaccharide produced from *Paenibacillus polymyxa*. *Journal of Hazardours Materials*, *172*, 1150-1155.
- Mokhtari, N., Daneshpajouh, Sh., Seyedbagheri, S., Atashdehghan, R., Abdi, K.h.,... Shahverdi, A.R. (2009). Biological synthesis of very small silver nanoparticles by culture supernatant of *Klebsiella pneumonia*: The effects of visible-light irradiation and the liquid mixing process. *Materials Research Bulletin*, 44, 1415-1421.
- Monshi, A., Monshi, M.R., & Foroughi, M.R. (2012). Modified scherrer equation to estimate more accurately nano-crystallite size using XRD. *World Journal of Nano Science and Engineering*, 2(3), 154-160.
- Moosavi-Nasab, M., Pashangeh, S., & Rafsanjani, M. (2010). Effect of fermentation time on xanthan gum production from sugar beet molasses. *World Academy of Science*. *Engineering and Technology*, 44, 1244-1247.
- Morillo, J., Aguilera, M., Ramos, A., & Sanchez, M. (2006). Production of a metalbinding exopolysaccharide by *Paenibacillus jamilae* using two-phase olive-mill waste as fermentation substrate. *Current Microbiology*, 53, 189-193.
- Morris, G., & Harding, S. (2009). Polysaccharides, microbial. *Encyclopedia of Microbiology*, 3, 482-494.
- Moussas, P.A. & Zouboulis, A.I. (2009). A new inorganic-organic composite coagulant, consisting of Polyferric Sulfate (PFS) and Polyacrylamide (PAA). *Water Research*, 43, 3511-3524.
- Murgatroyd, C., Berry, M., Bailey, K., & Whitehouse, P. (1996). A review of polyelectrolytes to identify priorities for EQS development. Foundation for water research, Allen house. *The Listons. R and D Technical Report (p.21). Environment Agency, Bristol.*
- Muthulakshmi, L., Rajini, N., Rajalu, A.V., Siengchin, S., & Kathiresan, T. (2017). Synthesis and characterization of cellulose/Silver nanocomposites from bioflocculant reducing agent. *International Journal of Biological Macromolecules*, 5, Article # 068.
 - Muthulakshmi, L., Nellaiah, H., Kathiresan, T., Rajini, N., & Christopher, F. (2017). Identification and production of bioflocculants by *Enterobacter* sp. and *Bacillus* sp. and their characterization studies. *Preparative Biochemistry and Biotechnology*, 47(5), 458-467.
 - Nakata, K. & Kurane, R. (1999). Production of extracellular polysaccharide biofloccculants by *Klebsiella pneumoniae*. Journal of Bioscience Biotechnology and Biochemistry, 63, 2064-2068.

- Narkis, N., & Rebhun, M. (1975). The mechanism of flocculation processes in the presence of humic substances. *Journal-American Water Works Associetion*, 67(2), 101-108.
- Natarajan, K. A. (2017). Use of bioflocculants for mining environmental control. *Transaction of the Indian Institute of Metals*, 70(2), 519-525.
- Natarajan, K. A. (2015). Production and characterization of bioflocculants for mineral processing applications. *International Journal of Mineral Processing*, 137, 15-25.
- Nichols, C., Bowman, J., & Guezennec, J. (2005). Effects of incubation temperature on growth and production of exopolysaccharides by an antarctic sea ice bacterium grown in batch culture. *Applied Evironmental Microbiology*, *71*(7), 3519-3523.
- Nikolaou, A., Meric, S & Fatta, D. (2007). Occurrence patterns of pharmaceuticals in water and wastewater environments. *Analytical Bioanalytical Chemistry*, 387(4) 1225-1234.
- Ntsaluba, L., Nwodo, U., Mabinya, L., & Okoh, A. (2013). Studies on bioflocculant production by a mixed culture of *Methylobacterium* sp. Obi and *Actinobacterium* sp. Mayor. *BMC Biotechnology*, *13*(1), Article # 62.
- Ntsaluba, L., Agundiade, O., Mabinya, L., & Okoh, A.I. (2011). Studies on bioflocculant production by *Methylobacterium* sp. Obi isolated from a freshwater environment in South Africa. *African Journal Microbiology Research*, 5(26), 4533-4540.
- Ntsangani, N., Okaiyeto, K., Nwodo, U., Olaniran, A., Mabinya, L., & Okoh, A. (2017). Bioflocculation potentials of a uronic acid-containing glycoprotein produced by *Bacillus* sp. AEMREG4 isolated from Tyhume River, South Africa. *Biotechnology*, 7(1), Article # 78.
- Nurul, A.B., Ngatimani, M., Loh, S.K., & Choo, Y.M. (2014). Characteristics of plam oil effluent (Pome) in an anaerobic biogas digester. Asian Journal of Microbiology, Biotechnology and Environmental Science, 16(1), 225-231.
- Nwodo, U., Green, E., Mabinya, L., Okaiyeto, K., Rumbold, K., Obi, L.,... Okoh, A. (2014). Bioflocculant production by a consortium of *Streptomyces* and *Cellulomonas* species and media optimization via surface response model. *Colloids and Surfaces B, Biointerfaces*, 116, 257-264.
- Nwodo, U., & Okoh, A. (2014). Mixed culture fermentation and media optimization by response surface model: *Streptomyces* and *Brachybacterium* species in bioflocculant production. *Molecule*, 19(8), 11131-11144.
- Nwodo, U. & Okoh, A. (2013). Characterization and flocculation properties of biopolymeric flocculant (glycosaminoglycan) produced by *Cellulomonas* sp. Okoh. *Journal of Applied Microbiology*, 114(5), 1325-1337.
- Nwodo, U., Agunbiade, M., Green, E., Nwamadi, M., Rumbold, K., & Okoh, A. (2013). Characterization of an exopolymeric flocculant produced by a *Brachybacteria* sp. *Materials*, 6(4), 1237-1254.

- Nwodo, U., Green, E., & Okoh, A. (2012). Bacterial exopolysaccharides: Functionality and prospects. *International Journal of Molecular Science*, *13*(11), 14002-14015.
- Okajima, M., Sornkamnerd, S & Kaneko, T. (2018). Development of functional bionanocomposites using *Cyanobacterial* polysaccharides. *Chemical Record*, 18(7-8), 1167-1177.
- Okaiyeto, K., Nwodo, U., Mabinya, L., & Okoh, A. (2016). Implications for public health demands alternatives to inorganic and synthetic flocculants: Bioflocculants as important candidates. *Microbiology Open*, 5(2), 177-211.
- Okaiyeto, K., Nwodo, U., Mabinya, L., & Okoh, A. (2015). Bacillus toyonensis strain AEMREG6, a bacterium isolated from South Africa marine environment sediment samples produces a glycoprotein bioflocculant. Molecule, 20(3), 5239-5259.
- Okaiyeto, K., Nwodo, U., Mabinya, L., Okoli, A., & Okoh, A. (2015). Characterization of a bioflocculant (MBF-UFH) produced by *Bacillus* sp. AEMREG7. *International Journal of Molecular Science*, *16*(6), 12986-13003.
- Okaiyeto, K., Nwodo, U., Mabinya, L., & Okoh, A. (2014). Evaluation of the flocculation potential and characterization of bioflocculant produced by *Micrococcus* sp. Leo. *Applied Biochemistry and Microbiology*, 50, 601-608
- Okaiyeto, K., Nwodo, U., Mabinya, L., & Okoh, A. (2013). Characterization of a bioflocculant produced by a consortium of *Halomonas* sp. Okoh and *Micrococcus* sp. Leo. *International Journal of Environmental Research Public Health*, 10(10), 5097-5110.
- Ogunsade, O., Bakare, M., & Adewale, I. (2015). Purification and characterization of bioflocculant produced by *Bacillus amyloliquefaciens* ABL 19 isolated from Adeti Stream, Ilesa, Osun State, Nigeria. *Nature and Science*, 13(2), 54-58.
- Ong, S., Ho, L., Wong, Y., & Raman, K. (2012). Performance and kinetics study on bioremediation of diazo dye (Reactive Black 5) in wastewater using spent GACbiofilm sequencing batch reactor. *Journal of Water Air and Soil Pollution*, 223(4), 1615-1623.
- Ozacar, M., & Sengil, A. (2000). Effectiveness of tannins obtained from valonia as a coagulant-aid for dewatering of sludge. *Water Research*, 34(4), 1407-1412.
- Ozdemir, G., Ozturk, T., Ceyhan, N., Isler, R., & Cosar, T. (2003). Heavy metal biosorption by biomass of *Ochrobactrum anthropi* producing exopolysaccharide in activated sludge. *Bioresource Technology*, *90*(1), 71-74.
- Panacek, A., Kvitek, L., Prucek, R., Kolar, M., Vecerova, R., Pizurova, N.,... Zboril, R. (2006). Silver colloid nanoparticles: Synthesis, characterization, and their antibacterial activity. *The Journal of Physical Chemistry*, 110(33), 16248-16253.
- Parikh, R., Ramanathan, R., Coloe, P., Bhargava, S., Patole, M., Shouche, Y.,... Bansal, V. (2011). Genus-wide physicochemical evidence of extracellular crystalline

silver nanoparticles biosynthesis by *Morganella* spp. *PLoS ONE*, 6(6), Article # e21401.

- Pathak, M., Devi, A., Bhattacharyya, K., Sarma, H., Subudhid, S., & Lald, B., (2015). Production of a non-cytotoxic bioflocculant by a bacterium utilizing a petroleum hydrocarbon source and its application in heavy metal removal. *Royal Society of Chemistry Advanced*, 5, 66037–66046.
- Pathak, M., Sarma, H., Bhattacharyya, K., Subudhi, S., Bisht, V., Lald, B.,... Devi, A. (2017). Characterization of a novel polymeric bioflocculant produced from bacterial utilization of *n*-Hexadecane and its application in removal of heavy metals. *Frontiers In Microbiology*, 7(8), Article # 170.
- Pathak, M., Arundhuti, D., Hridi, K., & Banwari, L. (2014). Application of bioflocculating property of *Pseudomonas aeruginosa* strain IASST201 in treatment of oil-field formation water. *Journal of Basic Microbiology*, 54, 1-12.
- Patil, S., Patil, C., Salunke, B., Salunkhe, R., Bathe, G., & Patil, D. (2011). Studies on characterization of bioflocculant exopolysaccharide of *Azotobacter indicus* and its potential for wastewater treatment. *Applied Biochemistry and Biotechnology*, 163(4), 463-472.
- Paulo, M., Boffo, F., Branco, A., Valente, M., Melo, S., & Ferreira, G. (2012).Production, extraction and characterization of exopolysaccharides produced by the native *Leuconostoc pseudomesenteroides* R2 strain. *Anais Da Academia Brasileira de Cià ncias*, 84(2), 495–508.
- Peng, L., Chunping, Y., Guangming, Z., Lu, W., Chuanhua, D., Zhiyong, L.,... Yuanyuan, Z. (2014). Characterization and application of bioflocculant prepared by *Rhodococcus erythropolis* using sludge and livestock wastewater as cheap culture media. *Applied Microbiology and Biotechnology*, 98, 6847-6858.
- Piyo, N., Cosa, S., Mabinya, L.V & Okoh, A.I. (2011). Assessment of bioflocculant production by *Bacillus* sp. Gilbert, a marine bacterium isolated from the bottom sediment of Algoa Bay. *Marine Drugs*, 9(7), 1232-1242.
- Poddar, P., & Sahu, O., (2017). Quality and management of wastewater in sugar industry. *Applied Water Science*, 7, 461-468.
- Polizzi, S., Pira, E., Ferrara, M., Buginani, M., Papaleo, A., Albera, R.,... Palmi, S. (2002). Neurotoxic effects of aluminum among foundry workers and Alzheimer's disease. *Neurotoxicology*, 23, 761-774.
- Prasertsan, P., Dermlim, W., Doelle, H., & Kennedy, J. (2006). Screening, characterization and flocculating property of carbohydrate polymer from newly isolated *Enterobacter cloacae* WD7. *Carbohydrate Polymers*, *66*(3), 289-297.
- Pu, S., Qin, L., Che, J., Zhang, B., & Xu, M. (2014). Preparation and application of a novel bioflocculant by two strains of *Rhizopus* sp. using potato starch wastewater as nutrilite. *Bioresource Technology*, 162, 184-191.

- Puranik, P.R & Pakniker, K.M. (1999). Biosorption of lead, cadmium and zinc by *Citrobacter* strain MCM B-181: Characterization studies. Biotechnology Progress, 15, 228–237.
- Rasulov, B., Li, L., Liu, Y., Mohamad, O., Xiao M., Ma, J.,... Wen-Jun, L. (2017). Production, characterization and structural modification of exopolysaccharidebased bioflocculant by *Rhizobium radiobacter* SZ4S7S14 and media optimization. *Biotechnology*, 7(3), Article # 179.
- Rasulov, B., Rustamova, N., Qing, Z., Yili, A., & Aisa, H. (2016c). Synthesis of silver nanoparticles on the basis of low and high molar-mass exopolysaccharides of Brady *Rhizobium japonicum* 36 and its antimicrobial activity against some pathogens. *Folia Microbiol*, 61(4), 283–293.
- Ran, J., Wang, D., Wang, C., Zhang, G., & Zhang, H. (2016). Heavy metal contents, distribution, and prediction in a regional soil-wheat system. *The Science of the Total Environment*, 544, 422-431.
- Rani, P., Sumit, M., & Gautam, S. (2013). Microwave based synthesis of polymethyl methacrylate grafted sodium alginate: its application as flocculant. *Carbohydrate Polymers*, 91(2), 686-692.
- Raveendran, S., Poulose, A.C., Yoshida, Y., Maekawa, T. & Kumar, D.S., (2013). Bacterial exopolysaccharide based nanoparticles for sustained drug delivery, cancer chemotherapy and bioimaging. *Carbohydrate Polymers*, 9, 22–32.
- Ravella, S., Quinones, T., Retter, A., Heiermann, M., Amon, T., & Hobbs, P. (2010). Extracellular polysacchride (EPS) production by a novel strain of yeast-like fungus Aureobasidium pullulans. Carbohydrate Polymers, 82(3), 728-732.
- Rawat, M., & Rai, J. (2013). Adsorption of heavy metals by *Paenibacillus validus* strain MP5 isolated from industrial effluent-polluted soil. *Bioremediation Journal*, 16(2), 66-73.
- Raza, W., Makeen, K., Wang, Y., Xu, Y., & Qirong, S. (2011). Optimization, purification, characterization and antioxidant activity of an extracellular polysaccharide produced by *Paenibacillus polymyxa* SQR-21. *Bioresource Technology*, 102(10), 6095-6103.
- Renault, F., Sancey, B., Badot, P.M., & Crini, G. (2009). Chitosan for coagulation/flocculation processes-An eco-friendly approach. *European Polymer Journal*, 45(5), 1337-1348.
- Rinaudo, M. (2006). Chitin and chitosan: Properties and applications. *Progress in Polymer Science*, 31(7), 603-632.
- Rong, H., Gao, B., Dong, M., Zhao, Y., Sun, S., Yue, W.,... Li, Q. (2013). Characterization of size, strength and structure of aluminum-polymer dualcoagulant flocs under different pH and hydraulic conditions. *Journal of Hazardous Material*, 252, 330-337.

- Roussy, J., Chastellan, P., Vooren, M.V., & Guibal, E. (2005). Treatment of inkcontaining wastewater by coagulation/flocculation using biopolymers. *Water SA*, 31(3), 369-376.
- Rout, C., & Sharma, A. (2011). Assessment of drinking water quality: A case study of Ambala cantonment area, Haryana, India. *International Journal of Environmental Science*, 2(2), 933-945.
- Ruden, C. (2004). Acrylamide and cancer risk-expert risk assessments and the public debate. *Food and Chemical Toxicology*, 42(3), 335-349.
- Sahoo, D., Kar, R., & Das, R. (1992). Bioaccumulation of heavy metal ions by *Bacillus* circulans. Bioresource Technology, 41, 177–179.
- Salehizadeh, H., & Shojaosadati, S. (2001). Extracellular biopolymeric flocculants: Recent trends and biotechnological importance. *Biotechnology Advances*, 19(5), 371-385.
- Salehizadeh, H. & Shojaosadati, S. (2003). Removal of metal ions from aqueous solution by polysaccharide produced from *Bacillus firmus*. Water Research, 37(17), 4231-4235.
- Salehizadeh, H. & Yan, N. (2014). Recent advances in extracellular biopolymer flocculants. *Biotechnology Advances*, 32(8), 1506-1522.
- Salehizadeh, H., Hekmatian, E., Sadeghi, M. & Kennedy, K. (2012). Synthesis and characterization of magnetic core-shell Fe₃O₄-gold-chitosan nanostructure. *Journal of Nanobiotechnology*, 10, 1–7.
- Sam, S., Kucukasik, F., Yenigun, O., Nicolaus, B., Oner, E., & Yukselen, M. (2011). Flocculating performances of exopoly-saccharides produced by a halophilic bacterial strain cultivated on agro-industrial waste. *Bioresource Technology*, 102(2), 1788-1794.
- Santos, LH., Araújo, AN., Fachini, A., Pena, A., Delerue-Matos, C., & Montenegro, MC. (2010). Ecotoxicological aspects related to the presence of pharmaceuticals in the aquatic environment. *Journal Hazard material*, 175(1-3), 45-95.
- Sathiyanarayanan, G., Seghal, G., & Selvin, J. (2013). Synthesis of silver nanoparticles by polysaccharide bioflocculant produced from marine *Bacillus subtilis* MSBN17. *Colloids and Surfaces B: Biointerfaces*, 102, 13-20.
- Sarkar, B., Chakrabarti, P., Vijaykumar, A & Kale, V. (2006). Wastewater treatment in dairy industries-possibility of reuse. *Desalination*, 195(1-3), 141-152.
- Saravanan, C., & Shetty, P. (2016). Isolation and characterization of exopolysaccharide from *Leuconostoc lactis* KC117496 isolated from idli batter. *International Journal of Biological Macromolecules*, 90, 100-106.
- Seymour, F. R., Knapp, R. D., Chen, E. C. M., Jeanes, A., & Bishop, S. H. (1979). Structural analysis of dextrans containing 4-O- α -d-glucosylated α -d-glucopyranosyl residues at the branch points, by use of ¹³c-nuclear magnetic

resonance spectroscopy and gas liquid chromatography-mass spectrometry. *Carbohydrate Research*, 75, 275–294.

- Shahadat, M., Teng, T., Rafatullah, M., Shaikh, Z., Sreekrishnan, S., & Ali, W. (2017). Bacterial biofloculants: A review of recent advances and perspectives. *Chemical Engineering Journal*, 328, 1139-1152.
- Sharma, B., Dhuldhoya, N., & Merchant, U. (2006). Flocculants an Ecofriendly Approach. *Journal of Polymers and the Environment*, 14, 195-202.
- Sharma, V.K., Yngard, R.A. & Lin, Y., (2009). Silver nanoparticles: Green synthesis and their antimicrobial activities. *Advances Colloid and Interface Science*, 145(1), 83– 96.
- Shen, X., Gao, B., Huang, X., Bu, F., Yue, Q., Li, R.,... Jin, B. (2017). Effect of the dosage ratio and the viscosity of PAC/PDMDAAC on coagulation performance and membrane fouling in a hybrid coagulation-ultrafiltration process. *Chemosphere*, 173, 288-298.
- Sheng, Y., Zhang, Q., Sheng, Y., Li, C., & Wang, H. (2006). Screening and flocculating properties of bioflocculant-producing microorganisms. Journal of University of Science and Technology Beijing, 13, 289-292.
- Sheng, G., Yu, H., & Li, X. (2010). Extracellular polymeric substances (EPS) of microbial aggregates in biological wastewater treatment systems: A review. *Biotechnology Advances*, 28, 882-894.
- Shi, B. & Tang, Z. (2006). Preparation and Characterization of organic polymer modified composite polyaluminium chloride. *Journal of Environmental Science*, 18, 214-220.
- Shih, I., Van, Y., Yeh, L., Lin, H., & Chang, Y. (2010). Production of a biopolymer flocculant from *Bacillus licheniformis* and its flocculation properties. *Bioresource Technology*, 78(3), 267-272.
- Shih, I., & Van, Y. (2001). The production of poly-(gamma-glutamic acid) from microorganisms and its various applications. *Bioresource Technology*, 79(3), 207-225.
- Shi-Jie, Y., Min, S., Guo-Ping, S., Yin, L., Wen-Wei, L., & Sheng, Y. (2011). Identification of key constituents and structure of the extracellular polymeric substances excreted by *Bacillus megaterium* TF10 for their flocculation capacity. *Environmental Science and Technology*, 45, 1152–1157.
- Shrivastava, S., Bera, T., Roy, A., Singh, G., Ramachandrarao, P., & Dash, D. (2007). Characterization of enhanced antibacterial effects of novel silver nanoparticles. *Nanotechnology*, 18, 1–9.
- Singh, R., Wahdwani, S., Shedbalkar, U., & Chopade, B.A. (2015). Bacteriagenic silver nanoparticles: Synthesis, mechanism, and applications. *Applied Microbiology and Biotechnology*, 99(11), 4579-4593.

- Sobeck, M., & Higgins, M. (2002). Examination of three theories for mechanism of cation-induced bioflocculation. *Water Research*, *36*, 527-538.
- Suárez, S., Carballa, M., Omil, F., & Lema, J.L., (2008). How are pharmaceutical and personal care products (PPCPs) removed from urban wastewaters: *Reviews in Environmental Science Biotechnology*, 7, 125-138.
- Subudhi, S., Bisht, V., Batta, N., Pathak, M., Devi, A., & Lal, B. (2016). Purification and characterization of exopolysaccharide bioflocculant produced by heavy metal resistant *Achromobacter xylosoxidans*. *Carbohydrate Polymers*, *137*, 441-451.
- Suh, H., Moon, H., Seo, T., Kim, K., Jeon, I., Park, G.,... Park, Y.I. (2002). Physicochemical and rheological properties of a bioflocculant BF-56 from *Bacillus* sp. A56. *Journal of Microbiology and Biotechnology*, 12, 209–216.
- Sun, P., Hui, C., Bai, N., Yang, S., Wan, L., & Zhang, Q. (2015a). Revealing the characteristics of a novel bioflocculant and its flocculation performance in *Microcystis aeruginosa* removal. *Scientific Reports*, 5, Article # 17465.
- Suopajarvi, T., Liimatainen, H., Hormi, O., & Niinimaki, J. (2013). Coagulationflocculation treatment of municipal wastewater based on an ionized nanocelluloses. *The Chemical Engineering Journal*, 231, 59-67.
- Szyguła, A., Guibal, E., Palacin, M., Ruiz, M., & Sastre, A. (2009). Removal of an anionic dye (Acid Blue 92) by coagulation-flocculation using chitosan. *Journal of Environmental Management*, 90(10), 2979-2986.
- Taha, G.M., Arifien, A.E., El-Nahas, S., (2011). Removal efficiency of potato peels as a new biosorbent material for uptake of Pb(II), Cd(II) and Zn(II) from the aqueous solutions. *Journal of Solid Waste Technology Management*, 37, 128–140.
- Tang, J., Qi, S., Li, Z., An, Q., Xie, M., Yang, B.,... Wang, Y. (2014). Production, purification and application of polysaccharide-based bioflocculant by *Paenibacillus mucilaginosus*. *Carbohydrate Polymer*, 113, 463-470.
- Tenny, M., & Stumm, W. (1965). Chemical flocculation of microorganisms in biological waste treatment. *Journal Water Pollution Control Federation*, 37(10), 1370-1388.
- Timbrell, J. (2002). *Principles of biochemical toxicology* (3rd ed.). London, UK: Taylor & Francis.
- Tingirikari, J. M., Kothari, D., Shukla, R., & Goyal, A. (2014). Structural and biocompatibility properties of dextran from *Weissella cibaria* JAG8 as food additive. *International Journal of Food Sciences and Nutrition*, 65(6), 686-691.
- Tiwari, A., Singh, A., Singh, P., & De Maio, M. (2016). Estimation of heavy hetal contamination in groundwater and development of a heavy metal pollution index by using GIS technique. *Bulletin Environmental Contamination and Toxicology*, 96(4), 508-515.

- Tiwari, A., De Maio, M., Singh, P., & Mahato, M. (2015). Evaluation of surface water quality by using GIS and a heavy metal pollution index (HPI) Model in a Coal Mining Area, India. *Bulletin Environmental Contamination Toxicology*, 95(3), 304-310.
- Toeda, K. & Kuranem, R. (1991). Microbial flocculant from *Alkaligenes cupidus* KT201. *Agricultural and Biological Chemistry*, 55, 2793–2799.
- Topp, E., Monteiro, S., Beck, A., Coelho, B., Boxall, A., Duenk, P., ... Metcalfe, C. (2008). Runoff of pharmaceuticals and personal care products following application of biosolids to an agricultural field. *The Science of the Total Environment*, 396(1), 52-59.
- Tsai, W., Lee, M., & Chang, Y. (2009). Fast pyrolysis of rice straw, sugarcane bagasse and coconut shell in an induction-heating reactor. *Journal of Analytical and Applied Pyrolysis*, 76(1-2), 230-237.
- Ugbenyen, A., & Okoh, A. (2014). Characteristics of a bioflocculant produced by a consortium of *Cobetia* and *Bacillus* species and its application in the treatment of wastewaters. *Water SA*, 40(1), 139-144.
- Ugbenyen, A., Cosa, S., Mabinya, L., & Okoh, A. (2014). Bioflocculant production by *Bacillus* sp. Gilbert isolated from a marine environment in South Africa. *Applied Biochemistry and Microbiology*, 50(1), 49-54.
- Ugbenyen, A., & Okoh, A. (2013). Flocculating properties of a bioflocculant produced by *Bacillus* sp. isolated from a marine environment in South Africa. *Chemical and Biochemical Engineering Quarterly*, 27(4), 511-518.
- Ugbenyen, A., Cosa, S., Mabinya, L., Babalola, O., Aghdasi, F & Okoh, A. (2012). Thermostable bacterial bioflocculant produced by *Cobetia* sp. isolated from Algoa Bay (South Africa). *International Journal of Environmental Research and Public Health*, 9(6), 2108-2120.
- Vieno, N., Tuhkanen, T., & Kronberg, L. (2007). Elimination of pharmaceuticals in sewage treatment plants in Finland. *Water Research*, 41(5), 1001-1012.
- Viswanathan, M., Arumugam, S., & Thangavel, B. (2016). In vitro anticancer activity of silver nanoparticle synthesized by *Escherichia coli* VM1 isolated from marine sediments of Ennoresoutheast coast of India. *Enzyme and Microbial Technology*, 95, 146-154.
- Wan, C., Xin-Qing, Z., Suo-Lian, G., Asraful, A., & Feng-Wu, B. (2013). Bioflocculant production from *Solibacillus silvestris* W01 and its application in cost-effective harvest of marine microalga *Nannochloropsis oceanica* by flocculation. *Bioresource Technology*, 135, 207-212.
- Wang, S.G., Gong, W.X., Liu, X.W., Tian, L., Yue, Q.Y., & Gao, B.Y. (2007). Production of a novel bioflocculant by culture of *Klebsiella mobilis* using dairy wastewater. *Biochemical Engineering Journal*, 36, 81-86.

- Wang, Y., Gao, B., Yue, Q., Wei, J.C., Zhou, W., & Gu, R. (2010). Color removal from textile industry wastewater using composite flocculants. *Environmental Technology*, 28(6), 629-637.
- Wang, Y., Ahmed, Z., Feng, W., Li, C., & Song, S. (2008). Physicochemical properties of exopolysaccharide produced by *Lactobacillus kefiranofaciens* ZW3 isolated from Tibet kefir. *International Journal of Biological Macromolecules*, 43(3), 283-288
- Wang, Y., Gao, B., Yue, Q., Wei, J., & Li, Q. (2007). The characterization and flocculation efficiency of composite flocculant iron salts-poly-dimethyl-diallyl almmonium chloride. *Chemical Engineering Journal*, *142*, 175-181.
- Wang, M., Kuo-Dahab, C., Dolan, S., & Park, C. (2014). Kinetics of nutrient removal and expression of extracellular polymeric substances of the microalgae, *Chlorella* sp. and *Micractinium* sp. in wastewater treatment. *Bioresource Technology*, 154, 131-137.
- Wang, L., Ma, F., Qu, Y., Sun, D., Li, A., Guo, J., & Yu, B. (2011). Characterization of a compound bioflocculant produced by mixed culture of *Rhizobium radiobacter* F2 and *Bacillus sphaeicus* F6. World Journal of Microbiology and Biotechnology, 27(11), 2559-2565.
- Wang, Y., Li X., Wu, C., Zhao Y., Gao, B.Y., & Yue, Q. (2014). The role of sodium alginate in improving floc size and strength and the subsequent effects on ultrafiltration membrane fouling. *Environmental Technology*, 35(1), 10-17.
- Wang, Y., Yang, Y., Ma, F., Xuan, L., Xu, Y., Huo, H.,... Dong, S. (2015). Optimization of *Chlorella vulgaris* and bioflocculant-producing bacteria co-culture: enhancing microalgae harvesting and lipid content. *Letters in Applied Microbiology*, 60(5), 497-503.
- Wang, L., Ma, F., Jong, L.D., Wang, A., & Ren, N. (2013b). Bioflocculants from hydrolysates of corn stover using isolated strain Ochrobactium ciceri W2. Bioresource Technology, 145, 259-263.
- Wei, X., Fang, L., Cai, P., Huang, Q., Chen, H., Liang, W.,... Rong, X., (2011). Influence of extracellular polymeric substances (EPS) on Cd adsorption by bacteria. *Environmental Pollutution*, 159, 1369–1374.
- Wei, Y., Mu, J., Zhu, X., Gao, Q., & Zhang, Y. (2011). Decolorization of dye solutions with *Ruditapes philippinarum* conglutination mud and the isolated bacteria. *Journal of Environmental Science*, 23, 142-145.
- Wei, X., Luo, M., Li, W., Yang, L., Liang, X., Xu, L., ... Liu, H. (2012). Synthesis of silver nanoparticles by solar irradiation of cell free *Bacillus amyloliquefeciens* extracts and AgNO₃. *Bioresource Technology*, 103, 273–278.
- Wen, L.T & Wang, S.L. (2015). Recent advances in exopolysaccharides from *Paenibacillus* sp. production, isolation, structure, and bioactivities. *Marine Drugs*, 13(4), 1847-1863.

- Wolfaardt, G., Lawrence, J., & Korber, D. (1999). Function of EPS. In Microbial extracellular polymeric substances (p. 171-200). Berlin, Heidelberg: Springer.
- Wong, Y., Ong, S., Teng, T., Nabilah, A., & Kumaran, K. (2012). Production of bioflocculant by *Staphylococcus cohnii* sp. from palm oil mill effluent (POME). *Water Air and Soil Pollution*, 223, 3775-3781.
- Wong, S., Teng, T., Ahmada, A., Zuhairi, A., & Najafpour, G. (2006). Treatment of pulp and paper mill wastewater by polyacrylamide (PAM) in polymer induced flocculation. *Journal of Hazardous Materials*, 135(1-3), 378-388.
- World Health Organization (2015). *Global Water Supply and Sanitation Assessment Report 2000*. Geneva, Switzerland: UNICEF.
- World Health Organization (2017). *Global Water Supply and Sanitation Assessment Report 2000.* Geneva, Switzerland: UNICEF.
- Wu, C., Wang, Y., Gao, B., Zhao, Y., & Yue, Q. (2012). Coagulation performance and floc characteristics of aluminum sulfate using sodium alginate as coagulant aid for synthetic dying wastewater treatment. *Separation and Purification Technology*, 95, 180-187.
- Wu, J. & Ye, H. (2007). Characterization and flocculating properties of an extracellular biopolymer produced from a *Bacillus subtillis* DYU1 isolate. *Process Biochemistry Journal*, 42, 1114-1123.
- Xia, S., Zhang, Z., Wang, X., Yang, A., Chen, L., Zhao, J., ... Renault, N. (2008). Production and characterization of bioflocculant by *Proteus mirabilis* TJ-1. *Bioresource Technology*, 99(14), 6520-6527.
- Xiang, X., Liang, Y., Lan, S., Li, X., Xie, Y & Yuan, W. (2018). Production and flocculating properties of a compound biopolymer flocculant from corn ethanol wastewater. *Bioresource Technology*, 247, 924-929.
- Xing, J., Yang, J., Li, A., Ma, F., Liu, K., Wu, D., ... Wei, W. (2013). Removal efficiency and mechanism of sulfamethoxazole in aqueous solution by bioflocculant MFX. *Journal of Analytical Methods in Chemistry*, 2013, Article # 568614.
- Xiong, Y., Wang, Y., Yu, Y., Li, Q., Wang, H., Chen, R.C., ... He, N. (2010). Production and characterization of a novel bioflocculant from *Bacillus licheniformis*. *Applied and Environmental Microbiology*, 76(9), 2778-2782.
- Yan, S., Wang, N., Zhen, C., Yuanpeng, W., Ning, H., Yajuan, P., ... Xu, D. (2013). Genes encoding the production of extracellular polysaccharide bioflocculant are clustered on a 30-kb DNA segment in *Bacillus licheniformis*. *Functional Integrative Genomics*, 13(4), 425-434.
- Yang, Q., Ming, H., Zhao, X., Zhang, J., Zou, W., Zhao, C., ... Guan, X. (2015). Screening of bioflocculant and preliminary application to treatment of tannery wastewater. *Journal of Residuals Science and Technology*, 12, 177-181.
- Yang, R., Li, H., Huang, M., Yang, H. & Li, A. (2016). A review on chitosan-based flocculants and their applications in water treatment. *Water Research*, 95, 59-89.
- Yang, Q., Luo, K., Liao, D., Li X., Wang, D., Liu, X., ... Li, X. (2012). A novel bioflocculant produced by *Klebsiella* sp. and its application to sludge dewatering. *Water and Environmental Journal*, 26, 560-566.
- Yan, S., Wang, N., Zhen, C., Yuanpeng, W., Ning, H., Yajuan, P., ... Deng, X. (2013). Genes encoding the production of extracellular polysaccharide bioflocculant are clustered on a 30-kb DNA segment in *Bacillus licheniformis*. *Functional Integrative Genomics*, 13(4), 425-434.
- Ye, G., Li, G., Wang, C., Ling, B., Yang, R., & Huang, S. (2019). Extraction and characterization of dextran from *Leuconostoc pseudomesenteroides* YB-2 isolated from mango juice. *Carbohydrate Polymers*, 207, 218-223.
- Yin, Y., Tian, Z., Tang, W., Li L., Song, L., Shawn P.,... McElmurry, S. (2014). Production and characterization of high efficiency bioflocculant isolated from *Klebsiella* sp. ZZ-3. *Bioresource Technology*, 171, 336-342.
- Yim, J.H., Kim, J.S., Ahn, S.H., & Lee, H.K. (2007). Characterization of a novel bioflocculant, p-KG03, from a marine dinoflagellate, *Gyrodinium impudicum* KG03. *Bioresource Technology*, 98, 361-367.
- Yokoi, H., Arima, T., Hirose, J., Hayashi, S., & Takasaki, Y. (1996). Flocculation properties of poly (γ-glutamic acid) produced by *Bacillus subtilis*. *Journal of Fermentation and Bioengineering*, 82, 84–87.
- Yokoi, H., Yoshida, T., Mori, S., Hirose, J., Hayashi, S., & Takasaki, Y. (1997). Biopolymer flocculant produced by an *Enterobacter* sp. *Biotechnolog Letters*, 19, 569-573.
- Yong, X., Raza, W., Yu, G., Ran, W, Shen, Q & Yang, X. (2011). Optimization of the production of poly-Y glutamic acid by *Bacillus antheyloliquefaciens* C1 in solidstate fermentation using dairy manure compost and monosodium glutamate production residues as basic substrates. *Bioresource Technology*, 102(16), 7548-4554.
- You, Y.N., Ren, A., Wang F., Ma, L., Gao, Y.P & Lee, D. (2008). Use waste fermenting liquor to produce bioflocculants with isolated strains. *International Journal Hydrogen Energy*, 33(13), 3295-3301.
- Yu, L., Hong-Guan, L., & Liu, F. (2017). Pollution in the urban soils of Lianyungang, China, evaluated using a pollution index, mobility of heavy metals, and enzymatic activities. *Environmental Monitoring and Assessment*, 189(1), Article # 34.
- Zaki, S., Elkady, M., Farag, S., & Abd-El-Haleem, D. (2013). Characterization and flocculation properties of a carbohydrate bioflocculant from a newly isolated *Bacillus velezensis* 40B. *Journal of Environmental Biology*, 34, 51-58.

- Zaki, S., Etarahony, M., Elkady, M., & Abd-El- Haleem, D. (2014). The use of bioflocculant and bioflocculant producing *Bacillus mojavensis* strain 32A to synthesize silver nanoparticles. *Journal of Nanomaterials*, 5, 2-9.
- Zemmouri, H., Kadouche, S., Lounici, H., Hadioui, M., & Mameri, N. (2011). Use of Chitosan in coagulation flocculation of raw water of Algeirs Dams. *Water Science* and Technology Water Supply, 11, 1-5.
- Zhang, C.L., Cui, Y.N., & Wang, Y. (2013). Bioflocculants Produced by Gram-positive Bacillus xn12 and Streptomyces xn17 for swine wastewater application. Chemical and Biochemical Engineering Quarterly, 27(2), 245–250.
- Zhang, J., Wang, R., Jiang, P., & Liu, Z. (2002b). Production of an exopolysaccharide bioflocculant by Sorangium cellulosum. Letters in Applied Microbiology, 34, 178–181.
- Zhang, Z., Xia, S., Zhao, J., & Zhang, J. (2010). Characterization and flocculation mechanism of high efficiency microbial flocculant TJ-F1 from *Proteus mirabilis*. *Colloids and Surfaces B: Biointerfaces*, 75(1), 247-251.
- Zhang, Z., Ling, B., Xia, S., Wang, X., & Yang, A. (2007). Production and application of a novel biofloculant by multiple-microorganism consortia using brewery wastewater as carbon source. *Journal of Environmental Science*, 19(6), 667-673.
- Zhang, Z., Chunde, W., Yue, W., & Caixia, H. (2014). Comparison of coagulation performance and floc properties of a novel zirconium-glycine complex coagulant with traditional coagulants. *Environmental Science and Pollution Research International*, 21(10), 6632-6639.
- Zhang, L., Liu Z., Cui, G., & Chen, L. (2015). Biomass-derived materials for electrochemical energy storages. *Progress in Polymers Science*, 43, 136-164.
- Zhang, J., Liu, Z., Wang, S., & Jiang, P. (2002). Characterization of a bioflocculant produced by the marine *Myxobacterium nannocystis* sp. NU-2. Applied Microbiology and Biotechnology, 59(4-5), 517-522.
- Zhang, Z., Lin, B., Xia, S., Wang, X., & Yang, A. (2007). Production and application of a novel bioflocculant by multiple microorganism consortia using brewery wastewater as carbon source. *Journal of Environmental Science*, 19(6), 667-673.
- Zhao, G., Ma, F., Wei, L., & Chua, H. (2012). Using rice straw fermentation liquor to produce bioflocculants during an anaerobic dry fermentation process. *Bioresource Technology*, 113, 83-88.
- Zheng, Y., Ye, Z., Fang, X., Li, Y., & Cai, W. (2008). Production and characteristics of a bioflocculant produced by *Bacillus* sp. F19. *Bioresource Technology*, 99(16), 7686-7691.
- Zhengshan, L., Li, C., Changhong, C., Wei, Z., Ming, L., Ye, H.,... Jiangang, Z. (2014). Production and characteristics of a bioflocculant by *Klebsiella pneumoniae* YZ-6 isolated from Human Saliva. *Applied Biochemistry and Biotechnology*, 172(3), 1282-1292.

- Zhong, C., Cao, G., Rong, K., Xia, Z., Peng, T., Chen, H., ... Zhou, J. (2018). Characterization of a microbial polysaccharide-based bioflocculant and its antiinflammatory and pro-coagulant activity. *Colloids Surfaces B: Biointerfaces*, 161, 636-644.
- Zhong, C., Xu, A., Chen, L., Yang, X., Yang, B., Hong, W., ... Zhou, J. (2014). Production of a bioflocculant from chromotropic acid waste water and its application in steroid estrogen removal. *Colloid Surfaces B: Biointerfaces*, 122, 729-737.
- Zhou, X., Wang, J., & Zhou, J. (2003). Studies on properties of a bioflocculant produced by *Pseudomonas* sp. GX4-1 in fish meal wastewater. *Research of Environmental Science*, 16(3), 31-34.
- Zhuang, X., Wang, Y., Li, Q., Yan, S., & He, N. (2012). The production of bioflocculant by *Bacillus licheniformis* using molasses and its application in the sugarcane industry. *Biotechnology and Bioprocess Engineering*, 17(5), 1041-1047.
- Zhu, C., Chen, C., Zhao, L., Zhang, Y., Yang, J., Song, L., ... Yang, S. (2012). Bioflocculant produced by *Chlomydomonase reinhardtii*. Journal of Applied Phycology, 24, 1245-1251.
- Zhu, S., Hou, H., & Xue, Y. (2008). Kinetic and isothermal studies of lead ion adsorption onto bentonite. *Applied Clay Science*, 40, 171–178.
- Zulkeflee, Z., Aris, A., Shamsuddin, Z., & Yusoff, M. (2012). Cation dependence, pH tolerance, and dosage requirement of a bioflocculant produced by *Bacillus* spp. UPMB13: flocculation performance optimization through kaolin assays. *The Scientific World Journal*, 2012, Article # 495659.